CHAPTER 1

Domain Perturbation for Linear and Semi-Linear Boundary Value Problems

Daniel Daners

School of Mathematics and Statistics, The University of Sydney, NSW 2006, Australia E-mail: D.Daners@maths.usyd.edu.au URL: www.maths.usyd.edu.au/u/daners/

Abstract

This is a survey on elliptic boundary value problems on varying domains and tools needed for that. Such problems arise in numerical analysis, in shape optimisation problems and in the investigation of the solution structure of nonlinear elliptic equations. The methods are also useful to obtain certain results for equations on non-smooth domains by approximation by smooth domains.

Domain independent estimates and smoothing properties are an essential tool to deal with domain perturbation problems, especially for non-linear equations. Hence we discuss such estimates extensively, together with some abstract results on linear operators.

A second major part deals with specific domain perturbation results for linear equations with various boundary conditions. We completely characterise convergence for Dirichlet boundary conditions and also give simple sufficient conditions. We then prove boundary homogenisation results for Robin boundary conditions on domains with fast oscillating boundaries, where the boundary condition changes in the limit. We finally mention some simple results on problems with Neumann boundary conditions.

The final part is concerned about non-linear problems, using the Leray-Schauder degree to prove the existence of solutions on slightly perturbed domains. We also demonstrate how to use the approximation results to get solutions to nonlinear equations on unbounded domains.

Keywords: Elliptic boundary value problem, Domain perturbation, Semilinear equations, A priori estimates, Boundary homogenisation

HANDBOOK OF DIFFERENTIAL EQUATIONS Stationary Partial Differential Equations, volume 6 Edited by M. Chipot

^{© 2008} Elsevier B.V. All rights reserved

Co	ntents	
1.	Introduction	3
2.	Elliptic boundary value problems in divergence form	5
	2.1. Weak solutions to elliptic boundary value problems	5
	2.2. Abstract formulation of boundary value problems	9
	2.3. Formally adjoint problems	2
	2.4. Global a priori estimates for weak solutions	3
	2.4.1. Sobolev inequalities associated with Dirichlet problems	6
	2.4.2. Maz'ya's inequality and Robin problems	7
	2.4.3. Sobolev inequalities associated with Neumann problems	8
	2.5. The pseudo-resolvent associated with boundary value problems	8
3.	Semi-linear elliptic problems	0
	3.1. Abstract formulation of semi-linear problems	0
	3.2. Boundedness of weak solutions	2
4.	Abstract results on linear operators	3
	4.1. Convergence in the operator norm	4
	4.2. A spectral mapping theorem	5
	4.3. Convergence properties of resolvent and spectrum	6
5.	Perturbations for linear Dirichlet problems	9
	5.1. Assumptions and preliminary results	9
	5.2. The main convergence result	1
	5.3. Necessary conditions for convergence	5
	5.4. Sufficient conditions for convergence	7
	5.5. Proof of the main convergence result	9
6.	Varying domains and Robin boundary conditions	2
	6.1. Summary of results	-2
	6.2. Preliminary results	5
	6.3. Small modifications of the original boundary	7
	6.4. Boundary homogenisation: Limit is a Dirichlet problem	0
	6.5. Boundary homogenisation: Limit is a Robin problem	3
7.	Neumann problems on varying domains	5
	7.1. Remarks on Neumann problems	5
	7.2. Convergence results for Neumann problems	6
8.	Approximation by smooth data and domains	8
	8.1. Approximation by operators having smooth coefficients	8
	8.2. Approximation by smooth domains from the interior	0
	8.3. Approximation from the exterior for Lipschitz domains	0
9.	Perturbation of semi-linear problems	52
	9.1. Basic convergence results for semi-linear problems	2
	9.2. Existence of nearby solutions for semi-linear problems	5
	9.3. Applications to boundary value problems	8
	9.4. Remarks on large solutions	0
	9.5. Solutions by domain approximation	2
	9.6. Problems on unbounded domains	3
Re	?ferences	5

1. Introduction

The purpose of this survey is to look at elliptic boundary value problems

$$\mathcal{A}_n u = f \quad \text{in } \Omega_n,$$

$$\mathcal{B}_n u = 0 \quad \text{on } \partial \Omega_n$$

with all major types of boundary conditions on a sequence of open sets Ω_n in \mathbb{R}^N ($N \ge 2$). We then study conditions under which the solutions converge to a solution of a limit problem

$$\mathcal{A}u = f \quad \text{in } \Omega,$$
$$\mathcal{B}u = 0 \quad \text{on } \partial \Omega$$

on some open set $\Omega \subset \mathbb{R}^N$. In the simplest case $\mathcal{A}, \mathcal{A}_n = -\Delta$ is the negative Laplace operator, and $\mathcal{B}_n, \mathcal{B}$ the Dirichlet, Robin or Neumann boundary operator, but we work with general non-selfadjoint elliptic operators in divergence form. We are interested in very singular perturbation, not necessarily of a type such that a change of variables can be applied to reduce the problem onto a fixed domain. For the theory of smooth perturbations rather complementary to ours we refer to [84] and references therein.

The main features of this exposition are the following:

- We present an L_p -theory of linear and semi-linear elliptic boundary value problems with domain perturbation in view.
- We establish domain perturbation results for linear elliptic problems with Dirichlet, Robin and Neumann boundary conditions, applicable to semi-linear problems.
- We show how to use the linear perturbation theory to deal with semi-linear problems on bounded and unbounded domains. In particular we show how to get multiple solutions for simple equations, discuss the issue of precise multiplicity and the occurrence of large solutions.
- We provide abstract perturbation theorems useful also for perturbations other than domain perturbations.
- We provide tools to prove results for linear and nonlinear equations on general domains by means of smoothing domains and operators (see Section 8).

Our aim is to build a domain perturbation theory suitable for applications to semi-linear problems, that is, problems where f = f(x, u(x)) is a function of x as well as the solution u(x). For nonlinearities with growth, polynomial or arbitrary, we need a good theory for the linear problem in L_p for 1 . Good in the context of domain perturbations means that in all estimates there is*control on domain dependence* $of the constants involved. We establish such a theory in Section 2.1, where we also introduce precise assumptions on the operators. Starting from a definition of weak solutions we prove smoothing properties of the corresponding resolvent operators with control on domain dependence. The main results are Theorems 2.4.1 and 2.4.2. In particular we prove that the resolvent operators have smoothing properties independent of the domain for Dirichlet and Robin boundary conditions, but not for Neumann boundary conditions. To be able to work in a common space we consider the resolvent operator as a map acting on <math>L_p(\mathbb{R}^N)$, so that it becomes a pseudo-resolvent (see Section 2.5).

The smoothing properties of the resolvent operators enable us to reformulate a semilinear boundary value problem as a fixed point problem in $L_p(\mathbb{R}^N)$. Which $p \in (1, \infty)$ we choose depends on the growth of the nonlinearity. The rule is, the faster the growth, the larger the choice of p. We also show that under suitable growth conditions, a solution in L_p is in fact in L_∞ . Again, the focus is on getting control over the domain dependence of the constants involved. For a precise formulation of these results we refer to Section 3.

Let the resolvent operator corresponding to the linear problems be denoted by $R_n(\lambda)$ and $R(\lambda)$. The key to be able to pass from perturbations of the linear to perturbations of the nonlinear problem is the following property of the resolvents:

If
$$f_n \rightarrow f$$
 weakly, then $R_n(\lambda) f_n \rightarrow R(\lambda) f$ strongly.

Hence for all types of boundary conditions we prove such a statement. If $R(\lambda)$ is compact, it turns out that the above property is equivalent to convergence in the operator norm. An issue connected with that is also the convergence of the spectrum. We show that the above property implies the convergence of every finite part of the spectrum of the relevant differential operators. We also show here that it is sufficient to prove convergence of the resolvent in $L_p(\mathbb{R}^N)$ for some p and some λ to have them for all. These abstract results are collected in Section 4.

The most complete convergence results are known for the Dirichlet problem (Section 5). The limit problem is always a Dirichlet problem on some domain. We extensively discuss convergence in the operator norm. In particular, we look at necessary and sufficient conditions for pointwise and uniform convergence of the resolvent operators. As a corollary to the characterisation we see that convergence is independent of the operator chosen. We also give simple sufficient conditions for convergence in terms of properties of the set $\Omega_n \cap \overline{\Omega}^c$. The main source for these results is [58].

The situation is rather more complicated for Robin boundary conditions, where the type of boundary condition can change in the limit problem. In Section 6 we present three different cases. First, we look at problems with only a small perturbation of $\partial \Omega$. We can cut holes and add thin pieces outside Ω , connected to Ω only near a set of capacity zero. Second, we look at approximating domains with very fast oscillating boundary. In that case the limit problem has Dirichlet boundary conditions. Third, we deal with domains with oscillating boundary, such that the limit problem involves Robin boundary conditions with a different weight on the boundary. The second and third results are really *boundary homogenisation* results. These results are all taken from [51].

The Neumann problem is very badly behaved, and without quite severe restriction on the sequence of domains Ω_n we cannot expect the resolvent to converge in the operator norm. In particular the spectrum does not converge, as already noted in [40, page 420]. We only prove a simple convergence result fitting into the general framework established for the other boundary conditions.

After dealing with linear equations we consider semi-linear equations. A lot of this part is inspired by Dancer's paper [45] and related work. The approach is quite different since we treat linear equations first, and then use their properties to deal with nonlinear equations. The idea is to use degree theory to get solutions on a nearby domain, given a solution of the limit problem. The core of the argument is an abstract topological argument which may be useful also for other types of perturbations (see Section 9.2). We also discuss the issue of

precise multiplicity of solutions and the phenomenon of large solutions. Finally, we show that the theory also applies to unbounded limit domains.

There are many other motivations to look at domain perturbation problems, so for instance variational inequalities (see [102]), numerical analysis (see [77,107,110,116–119]), potential and scattering theory (see [10,108,113,124]), control and optimisation (see [31,34,82,120]), Γ -convergence (see [24,42]) and solution structures of nonlinear elliptic equations (see [45,47,52,69]). We mention more references in the discussion on the specific boundary conditions. Some results go back a long time, see for instance [19] or [40]. The techniques are even older for the Dirichlet problem for harmonic functions with the pioneering work [93].

Finally, there are many results we do not even mention, so for instance for convergence in the L_{∞} -norm we refer to [8,9,14,23,26]. Furthermore, similar results can be proved for parabolic problems. The key for that are domain-independent heat kernel estimates. See for instance [7,17,47,52,59,78] and references therein. The above is only a small rather arbitrary selection of references.

2. Elliptic boundary value problems in divergence form

The purpose of this section is to give a summary of results on elliptic boundary value problems in divergence form with emphasis on estimates with control over the domain dependence.

2.1. Weak solutions to elliptic boundary value problems

We consider boundary value problems of the form

$$\begin{aligned} \mathcal{A}u &= f & \text{in } \Omega, \\ \mathcal{B}u &= 0 & \text{on } \partial\Omega \end{aligned}$$
 (2.1.1)

on an open subset of \mathbb{R}^N , not necessarily bounded or connected. Here \mathcal{A} is an elliptic operator in divergence form and \mathcal{B} a boundary operator to be specified later in this section. The operator \mathcal{A} is of the form

$$-\operatorname{div}(A_0\nabla u + au) + b \cdot \nabla u + c_0 u \tag{2.1.2}$$

with $A_0 \in L_{\infty}(\Omega, \mathbb{R}^{N \times N})$, $a, b \in L_{\infty}(\Omega, \mathbb{R}^N)$ and $c_0 \in L_{\infty}(\Omega)$. Moreover, we assume that $A_0(x)$ is positive definite, uniformly with respect to $x \in \Omega$. More precisely, there exists a constant $\alpha_0 > 0$ such that

$$\alpha_0|\xi|^2 \le \xi^T A_0(x)\xi \tag{2.1.3}$$

for all $\xi \in \mathbb{R}^N$ and almost all $x \in \Omega$. We call α_0 the *ellipticity constant*.

REMARK 2.1.1. We only defined the operator \mathcal{A} on Ω , but we can extend it to \mathbb{R}^N by setting a = b = 0, $c_0 = 0$ and $A(x) := \alpha_0 I$ on Ω^c . Then the extended operator \mathcal{A} also

satisfies (2.1.3). In particular the ellipticity property (2.1.3) holds. Hence without loss of generality we can assume that A is defined on \mathbb{R}^N .

We further define the *co-normal derivative* associated with A on $\partial \Omega$ by

$$\frac{\partial u}{\partial v_{\mathcal{A}}} := (A_0(x)\nabla u + a(x)u) \cdot v,$$

where ν is the outward pointing unit normal to $\partial\Omega$. Assuming that $\partial\Omega$ is the disjoint union of Γ_1 , Γ_2 and Γ_3 we define the boundary operator \mathcal{B} by

$$\mathcal{B}u := \begin{cases} u|_{\Gamma_1} & \text{on } \Gamma_1 \text{ (Dirichlet b.c.),} \\ \frac{\partial u}{\partial \nu_{\mathcal{A}}} & \text{on } \Gamma_2 \text{ (Neumann b.c.),} \\ \frac{\partial u}{\partial \nu_{\mathcal{A}}} + b_0 u & \text{on } \Gamma_3 \text{ (Robin b.c.)} \end{cases}$$
(2.1.4)

with $b_0 \in L_{\infty}(\Gamma_3)$ nonnegative. If all functions involved are sufficiently smooth, then by the product rule

 $-v\operatorname{div}(A_0\nabla u + au) = (A_0\nabla u + au) \cdot \nabla v - \operatorname{div}(v(A_0\nabla u + au))$

and therefore, if Ω admits the divergence theorem, then

$$-\int_{\Omega} v \operatorname{div}(A_0 \nabla u + au) \, dx$$

= $\int_{\Omega} (A_0 \nabla u + au) \cdot \nabla v \, dx - \int_{\partial \Omega} (v (A_0 \nabla u + au)) \cdot v \, d\sigma,$

where σ is the surface measure on $\partial \Omega$. Hence, if *u* is sufficiently smooth and $v \in C^1(\overline{\Omega})$ with v = 0 on Γ_1 , then

$$\int_{\Omega} v \mathcal{A} u \, dx = \int_{\Omega} (A_0 \nabla u + au) \cdot \nabla v + (a \cdot \nabla u + c_0 u) v \, dx + \int_{\Gamma_3} b_0 u v \, d\sigma.$$

The expression on the right-hand side defines a bilinear form. We denote by $H^1(\Omega)$ the usual Sobolev space of square integrable functions having square integrable weak partial derivatives. Moreover, $H_0^1(\Omega)$ is the closure of the set of test functions $C_c^{\infty}(\Omega)$ in $H^1(\Omega)$.

DEFINITION 2.1.2. For $u, v \in H^1(\Omega)$ we set

$$a_0(u,v) := \int_{\Omega} (A_0 \nabla u + au) \cdot \nabla v + (b \cdot \nabla u + c_0 u) v \, dx.$$

The expression

$$a(u, v) := a_0(u, v) + \int_{\Gamma_3} b_0 u v \, d\sigma$$

is called the *bilinear form associated with* $(\mathcal{A}, \mathcal{B})$.

If u is a sufficiently smooth solution of (2.1.1), then

$$a(u, v) = \langle f, v \rangle := \int_{\Omega} f v \, dx \tag{2.1.5}$$

for all $v \in C^1(\overline{\Omega})$ with v = 0 on Γ_1 . Note that (2.1.5) does not just make sense for classical solutions of (2.1.1), but for $u \in H^1(\Omega)$ as long as the boundary integral is defined. We therefore generalise the notion of solution and just require that u is in a suitable subspace V of $H^1(\Omega)$ and (2.1.5) for all $v \in V$.

ASSUMPTION 2.1.3. We require that V be a Hilbert space such that V is dense in $L_2(\Omega)$, that

$$H_0^1(\Omega) \hookrightarrow V \hookrightarrow H^1(\Omega),$$

and that

$$\left\{u\in C^1(\bar{\Omega})\colon \operatorname{supp} u\subset \bar{\Omega}\setminus \Gamma_1, \int_{\Gamma_3} b_0|u|^2\,d\sigma<\infty\right\}\subset V.$$

If Γ_3 is nonsmooth we replace the surface measure σ by the (N-1)-dimensional Hausdorff measure so that the boundary integral makes sense. We also require that

$$\|u\|_{V} := \left(\|u\|_{H^{1}(\Omega)}^{2} + \alpha_{0}^{-1}\|u\sqrt{b_{0}}\|_{L_{2}(\Gamma_{3})}^{2}\right)^{1/2}$$
(2.1.6)

is an equivalent norm on V.

We next consider some specific special cases.

EXAMPLE 2.1.4. (a) For a homogeneous Dirichlet problem we assume that $\Gamma_1 = \partial \Omega$ and let $V := H_0^1(\Omega)$. For the norm we can choose the usual H^1 -norm, but on bounded domains we could just use the equivalent norm $\|\nabla u\|_2$. More generally, on domains Ω lying between two hyperplanes of distance D, we can work with the equivalent norm $\|\nabla u\|_2$ because of Friedrich's inequality

$$\|u\|_{2} \le D \|\nabla u\|_{2} \tag{2.1.7}$$

valid for all $u \in H_0^1(\Omega)$ (see [111, Theorem II.2.D]).

(b) For a homogeneous Neumann problem we assume that $\Gamma_2 = \partial \Omega$ and let $V := H^1(\Omega)$ with the usual norm.

(c) For a homogeneous Robin problem we assume that $\Gamma_3 = \partial \Omega$. In this exposition we will always assume that Ω is a Lipschitz domain when working with Robin boundary conditions and choose $V := H^1(\Omega)$. On a bounded domain we can work with the equivalent norm

$$\|v\|_{V} := \left(\|\nabla u\|_{2}^{2} + \|u\|_{L_{2}(\partial\Omega)}^{2}\right)^{1/2}$$
(2.1.8)

(see [111, Theorem III.5.C] or [56]). It is possible to admit arbitrary domains as shown in [11,56].

We finally define what we mean by a weak solution of (2.1.1).

DEFINITION 2.1.5 (Weak solution). We say u is a weak solution of (2.1.1) if $u \in V$ and (2.1.5) holds for all $v \in V$. Moreover, we say that u is a weak solution of $\mathcal{A} = f$ in Ω if $u \in H^1_{loc}(\Omega)$ such that (2.1.5) holds for all $v \in C^{\infty}_c(\Omega)$.

Note that a weak solution of $\mathcal{A} = f$ on Ω does not need to satisfy any boundary conditions. As we shall see, it is often easy to get a weak solution in Ω by domain approximation. The most difficult part is to verify that it satisfies boundary conditions. We next collect some properties of the form $a(\cdot, \cdot)$ on V. In what follows we use the norms

$$||A||_{\infty} := \left(\sum_{i,j=1}^{N} ||a_{ij}||_{\infty}^{2}\right)^{1/2} \text{ and } ||a||_{\infty} := \left(\sum_{i=1}^{N} ||a_{i}||_{\infty}^{2}\right)^{1/2}$$

for matrices $A = [a_{ij}]$ and vectors $a = (a_1, \ldots, a_N)$.

PROPOSITION 2.1.6. Suppose that $(\mathcal{A}, \mathcal{B})$ is defined as above. Then there exists M > 0 such that

$$|a(u,v)| \le M \|u\|_V \|v\|_V \tag{2.1.9}$$

for all $u, v \in V$. More precisely we can set

$$M = ||A||_{\infty} + ||a||_{\infty} + ||b||_{\infty} + ||c_0||_{\infty} + \alpha_0.$$

Moreover, if we let

$$\lambda_{\mathcal{A}} := \|c_0^-\|_{\infty} + \frac{1}{2\alpha_0} \|a + b\|_{\infty}^2, \qquad (2.1.10)$$

where α_0 is the ellipticity constant from (2.1.3), then

$$\frac{\alpha_0}{2} \|\nabla u\|_2^2 \le a_0(u, u) + \lambda_{\mathcal{A}} \|u\|_2^2$$
(2.1.11)

for all $u \in H^1(\Omega)$. Finally, setting $\lambda_0 := \lambda_A + \alpha_0/2$ we see that

$$\frac{\alpha_0}{2} \|u\|_V^2 \le a(u, u) + \lambda_0 \|u\|_2^2$$

for all $u \in V$.

PROOF. By the Cauchy–Schwarz inequality and the definition of $a_0(\cdot, \cdot)$

$$\begin{aligned} |a_0(u, v)| &\leq \|A\nabla u\|_2 \|\nabla v\|_2 + \|au\|_2 \|\nabla v\|_2 + \|b\nabla u\|_2 \|v\|_2 + \|c_0 u\|_2 \|v\|_2 \\ &\leq \|A\|_{\infty} \|\nabla u\|_2 \|\nabla v\|_2 + \|a\|_{\infty} \|u\|_2 \|\nabla v\|_2 \\ &+ \|b\|_{\infty} \|\nabla u\|_2 \|v\|_2 + \|c_0\|_{\infty} \|u\|_2 \|v\|_2 \\ &\leq (\|A\|_{\infty} + \|a\|_{\infty} + \|b\|_{\infty} + \|c_0\|_{\infty}) \|u\|_V \|v\|_V \end{aligned}$$

for all $u, v \in V$. Similarly for the boundary integral

$$\int_{\Gamma_3} b_0 u v \, d\sigma \leq \| u \sqrt{b_0} \|_{L_2(\Gamma_3)} \| v \sqrt{b_0} \|_{L_2(\Gamma_3)} \leq \alpha_0 \| u \|_V \| v \|_V$$

for all $u, v \in V$, where we used (2.1.6) for the definition of the norm in V. Combining the two inequalities, (2.1.9) follows. We next prove (2.1.11). Given $u \in H^1(\Omega)$, using (2.1.3) we get

$$\begin{aligned} \alpha_0 \|\nabla u\|_2^2 &\leq \int_{\Omega} (A\nabla u) \cdot \nabla u \, dx \\ &\leq a_0(u, u) - \int_{\Omega} (a+b)u \cdot \nabla u + c_0^- |u|^2 \, dx \\ &\leq a_0(u, u) + \|a+b\|_{\infty} \|u\|_2 \|\nabla u\|_2 + \|c_0^-\|_{\infty} \|u\|_2^2 \\ &\leq a_0(u, u) + \frac{1}{2\alpha_0} \|a+b\|_{\infty}^2 \|u\|_2^2 + \frac{\alpha_0}{2} \|\nabla u\|_2^2 + \|c_0^-\|_{\infty} \|u\|_2^2 \end{aligned}$$

if we use the elementary inequality $xy \le x^2/2\varepsilon + \varepsilon y^2/2$ valid for $x, y \ge 0$ and $\varepsilon > 0$ in the last step. Rearranging the inequality we get (2.1.11). If we add $\alpha_0 ||u||_2^2/2$ and the boundary integral if necessary to (2.1.11), then the final assertion follows.

2.2. Abstract formulation of boundary value problems

We saw in Section 2.1 that all boundary value problems under consideration have the following structure.

ASSUMPTION 2.2.1 (Abstract elliptic problem). There exist Hilbert spaces V and H such that $V \hookrightarrow H$ and V is dense in H. Suppose there exists a bilinear form

 $a(\cdot, \cdot) \colon V \times V \to \mathbb{R}$

with the following properties. There exists a constant M > 0 such that

$$|a(u,v)| \le M \|u\|_V \|v\|_V \tag{2.2.1}$$

for all $u, v \in V$. Also, there exist constants $\alpha > 0$ and $\lambda_0 \ge 0$ such that

$$\alpha \|u\|_V^2 \le a(u, u) + \lambda_0 \|u\|_H^2 \tag{2.2.2}$$

for all $u \in V$.

By assumption on V and H we have

 $V \hookrightarrow H \hookrightarrow V'$

if we identify H with its dual H' by means of the Riesz representation theorem with both embeddings being dense. In particular

 $|\langle u, v \rangle| \le \|u\|_{V'} \|u\|_V$

for all $u, v \in H$ and duality coincides with the inner product in *H*. By (2.2.1) the map $v \to a(u, v)$ is bounded and linear for every fixed $u \in V$. If we denote that functional by $Au \in V'$, then

$$a(u, v) = \langle Au, v \rangle$$

for all $u, v \in V$. The operator $A: V \to V'$ is linear and by (2.2.1) we have $A \in \mathcal{L}(V, V')$ with

$$\|A\|_{\mathcal{L}(V,V')} \le M.$$

We say A is the operator induced by the form $a(\cdot, \cdot)$. We can also consider it as an operator on V' with domain V.

THEOREM 2.2.2. Let $A \in \mathcal{L}(V, V')$ be defined as above. Then A is a densely defined closed operator on V' with domain V. Moreover,

$$[\lambda_0,\infty) \subset \varrho(-A), \tag{2.2.3}$$

and

$$\|(\lambda I + A)^{-1}\|_{\mathcal{L}(V',V)} \le \alpha^{-1}$$
(2.2.4)

for all $\lambda \geq \lambda_0$.

PROOF. From the Lax–Milgram theorem (see [62, Section VI.3.2.5, Theorem 7]) it follows that $(\lambda I + A)^{-1} \in \mathcal{L}(V', V)$ exists for every $\lambda \ge \lambda_0$. In particular $(\lambda_0 I + A)^{-1} \in \mathcal{L}(V', V')$ since $V \hookrightarrow V'$, so $(\lambda_0 I + A)^{-1}$ is closed on V'. As the inverse of a closed linear operator is closed we get that $\lambda_0 I + A$. Hence A is closed as an operator on V'. Since V is dense in H by assumption, V is dense in V' as well. Also from the above, (2.2.3) is true. Next let $f \in V'$ and $u \in V$ with $Au + \lambda u = f$. If $\lambda \ge \lambda_0$, then

$$\alpha \|u\|_{V}^{2} \leq a(u, u) + \lambda \|u\|_{H}^{2} = \langle f, u \rangle \leq \|f\|_{V'} \|u\|_{V},$$

from which (2.2.4) follows by dividing by $||u||_V$.

We now look at the abstract elliptic equation

$$Au + \lambda_0 u = f \quad \text{in } V', \tag{2.2.5}$$

which is equivalent to the "weak" formulation that u in V with

$$a(u, v) + \lambda_0 \langle u, v \rangle = \langle f, v \rangle \tag{2.2.6}$$

for all $v \in V$. We admit $f \in V'$ in both cases, but note that for $f \in H$, the expression $\langle f, v \rangle$ is the inner product in H. The above theorem tells us that (2.2.5) has a unique solution for every $f \in V'$ whenever $\lambda \ge \lambda_0$. We now summarise the values for λ_0 for the various boundary conditions.

EXAMPLE 2.2.3. (a) If Ω is a bounded domain, or more generally if Ω is an open set lying between two parallel hyperplanes, then (2.1.7) shows that $\|\nabla u\|_2$ is an equivalent norm on $H_0^1(\Omega)$. Hence according to Proposition 2.1.6 we can choose $\lambda_0 := \lambda_A$ in Theorem 2.2.2.

(b) In general, the Neumann problem has a zero eigenvalue. Hence by Proposition 2.1.6 we choose $\lambda_0 = \lambda_A + \alpha_0/2$.

(c) In Example 2.1.4(c) we introduced the equivalent norm (2.1.8). If the boundary coefficient b_0 is bounded from below by a positive constant $\beta > 0$, then by Proposition 2.1.6

$$\|\nabla u\|_{2}^{2} + \|u\|_{L_{2}(\partial\Omega)}^{2} \leq 2 \max\left\{\frac{1}{\alpha_{0}}, \frac{1}{\beta}\right\} \left(\frac{\alpha_{0}}{2} \|\nabla u\|_{2}^{2} + \beta \|u\|_{L_{2}(\partial\Omega)}^{2}\right)$$
$$\leq 2 \max\left\{\frac{1}{\alpha_{0}}, \frac{1}{\beta}\right\} \left(a_{0}(u, u) + \lambda_{\mathcal{A}} \|u\|_{2}^{2} + \int_{\partial\Omega} b_{0}u^{2} d\sigma\right).$$
(2.2.7)

Hence in Theorem 2.2.2 we can choose $\lambda_0 := \lambda_A$ if Ω is a bounded Lipschitz domain and $b_0 \ge \beta$ for some constant $\beta > 0$.

We frequently look at solutions in spaces other than V. To deal with such cases we look at the *maximal restriction* of the operator A to some other Banach E space with $E \hookrightarrow V'$. We let

$$D(A_E) := \{ u \in V \colon Au \in E \}$$

with $A_E u := Au$ for all $u \in D(A_E)$ and call A_E the maximal restriction of A to E.

PROPOSITION 2.2.4. Suppose that A_E is the maximal restriction of A to $E \hookrightarrow V'$ and that $(\lambda I + A)^{-1}(E) \subset E$ for some $\lambda \in \varrho(-A)$. Then A_E is closed and $\varrho(A) \subset \varrho(A_E)$.

PROOF. We first prove A_E is closed. Suppose that $u_n \in D(A_E)$ with $u_n \to u$ and $Au_n \to v$ in E. As $E \hookrightarrow V'$ convergence is also in V'. Because A is closed in V' with domain E we conclude that $u \in V$ and Au = v. We know that $u, v \in E$, so $v \in D(A_E)$ and $A_Eu = v$, proving that A_E is closed. By assumption $(\lambda I + A)^{-1}(E) \subset E$. Because $\lambda I + A_E$ is closed, also its inverse is a closed operator on E. Hence by the closed graph theorem $(\lambda I + A_E)^{-1} \in \mathcal{L}(E)$. In particular, the above argument shows that $\varrho(A) \subset \varrho(A_E)$.

As a special case in the above proposition we can set E := H. Sometimes it is useful to prove properties of the operator A via the associated semigroup it generates on H. A proof of the following proposition can be found in [64, XVII.6, Proposition 3].

PROPOSITION 2.2.5. Under the above assumption, $-A_H$ generates a strongly continuous analytic semigroup e^{-tA_H} on H. Moreover,

$$\|(\lambda I + A)^{-1}\|_{\mathcal{L}(H)} \le \frac{1}{\lambda - \lambda_0}$$

for all $\lambda > \lambda_0$.

PROOF. We only prove the resolvent estimate. If $u = (\lambda I + A)^{-1} f$ for some $\lambda > \lambda_0$ and $f \in H$, then

$$(\lambda - \lambda_0) \|u\|_H^2 \le a(u, u) + \lambda \|u\|_H^2 = \langle f, u \rangle \le \|f\|_H \|u\|_H.$$

If we rearrange the inequality, then the resolvent estimate follows.

2.3. Formally adjoint problems

When working with non-selfadjoint problems it is often necessary to consider the adjoint problem. Suppose that $a(\cdot, \cdot)$ is a bilinear form on a Hilbert space $V \hookrightarrow H$ as in the previous section. We define a new bilinear form

$$a^{\sharp} \colon V \times V \to \mathbb{R}$$

by setting

$$a^{\sharp}(u,v) := a(v,u) \tag{2.3.1}$$

for all $u, v \in V$. If $a(\cdot, \cdot)$ satisfies (2.2.1) and (2.2.2), then clearly $a^{\sharp}(\cdot, \cdot)$ has the same properties with the same constants M, α and λ_0 . We denote the operator induced on V by A^{\sharp} . It is given by

$$a^{\sharp}(u, v) = \langle A^{\sharp}u, v \rangle$$

for all $u, v \in V$. We now relate A^{\sharp} to the dual A' of A.

PROPOSITION 2.3.1. Suppose $a(\cdot, \cdot)$ is a bilinear form satisfying Assumption 2.2.1 and $a^{\sharp}(\cdot, \cdot)$ and A^{\sharp} as defined above. Then $A' = A^{\sharp} \in \mathcal{L}(V, V')$. Moreover, $A'_{H} = A^{\sharp}_{H} \in \mathcal{L}(H, H)$ is the adjoint of the maximal restriction A_{H} . Finally, if $a(\cdot, \cdot)$ is a symmetric form, then A_{H} is self-adjoint.

PROOF. Since every Hilbert space is reflexive V'' = V and so $A' \in \mathcal{L}(V'', V') = \mathcal{L}(V, V')$. Now by definition of A and A^{\sharp} we have

$$\langle Au, v \rangle = a(u, v) = a^{\sharp}(v, u) = \langle A^{\sharp}v, u \rangle$$

for all $u, v \in V$. Next look at the maximal restriction A_H . From the above

$$\langle A_H u, v \rangle = \langle u, A_H^{\sharp} v \rangle$$

for all $u \in D(A_H)$ and all $v \in D(A_H^{\sharp})$, so $A'_H = A^{\sharp}$. Finally, since $A' = A^{\sharp} = A$ if A is a symmetric form, the maximal restrictions A_H and A_H^{\sharp} are the same, so A_H is self-adjoint.

Let us now look at boundary value problems $(\mathcal{A}, \mathcal{B})$ given by (2.1.2) and (2.1.4) satisfying the assumptions made in Section 2.1. Let $a(\cdot, \cdot)$ be the form associated with $(\mathcal{A}, \mathcal{B})$ as in Definition 2.1.2. Then $a^{\sharp}(\cdot, \cdot)$ is the form associated with the *formally adjoint* boundary value problem $(\mathcal{A}^{\sharp}, \mathcal{B}^{\sharp})$ given by

$$\mathcal{A}^{\sharp} = -\operatorname{div}(A_0^T(x)\nabla u + b(x)u) + a(x) \cdot \nabla u + c_0 u$$
(2.3.2)

Table 2.1. Constants in (2.4.1) for Difference problems							
Condition on Ω	Value of d	Value of λ_0	Value of c_a				
$N \ge 3$, any Ω	N	$\lambda_{\mathcal{A}}$	$c(N)/\alpha_0$				
$N = 2$, any Ω	any $d \in (2, \infty)$	$\lambda_{\mathcal{A}} + \alpha_0/2$	$c(d)/\alpha_0$				
$N = 2, \Omega$ between parallel	any $d \in (2, \infty)$	$\lambda_{\mathcal{A}}$	$c(d)(1+D^2)/\alpha_0$				
hyperplanes of distance D							

Table 2.1. Constants in (2.4.1) for Dirichlet problems

and

$$\mathcal{B}^{\sharp} u := \begin{cases} u|_{\Gamma_{1}} & \text{on } \Gamma_{1} \text{ (Dirichlet b.c.),} \\ \frac{\partial u}{\partial \nu_{\mathcal{A}^{\sharp}}} & \text{on } \Gamma_{2} \text{ (Neumann b.c.),} \\ \frac{\partial u}{\partial \nu_{\mathcal{A}^{\sharp}}} + b_{0}u & \text{on } \Gamma_{3} \text{ (Robin b.c.),} \end{cases}$$
(2.3.3)

where

$$\frac{\partial u}{\partial v_{\mathcal{A}^{\sharp}}} := (A_0^T(x)\nabla u + b(x)u) \cdot v.$$

Note that $(\mathcal{A}^{\sharp}, \mathcal{B}^{\sharp})$ has the same structure as $(\mathcal{A}, \mathcal{B})$ with A_0 replaced by its transposed A_0^T and the roles of *a* and *b* interchanged. If *A* and A^{\sharp} are the corresponding operators induced on *V* and *H*, then all the assertions of Proposition 2.3.1 apply.

2.4. Global a priori estimates for weak solutions

In our treatment of domain perturbation problems, global L_p - L_q -estimates for weak solutions to (2.1.1) play an essential role, especially in the nonlinear case with f depending on u. We provide a simple test to obtain such estimates and apply them to the three boundary conditions. The estimates are only based on an embedding theorem for the space V of weak solutions introduced in Section 2.1.

As it turns out, the key to control the domain dependence of L_p -estimates for solutions of (2.1.1) is the constant in a Sobolev-type inequality. More precisely, let $a(\cdot, \cdot)$ be the form associated with the boundary value problem $(\mathcal{A}, \mathcal{B})$ as given in Assumption 2.1.2. We require that there are constants d > 2, $c_a > 0$ and $\lambda_0 \ge 0$ such that

$$\|u\|_{2d/(d-2)}^2 \le c_a(a(u,u) + \lambda_0 \|u\|_2^2)$$
(2.4.1)

for all $u \in V$, where V is the space of weak solutions associated with $(\mathcal{A}, \mathcal{B})$. We always require that V satisfies Assumption 2.1.3. Note that d = N is the smallest possible d because of the optimality of the usual Sobolev inequality in $H_0^1(\Omega)$. To obtain control over domain dependence of the constants d, c_a and λ_0 , we need to choose d larger in some cases. Explicit values for the three boundary conditions are listed in Tables 2.1–2.3. In these tables, the constant $c(\cdot)$ only depends on its argument, α_0 is the ellipticity constant from (2.1.3), and λ_A is given by (2.1.10). Proofs of (2.4.1) for the various cases are given

Table 2.2. Constants in (2.4.1) for Robin problems if $b_0 \ge \beta$ for some $\beta > 0$

Condition on Ω	Value of d	Value of λ_0	Value of c_a
$N \ge 2, \Omega < \infty$	2N	$\lambda_{\mathcal{A}}$	$c(N)(1+ \Omega ^{1/N})\max\{\frac{1}{\alpha_0},\frac{1}{\beta}\}$
$N \ge 2, \Omega \le \infty$	2N	$\lambda_{\mathcal{A}} + \alpha_0/2$	$c(N) \max\{\frac{1}{\alpha_0}, \frac{1}{\beta}\}$

Table 2.3. Constants in (2.4.1) for Neumann problems

Condition on Ω	Value of d	Value of λ_0	Value of c_a
$N \ge 3$, cone condition	Ν	$\lambda_{\mathcal{A}} + \alpha_0/2$	$c(d, \operatorname{cone})/\alpha_0$
N = 2, cone condition	any $d \in (2, \infty)$	$\lambda_{\mathcal{A}} + \alpha_0/2$	$c(d, \operatorname{cone})/\alpha_0$
$N \ge 2$, special class of Ω	$d > 2$ depending on Ω	$\lambda_{\mathcal{A}} + \alpha_0/2$	$c(d, \text{class})/\alpha_0$
No condition on Ω	" $d = \infty$ " no smoothing	N/A	N/A

in Sections 2.4.2 and 2.4.3. It turns out that there are domain-independent estimates and smoothing properties for Dirichlet and Robin boundary conditions, but not for Neumann boundary conditions.

THEOREM 2.4.1. Suppose that $\Omega \subset \mathbb{R}^N$ is an open set and $u \in V$ is a weak solution of (2.1.1). If (2.4.1) holds, then there exists a constant C > 0 only depending on d and $p \ge 2$ such that

$$\|u\|_{dp/(d-2p)} \le c_a C(\|f\|_p + \lambda_0 \|u\|_p)$$
(2.4.2)

if $p \in [2, d/2)$ *, and*

$$\|u\|_{\infty} \le c_a C(\|f\|_p + \lambda_0 \|u\|_p) + \|u\|_p$$
(2.4.3)

or

$$\|u\|_{\infty} \le c_a C(\|f\|_p + \lambda_0 \|u\|_p) + \|u\|_{2d/(d-2)}$$
(2.4.4)

if p > d/2. Moreover, if $\lambda_0 = 0$ or $u \in L_p(\Omega)$ (if $|\Omega| < \infty$ for instance), then the above estimates are valid for $p \in [2d/(d+2), 2)$ as well.

PROOF. Let A be the operator induced by the form $a(\cdot, \cdot)$ on V. Given $u \in V$ we set $u_q := |u|^{q-2}u$ for $q \ge 2$. Assuming that (2.4.1) holds, it follows from [57, Proposition 5.5] that

$$\|u\|_{dq/(d-2)}^q \le c_a \frac{q}{2} (\langle Au, u_q \rangle + (q-1)\lambda_0 \langle u, u_q \rangle)$$

for all $q \ge 2$ and $u \in V$ for which the expression on the right-hand side is finite. Then we apply [57, Theorem 4.5] to get the estimates. Compared to that reference, we have replaced the term $||u||_p$ by $||u||_{2d/(d-2)}$ in (2.4.4). We can do this by replacing q_0 in equation (4.11) in the proof of [57, Theorem 4.5] by $q_0 := 1 + \frac{2d}{p'(d-2)}$ and then complete the proof in a similar way. Finally, the limitation that $p \ge 2$ comes from proving that $u \in L_p(\Omega)$ first if $\lambda_0 \neq 0$. If $\lambda_0 = 0$, or if we know already that $u \in L_p(\Omega)$, then this is not necessary (see also [61, Theorem 2.5]), and we can admit $p \in [2d/(d+2), 2)$.

The above theorem tells us that for $p \ge 2$ and $\lambda \ge \lambda_A$

$$(\lambda I + A)^{-1}$$
: $L_p(\Omega) \cap L_2(\Omega) \to L_{m(p)}(\Omega)$

if we set

$$m(p) := \begin{cases} \frac{dp}{d - 2p} & \text{if } p \in (1, d/2), \\ \infty & \text{if } p > d/2 \end{cases}$$
(2.4.5)

and A is the operator associated with the problem (5.1.2) as constructed in Section 2.2.

We next want to derive domain-independent bounds for the norm of the resolvent operator by constructing an operator in $L_p(\Omega)$ for $p \in (1, \infty)$. Let A_2 denote the maximal restriction of A to $L_2(\Omega)$. By Proposition 2.2.5, the operator $-A_2$ is the generator of a strongly continuous analytic semigroup on $L_2(\Omega)$. Moreover, still assuming that (2.4.1) holds, e^{-tA_2} has an kernel satisfying pointwise Gaussian estimates and therefore interpolates to $L_p(\Omega)$ for all $p \in (1, \infty)$ (see [55]). Denote by $-A_p$ its infinitesimal generator. The dual semigroup is a strongly continuous analytic semigroup on $L_{p'}(\Omega)$ and its generator is A'_p . Let A^{\sharp}_p be the corresponding operators associated with the formally adjoint problem. Since $(A^{\sharp}_2)' = A_2$ by Proposition 2.3.1 we get $(A^{\sharp}_p)' = A_{p'}$. We denote the exponent conjugate to p by p', that is,

$$\frac{1}{p} + \frac{1}{p'} = 1.$$

From [55, Theorem 5.1]

$$||e^{-tA_p}||_{\mathcal{L}(L_p)} \le e^{\omega_p t}, \qquad \omega_p := \max\{p-1, p'-1\}\lambda_0$$
 (2.4.6)

for all t > 0 and $p \in (1, \infty)$.

Solutions of the abstract equation $(\lambda I + A_p)u = f$ with $f \in L_p(\Omega)$ are called generalised solutions of the corresponding elliptic problem in $L_p(\Omega)$. If 1 they are not weak solutions in general, but solutions in an even weaker $sense. Note that we defined <math>A_p$ by means of semigroup theory to be able to easily get a definition for domains with unbounded measure, because for such domains we cannot expect $L_p(\Omega) \subset V'$ for p > 2.

THEOREM 2.4.2. Let A_p as defined above and $p \in (1, \infty)$. Then $(\omega_p, \infty) \subset \varrho(-A_p)$ with $\omega_p := \max\{p-1, p'-1\}\lambda_0$, and

$$\|(\lambda I + A_p)^{-1}\|_{\mathcal{L}(L_p)} \le \frac{1}{\lambda - \omega_p}$$
(2.4.7)

for all $\lambda > \omega_p$. Furthermore, for every p > 1, $p \neq N/2$, there exists a constant C > 0 only depending on d, p and c_a and λ , λ_0 such that

$$\|(\lambda I + A_p)^{-1}\|_{\mathcal{L}(L_p, L_m(p))} \le C$$
(2.4.8)

for all $\lambda > \lambda_0$. If Ω is bounded, then $(\lambda I + A)^{-1}$: $L_p(\Omega) \to L_q(\Omega)$ is compact for all $q \in [p, m(p))$. Finally, if A_p^{\sharp} is the operator associated with the formally adjoint problem, then $A'_p = A_{p'}^{\sharp}$.

PROOF. As $-A_p$ generates a strongly continuous semigroup, (2.4.6) implies that $(\omega_p, \infty) \subset \varrho(-A_p)$ and that

$$(\lambda I + A_p)^{-1} = \int_0^\infty e^{-tA_p} e^{-\lambda t} dt$$

for all $\lambda > \omega_p$ (see [125, Section IX.4]). Hence (2.4.7) follows if we take into account (2.4.6). Now for $f \in L_2(\Omega) \cap L_p(\Omega)$ we have $u := (\lambda I + A_p)^{-1} \in L_p(\Omega)$. Hence, if $2d/(d+2) \le p < d/2$, then by (2.4.2)

$$\|u\|_{dp/(d-2p)} \le c_a C(\|f+\lambda u\|_p+\lambda_0\|u\|_p) \le c_a C\left(1+\frac{\lambda+\lambda_0}{\lambda-\omega_p}\right)\|f\|_p.$$

A similar estimate is obtained by using (2.4.3) if p > d/2. Now (2.4.8) follows since $L_2(\Omega) \cap L_p(\Omega)$ is dense in $L_p(\Omega)$ if we choose C appropriately. If 1 , then we use a duality argument. In that case <math>q := m(p)' > 2d/(d-2) and a simple calculation reveals that p = m(q)'. Because

$$(\lambda I + A_q^{\sharp})^{-1} \in \mathcal{L}(L_q(\mathbb{R}^N), L_{m(q)}(\mathbb{R}^N)),$$

by duality

$$((\lambda I + A_q^{\sharp})^{-1})' = (\lambda I + (A_q^{\sharp})')^{-1} = (\lambda I + A_p)^{-1} \in \mathcal{L}(L_p(\mathbb{R}^N), L_{m(p)}(\mathbb{R}^N))$$

with equal norm. Compactness of the resolvent on $L_p(\Omega)$ for $1 follows from [57, Section 7]. Now compactness as an operator from <math>L_p(\Omega)$ to $L_q(\Omega)$ for $q \in [p, m(p))$ follows from a compactness property of the Riesz–Thorin interpolation theorem (see [94]).

REMARK 2.4.3. Note that the above theorem is not optimal, but it is sufficient for our purposes. In particular the condition $\lambda > \omega_p$ could be improved by various means. If A has compact resolvent, then the spectrum of A_p is independent of p because the above smoothing properties of the resolvent operator show that every eigenfunction is in $L^{\infty}(\Omega)$. Also if p = N, then the spectrum is independent of p by [95] because of Gaussian bounds for heat kernels (see [55]).

2.4.1. Sobolev inequalities associated with Dirichlet problems

If $N \ge 3$ there exists a constant c(N) only depending on the dimension N such that

$$\|u\|_{2N/(N-2)} \le c(N) \|\nabla u\|_2 \tag{2.4.9}$$

for all $u \in H^1(\mathbb{R}^N)$ (see [76, Theorem 7.10]). If N = 2, then for every $q \in [2, \infty)$ there exists a constant c_q only depending on q such that

 $||u||_q \leq c_q ||u||_{H^1(\mathbb{R}^N)}$

for all $u \in H^1(\mathbb{R}^2)$ (see [96, Theorem 8.5]). If $q \in (2, \infty)$ and d := 2q/(q-2), then q = 2d/(d-2). Hence for every d > 2 there exists $c_d > 0$ only depending on d such that

$$\|u\|_{2d/(d-2)} \le c_d \|u\|_{H^1(\mathbb{R}^N)}.$$
(2.4.10)

If Ω is lying between two parallel hyperplanes of distance *D*, then using (2.1.7) we conclude that

$$\|u\|_{2d/(d-2)} \le c_d \|u\|_{H^1(\mathbb{R}^N)} \le c_d \sqrt{1+D^2} \|\nabla u\|_2$$
(2.4.11)

for all $u \in H_0^1(\Omega)$. Combining the above with the basic inequalities in Proposition 2.1.6 we can summarise the constants appearing in (2.4.1) in Table 2.1.

The L_{∞} -estimates for Dirichlet problems are very well known, see for instance [76, Chapter 8]. The estimates for p < N/2 are not as widely known, and sometimes stated with additional assumptions on the structure of the operators, see [38, Appendix to Chapter 3], [112, Theorem 4.2] or without proof in [45, Lemma 1]. A complete proof is contained in [57].

2.4.2. Maz'ya's inequality and Robin problems

It may be surprising that solutions of the elliptic problem with Robin boundary conditions

$$\mathcal{A}u = f \quad \text{in } \Omega,$$

$$\frac{\partial u}{\partial v_{\mathcal{A}}} + b_0 u = 0 \quad \text{on } \partial \Omega$$
(2.4.12)

with $b_0 \ge \beta$ for some constant $\beta > 0$ satisfy domain-independent estimates similar to the ones for problems with Dirichlet boundary conditions. The estimates were first established in [56]. In the present discussion we will only work with bounded Lipschitz domains $\Omega \subset \mathbb{R}^N$, but note that the result could be generalised to arbitrary domains. We refer to [11,56] for details.

The weak solutions of the Robin problem on a Lipschitz domain are in $H^1(\Omega)$ as discussed in Example 2.1.4(c). In the usual Sobolev inequality $||u||_{2N/(N-2)} \le c ||u||_{H^1}$, the constant *c* depends on the shape of the domain as examples of domains with a cusp show (see [2, Theorem 5.35]). Hence only a weaker statement can be true. The key is an inequality due to Maz'ya from [100] (see [101, Section 3.6]) stating that

$$\|u\|_{N/(N-1)} \le c(N)(\|\nabla u\|_1 + \|u\|_{L_1(\partial\Omega)})$$

for all $u \in W_1^1(\Omega) \cap C(\overline{\Omega})$, where c(N) is the isoperimetric constant depending only on $N \ge 2$. Substituting u^2 into the above inequality we get

$$\|u\|_{2N/(N-1)}^{2} \leq c(N)(2\|u\nabla u\|_{1} + \|u\|_{L_{2}(\partial\Omega)}^{2})$$

$$\leq c(N)(\|u\|_{H^{1}}^{2} + \|u\|_{L_{2}(\partial\Omega)}^{2}).$$
(2.4.13)

By the density of $H^1(\Omega) \cap C(\overline{\Omega})$ in $H^1(\Omega)$ the inequality is valid for all $u \in H^1(\Omega)$. If Ω has finite measure, then similarly

$$\|u\|_{2N/(N-1)}^2 \le c(N)(1+|\Omega)|^{1/N})(\|\nabla u\|_2^2 + \|u\|_{L_2(\partial\Omega)}^2)$$
(2.4.14)

for all $u \in H^1(\Omega)$ with a constant c(N) different from the original one, but only depending on N. Combining (2.2.7) with (2.4.13) or (2.4.14) we therefore get (2.4.1) with d = 2Nas displayed in Table 2.2.

2.4.3. Sobolev inequalities associated with Neumann problems

The smoothing properties for Dirichlet and Robin problems we established in the previous sections were based on the validity of a Sobolev-type inequality for functions in the space of weak solutions with a constant independent of the shape of the domain. The space of weak solutions for the Neumann problem

$$\mathcal{A}u = f \quad \text{in } \Omega,$$
$$\frac{\partial u}{\partial v_{\mathcal{A}}} = 0 \quad \text{on } \partial \Omega$$

is $H^1(\Omega)$. In the case of the Robin problem we could use an equivalent norm on $H^1(\Omega)$ involving a boundary integral. As pointed out in Section 2.4.2, the boundary integral is of higher order than the H^1 -norm if the domain is bad. Hence for the Neumann problem the constant *c* in the Sobolev inequality

$$\|u\|_{2N/(N-2)} \le c \|u\|_{H^1} \tag{2.4.15}$$

depends on the shape and not just the measure of Ω . There are no easy necessary and sufficient conditions for the inequality to be true. A sufficient condition is that Ω satisfies an (interior) cone condition, that is, there exists an open cone $C \subset \mathbb{R}^N$ with vertex at zero such that for every $x \in \partial \Omega$ there is an orthogonal transformation T such that $x+T(C) \subset \Omega$ (see [2, Definition 4.3]). The constant c in (2.4.15) depends on the length and the angle of the cone C (see [2, Lemma 5.12]). A cone condition is however not necessary for getting a Sobolev inequality uniformly with respect to a family of domains. Shrinking holes of fixed shape to a point is sufficient (see [53, Section 2]). Alternatively an extension property is also sufficient. This includes domains with fractal boundary (quasi-disks) as shown in [101, Section 1.5.1]. We could replace the cone C by a standard polynomial cusp and get an inequality of the form

$$\|u\|_{2d/(d-2)} \le c \|u\|_{H^1} \tag{2.4.16}$$

with $d \ge N$ depending on the sharpness of the cusp (see [2, Theorem 5.35] or [101, Section 4.4]). For general domains there is no such d > 2, and there are no smoothing properties of the resolvent operator. This corresponds to the degenerate case " $d = \infty$ " because $2d/(d-2) \rightarrow 2$ as $d \rightarrow \infty$. Combining Proposition (2.1.11) with the above we get (2.4.1) with constants as displayed in Table 2.3.

2.5. The pseudo-resolvent associated with boundary value problems

When dealing with varying domains we want to embed our problem into a fixed large space. In this section we want to explain how to do that. We define the inclusion $i_{\Omega}: L_p(\Omega) \to L_p(\mathbb{R}^N)$ to be the trivial extension

$$i_{\Omega}(u) := \begin{cases} u & \text{on } \Omega, \\ 0 & \text{on } \Omega^{c}. \end{cases}$$
(2.5.1)

We also sometimes write $\tilde{u} := i_{\Omega}(u)$ for the trivial extension. The above extension operator also acts as an operator

$$i_{\Omega} \colon H_0^1(\Omega) \to H^1(\mathbb{R}^N).$$

Indeed, by definition $C_c^{\infty}(\Omega)$ is dense in $H_0^1(\Omega)$. We can identify $C_c^{\infty}(\Omega)$ with the set $\{u \in C_c^{\infty}(\mathbb{R}^N) : \sup p u \subset \Omega\}$, and view $H_0^1(\Omega)$ as the closure of $C_c^{\infty}(\Omega)$ in $H^1(\mathbb{R}^N)$.

Furthermore, we let $r_{\Omega} \colon L_p(\mathbb{R}^N) \to L_p(\Omega)$ be the restriction

$$r_{\Omega}(u) := u|_{\Omega}. \tag{2.5.2}$$

If $f \in H^{-1}(\mathbb{R}^N)$, then we restrict the functional to the closed subspace $H_0^1(\Omega)$ of $H^1(\mathbb{R}^N)$ to get an element of $H^{-1}(\Omega)$. More formally we define the restriction operator $r_{\Omega} \colon H^{-1}(\mathbb{R}^N) \to H^{-1}(\Omega)$ by

$$r_{\Omega}(f) := f|_{H_0^1(\Omega)}.$$
(2.5.3)

On the subspace $L_2(\mathbb{R}^N)$ of $H^{-1}(\mathbb{R}^N)$, the two definitions coincide. We next prove that the operators i_{Ω} and r_{Ω} are dual to each other.

LEMMA 2.5.1. Let $\Omega \subset \mathbb{R}^N$ be open. Let i_Ω and r_Ω as defined above and 1 .Then

$$i_{\Omega} \in \mathcal{L}(L_p(\Omega), L_p(\mathbb{R}^N)) \cap \mathcal{L}(H_0^1(\Omega), H^1(\mathbb{R}^N))$$

and $||i_{\Omega}|| = ||r_{\Omega}|| = 1$. Moreover

$$i'_{\Omega} = r_{\Omega} \in \mathcal{L}(L_{p'}(\mathbb{R}^N), L_{p'}(\Omega)) \cap \mathcal{L}(H^{-1}(\mathbb{R}^N), H^{-1}(\Omega))$$

and $r'_{\Omega} = i_{\Omega}$, where p' is the conjugate exponent to p.

PROOF. The first assertion follows directly from the definition of the operators. If $f \in H^{-1}(\mathbb{R}^N)$ or $L_{p'}(\mathbb{R}^N)$, then by definition of i_{Ω} and r_{Ω}

$$\langle f, i_{\Omega}(u) \rangle = \langle r_{\Omega}(f), u \rangle$$

for all $u \in C_c^{\infty}(\Omega)$. By density of $C_c^{\infty}(\Omega)$ in $H_0^1(\Omega)$ and $L_p(\Omega)$ we get $i'_{\Omega} = r_{\Omega}$. By the reflexivity of the spaces involved we therefore also have $r'_{\Omega} = i_{\Omega}$.

Let $(\mathcal{A}_n, \mathcal{B}_n)$ and $(\mathcal{A}, \mathcal{B})$ be elliptic boundary value problems on open sets Ω_n and Ω , respectively. Suppose that A_n and A are the corresponding operators induced as discussed in Section 2.2. We can then embed the problems in \mathbb{R}^N as follows.

DEFINITION 2.5.2. Let A_n , A defined as above. We set

$$R_n(\lambda) := i_{\Omega_n}(\lambda I + A_n)^{-1} r_{\Omega_n}$$
 and $R(\lambda) := i_{\Omega}(\lambda I + A)^{-1} r_{\Omega}$

whenever the inverse operators exist. Similarly we define $R_n^{\sharp}(\lambda)$ and $R^{\sharp}(\lambda)$ for the formally adjoint problem.

The family of operators $R(\lambda)$ and $R_n(\lambda)$ form a pseudo-resolvent as defined for instance in [125, Section VIII.4]. In particular they satisfy the resolvent equation

$$R(\lambda) - R(\mu) = (\mu - \lambda)R(\lambda)R(\mu)$$

for all $\lambda, \mu \in \varrho(-A)$. Using Lemma 2.5.1 and the last assertion in Theorem 2.4.2 we get the following properties of the pseudo-resolvent.

LEMMA 2.5.3. If $\lambda \in \rho(A)$, then $R(\lambda)' = R^{\sharp}(\lambda)$.

REMARK 2.5.4. From the above it also follows that we can replace $(\lambda + A)^{-1}$ by $R(\lambda)$ in (2.4.7) and (2.4.8) with all constants being the same.

3. Semi-linear elliptic problems

The purpose of this section is to formulate semi-linear boundary value problems as a fixed point equation in $L_p(\mathbb{R}^N)$. Then we derive an L_∞ -estimate for solutions in terms of an L_p -norm, provided the nonlinearity satisfies a growth condition.

3.1. Abstract formulation of semi-linear problems

We now want to look at properties of weak solutions of the semi-linear boundary value problem

$$\mathcal{A}u = f(x, u(x)) \quad \text{in } \Omega, \mathcal{B}u = 0 \qquad \text{on } \partial\Omega,$$
(3.1.1)

where $(\mathcal{A}, \mathcal{B})$ are as discussed in Section 2.1. We use the smoothing properties from Section 2.4 to show that under suitable growth conditions on f, the boundary value problem (3.1.1) can be viewed as a fixed point equation in $L_p(\mathbb{R}^N)$ for a some range of $p \in (1, \infty)$.

We assume that $V \subset H^1(\Omega)$ is the space of weak solutions for the boundary conditions under consideration as introduced in Section 2.1 and discussed in detail for different boundary conditions in Section 2.4. We also assume that $f: \Omega \times \mathbb{R} \to \mathbb{R}$ is a function with properties to be specified. Given $u: \Omega \to \mathbb{R}$ we define the *superposition operator* F(u) by

$$F(u)(x) := f(x, u(x))$$
(3.1.2)

for all $x \in \Omega$, provided that $F(u) \in V'$. We call $u \in V$ a weak solution of (3.1.1) if

$$a(u, v) = \langle F(u), v \rangle$$

for all $v \in V$. Here $a(\cdot, \cdot)$ is the form associated with $(\mathcal{A}, \mathcal{B})$ as in Definition 2.1.2. If A is the operator induced by $(\mathcal{A}, \mathcal{B})$ we can rewrite (3.1.1) as

$$Au = F(u). \tag{3.1.3}$$

To be able to show that $F(u) \in V'$ we need to make some assumptions on f.

ASSUMPTION 3.1.1. Suppose that $f: \Omega \times \mathbb{R} \to \mathbb{R}$ is a Carathéodory function, that is, $f(\cdot, \xi): \Omega \to \mathbb{R}$ is measurable for all $\xi \in \mathbb{R}$, and $f(x, \cdot) \in C(\mathbb{R})$ for almost all $x \in \Omega$. Further suppose that there exist a function $g \in L_1(\Omega) \cap L_\infty(\Omega)$ and constants $1 \le \gamma < \infty$ and $c \ge 0$ such that

$$|f(x,\xi)| \le g(x) + c|\xi|^{\gamma}$$
(3.1.4)

for all $(x, \xi) \in \Omega \times \mathbb{R}$.

The above growth conditions on f lead to the following mapping properties of the superposition operator.

LEMMA 3.1.2. Suppose that f satisfies Assumption 3.1.1 with $1 \le \gamma \le p$. Then the corresponding superposition operator F is in $C(L_p(\Omega), L_{p/\gamma}(\Omega))$. Moreover,

$$||F(u)||_{p/\gamma} \le ||g||_1 + ||g||_{\infty} + c||u||_p^{\gamma}$$

for all $u \in L_p(\Omega)$.

PROOF. By [6, Theorem 3.1] we have

$$\|F(u)\|_{p/\gamma} \le \|g\|_{p/\gamma} + c\|u\|_p^{\gamma} \le \|g\|_1^{\gamma/p} \|g\|_{\infty}^{1-\gamma/p} + c\|u\|_p$$

Then use Young's inequality to get

$$\|g\|_{1}^{\gamma/p}\|g\|_{\infty}^{1-\gamma/p} \leq \frac{\gamma}{p}\|g\|_{1} + \left(1 - \frac{\gamma}{p}\right)\|g\|_{\infty} \leq \|g\|_{1} + \|g\|_{\infty}.$$

Continuity is proved in [6, Theorem 3.7].

Note that [6, Theorem 3.1] shows that the growth conditions are necessary and sufficient for *F* to map $L_p(\Omega)$ into $L_{p/\gamma}(\Omega)$. In particular, if *F* maps $L_p(\Omega)$ into itself and $p \in (1, \infty)$, then $\gamma = 1$, that is, *f* grows at most linearly.

From now on we assume that (2.4.1) is true for all $v \in V$. Then by Theorem 2.4.2

$$(\lambda I + A)^{-1} \in \mathcal{L}(L_p(\Omega)) \cap \mathcal{L}(L_p(\Omega), L_{m(p)}(\Omega))$$

for all $\lambda \in \varrho(-A)$ with m(p) given by (2.4.5). If we fix $\lambda \in \varrho(-A)$, then we can rewrite (3.1.3) as $Au + \lambda u = F(u) + \lambda u$ and hence in form of the fixed point equation

 $u = (\lambda I + A)^{-1} (F(u) + \lambda u).$

To be able to consider this as an equation in $L_p(\Omega)$ we need that the right-hand side is in $L_p(\Omega)$ if $u \in L_p(\Omega)$. Taking into account Lemma 3.1.2 and the smoothing property of the resolvent we need that $m(p/\gamma) \ge p$. When looking at boundedness of weak solutions and convergence properties with respect to the domain it is necessary to require $m(p/\gamma) > p$, or equivalently,

$$1 \le \gamma < 1 + \frac{2p}{d}.\tag{3.1.5}$$

Because every weak solution of (3.1.1) lies in $L_{2d/(d-2)}(\Omega)$ we can assume that $p \ge 2d/(d-2)$. Then automatically $1 \le \gamma \le p$ as required in Lemma 3.1.2.

PROPOSITION 3.1.3. Suppose that $(\mathcal{A}, \mathcal{B})$ is such that (2.4.1) holds for some d > 2. Moreover, let $2d/(d-2) \leq p < \infty$ such that Assumption 3.1.1 holds with γ satisfying (3.1.5). Fix $\lambda \in \varrho(-A)$ and set

$$G(u) := (\lambda I + A)^{-1} (F(u) + \lambda u).$$

Then $G \in C(L_p(\Omega), L_p(\Omega))$ and

$$\|G(u)\|_{p} \leq \|R(\lambda)\|_{\mathcal{L}(L_{p/\gamma}, L_{p})}(\|g\|_{1} + \|g\|_{\infty} + c\|u\|_{p}^{\gamma}) + |\lambda|\|R(\lambda)\|_{\mathcal{L}(L_{p})}\|u\|_{p}$$
(3.1.6)

for all $u \in L_p(\Omega)$. Furthermore, if Ω is bounded, then G is compact, that is, G maps bounded sets of $L_p(\Omega)$ onto relatively compact sets of $L_p(\Omega)$. Finally, $u \in L_p(\Omega) \cap V$ is a weak solution of (3.1.1) if and only if u is a fixed point of

$$u = G(u)$$

in $L_p(\Omega)$.

PROOF. By Lemma 3.1.2, $F \in C(L_p(\Omega), L_{p/\gamma}(\Omega))$ is bounded with

 $\|F(u)\|_{p/\gamma} \le \|g\|_1 + \|g\|_{\infty} + c\|u\|_p^{\gamma},$

so *F* is bounded. Clearly (3.1.5) implies $m(p/\gamma) > p$, so Theorem 2.4.2 shows that $R(\lambda) \in \mathcal{L}(L_{p/\gamma}(\Omega), L_p(\Omega)) \cap \mathcal{L}(L_p(\Omega))$ with the operator being compact if Ω is bounded. Hence *G* is continuous as claimed and compact if Ω is bounded. From the definition of *G*

$$\|G(u)\|_{p} \leq \|R(\lambda)\|_{\mathcal{L}(L_{p/\gamma},L_{p})}\|F(u)\|_{p/\gamma} + |\lambda|\|R(\lambda)\|_{\mathcal{L}(L_{p})}\|u\|_{p}.$$

Combining it with the estimate of $||F(u)||_{p/\gamma}$ from above we obtain (3.1.6). The last assertion is evident from the definition of *G*.

REMARK 3.1.4. Because every weak solution lies in $L_{2d/(d-2)}(\Omega)$ the above condition is automatically satisfied if $\gamma < (d+2)/(d-2)$, that is, the growth is subcritical for the exponent *d*.

3.2. Boundedness of weak solutions

We apply results from Section 2.4 to show that weak solutions of (2.1.1) are in L_{∞} if they are in $L_p(\mathbb{R}^N)$ and the nonlinearity satisfies a growth condition.

THEOREM 3.2.1. Suppose that $(\mathcal{A}, \mathcal{B})$ is such that (2.4.1) holds for some d > 2 and $\lambda_0 \ge 0$. Moreover, let $p \ge 2d/(d-2)$ such that Assumption 3.1.1 holds with γ satisfying (3.1.5). Suppose that $u \in V \cap L_p(\Omega)$ is a weak solution of (3.1.1). If $\lambda_0 = 0$, then $u \in L_{\infty}(\Omega)$ and there exists an increasing function $q : [0, \infty) \to [0, \infty)$ such that

 $\|u\|_{\infty} \leq q(\|u\|_p).$

That function only depends on γ , p, an upper bound for $||g||_1 + ||g||_{\infty}$ and c from Assumption 3.1.1 and the constants c_a , C from Theorem 2.4.1.

If $\lambda_0 > 0$ and in addition $u \in L_{p/\gamma}(\Omega)$, then $u \in L_{\infty}(\Omega)$ and

$$||u||_{\infty} \leq q(||u||_{p}, ||u||_{p/\gamma})$$

with the function q also depending on λ_0 .

PROOF. Suppose that $u \in L_p(\Omega)$ is a solution of (3.1.1) with p and γ satisfying (3.1.5). We set

 $p_{k+1} := m(p_k/\gamma)$ and $p_0 := p$

and note that $p_0/\gamma = p/\gamma \ge 2d/(d+2)$. Using (2.4.2) with $\lambda_0 = 0$ and Lemma 3.1.2

$$\|u\|_{p_{k+1}} \le c_a C \|F(u)\|_{p_k/\gamma} \le c_a C (\|g\|_1 + \|g\|_{\infty} + \|u\|_{p_k}^{\gamma})$$
(3.2.1)

as long as $p_k < \gamma d/2$. From (3.1.5) the sequence (p_k) is increasing. Hence, again using (3.1.5) we get

$$p_{k+1} - p_k = \frac{dp_k}{d\gamma - 2p_k} - p_k = \left(\frac{d}{d\gamma - 2p_k} - 1\right) p_k$$
$$\geq \left(\frac{d}{d\gamma - 2p} - 1\right) p > 0$$

as long as $p_k/\gamma < d/2$. Therefore we can choose $m \in \mathbb{N}$ such that $p_m < \gamma d/2 < p_{m+1}$. Then by (2.4.2) and Lemma 3.1.2

$$\begin{aligned} \|u\|_{\infty} &\leq c_a C \|F(u)\|_{p_{m+1}/\gamma} + \|u\|_{p_{m+1}} \\ &\leq c_a C (\|g\|_1 + \|g\|_{\infty} + \|u\|_{p_{m+1}}^{\gamma}) + \|u\|_{p_{m+1}} \end{aligned}$$
(3.2.2)

and we are done. We now obtain the L_{∞} -bound for *u* by applying (3.2.1) inductively to k = 0, ..., m, and finally using (3.2.2). It is now obvious how to define the function *q* having the required properties. If $\lambda_0 > 0$, then (3.2.1) has to be replaced by

$$||u||_{p_{k+1}} \leq c_a C(||F(u)||_{p_k/\gamma} + \lambda_0 ||u||_{p_k/\gamma})$$

$$\leq c_a C(||g||_1 + ||g||_{\infty} + ||u||_{p_k}^{\gamma} + \lambda_0 ||u||_{p_k/\gamma}).$$

Now the assertion follows in a similar manner as in the case $\lambda_0 = 0$.

REMARK 3.2.2. On domains with finite measure we often work with (3.1.4), where g is a constant. In that case dependence on $||g||_1 = g|\Omega|$ means dependence on the measure of the domain and the magnitude of g. Moreover, if Ω has finite measure, then the condition $u \in L_{p/\gamma}(\Omega)$ is automatically satisfied because $p/\gamma \leq p$.

4. Abstract results on linear operators

Many convergence properties of the resolvents reduce to an abstract perturbation theorem. We collect these results here. The first is a characterisation of convergence in the operator norm if the limit is compact. Then we discuss a spectral mapping theorem and how to apply it to get continuity of the spectrum and the corresponding projections. Finally we use an interpolation argument to extend convergence in $L_p(\mathbb{R}^N)$ for some p to all p.

4.1. Convergence in the operator norm

The aim of this section is to prove a characterisation of convergence in the operator norm useful in the context of domain perturbations. Recall that a sequence of operators (T_n) on a Banach space *E* is called *strongly convergent* if $T_n f \to Tf$ for all $f \in E$.

PROPOSITION 4.1.1. Suppose that E, F are Banach spaces, E is reflexive and that $T_n, T \in \mathcal{L}(E, F)$. Then the following assertions are equivalent.

- (1) *T* is compact and $T_n \to T$ in $\mathcal{L}(E, F)$;
- (2) $T_n f_n \rightarrow Tf$ in F whenever $f_n \rightharpoonup f$ weakly in E;
- (3) $T_n \rightarrow T$ strongly and $T_n f_n \rightarrow 0$ in F whenever $f_n \rightarrow 0$ weakly in E.

PROOF. We first prove that (1) implies (2). Assuming that $f_n \rightarrow f$ weakly in E we have

$$||T_n f_n - Tf||_F \le ||T_n - T||_{\mathcal{L}(E,F)} ||f_n||_E + ||T(f_n - f)||_F$$

The first term on the right-hand side converges to zero because $T_n \to T$ in $\mathcal{L}(E, F)$ by assumption and weakly convergent sequences are bounded. By compactness of T and since $f_n - f \rightarrow 0$ weakly in E, also the second term converges to zero, proving (2).

Clearly (2) implies (3) so it remains to prove that (3) implies (1). We start by showing that *T* is compact. Because *E* is reflexive we only need to show that $Tf_n \rightarrow 0$ in *F* whenever $f_n \rightarrow 0$ weakly in *E* (see [39, Proposition VI.3.3]). Assume now that $f_n \rightarrow 0$ weakly in *E*. Clearly (2) in particular shows that $T_n \rightarrow T$ strongly, so $T_k f_n \rightarrow T f_n$ as $k \rightarrow \infty$ for every fixed $n \in \mathbb{N}$. Hence for every $n \in \mathbb{N}$ there exists $k_n \geq n$ such that $||T_{k_n} f_n - Tf_n||_F \leq 1/n$, and thus

$$\|Tf_n\|_F \le \|Tf_n - T_{k_n} f_n\|_F + \|T_{k_n} (f_n - f_{k_n})\|_F + \|T_{k_n} f_{k_n}\|_F$$

$$\le \frac{1}{n} + \|T_{k_n} (f_n - f_{k_n})\|_F + \|T_{k_n} f_{k_n}\|_F.$$

By assumption $||T_{k_n} f_{k_n}||_F \to 0$ as $n \to \infty$ since $f_n \to 0$ and likewise $||T_{k_n} (f_n - f_{k_n})||_F \to 0$ as $n \to \infty$ because $f_n - f_{k_n} \to 0$. Hence the right-hand side of the above inequality converges to zero as $n \to \infty$, so $Tf_n \to 0$ in F and thus T is compact.

To prove that T_n converges in $\mathcal{L}(E, F)$, we assume to the contrary that this is not the case and derive a contradiction. Then there exist $\varepsilon > 0$ and $f_n \in E$ with $||f_n|| = 1$ such that $\varepsilon \leq ||T_n f_n - Tf_n||_F$ for all $n \in \mathbb{N}$. As bounded sets in a reflexive space are weakly sequentially compact there exists a subsequence (f_{n_k}) such that $f_{n_k} \rightharpoonup f$ weakly in E. Therefore

$$0 < \varepsilon \le \|T_{n_k} f_{n_k} - T f_{n_k}\|_F$$

$$\le \|T_{n_k} (f_{n_k} - f)\|_F + \|T_{n_k} f - T f\|_F + \|T (f - f_{n_k})\|_F.$$
(4.1.1)

The first term converges to zero by assumption as $f_{n_k} - f \rightarrow 0$ weakly in *E*. The second term converges to zero as $T_n \rightarrow T$ strongly, and the last term converges to zero as *T* is compact and $f - f_{n_k} \rightarrow 0$ weakly in *E*. However, this contradicts (4.1.1), showing that T_n must converge in $\mathcal{L}(E, F)$. Hence (1) holds, completing the proof of the proposition. \Box

Note that we do not require the T_n to be compact. Hence the sequence of operators is not necessarily collectively compact as in [5].

4.2. A spectral mapping theorem

When looking at domain perturbation problems we embedded the problems involved into one single space by making use of inclusions and restrictions as introduced in Section 2.5. The purpose of this section is to show that we can still apply the standard perturbation theory of linear operators to show continuity of the spectrum and the corresponding projections.

Suppose that *E*, *F* are Banach spaces, and that *A* is a closed densely defined operator on *F* with domain D(A). Moreover, suppose that there exist $i \in \mathcal{L}(F, E)$ and $r \in \mathcal{L}(E, F)$ such that $ri = I_F$. For $\lambda \in \varrho(A)$ we consider the pseudo-resolvent

$$R(\lambda) := i(\lambda I - A)^{-1}r.$$

We then have the following spectral mapping theorem. A pseudo-resolvent is a family of linear operators satisfying the resolvent equation

$$R(\lambda) - R(\mu) = (\mu - \lambda)R(\lambda)R(\mu)$$

for λ , μ in some open subset of \mathbb{C} .

PROPOSITION 4.2.1. Suppose that $\lambda \in \varrho(A)$, and that $\mu \neq \lambda$. Then $\mu \in \varrho(A)$ if and only if $(\mu - \lambda)^{-1} \in \varrho(R(\lambda))$. If that is the case, then

$$R(\mu) = \frac{1}{\mu - \lambda} R(\lambda) \left(\frac{1}{\mu - \lambda} I_E - R(\lambda) \right)^{-1}.$$
(4.2.1)

PROOF. Replacing A by $\lambda I_F - A$ we can assume without loss of generality that $\lambda = 0$ and thus $A^{-1} \in \mathcal{L}(F)$. Now $0 \neq \mu \in \varrho(A)$ if and only if $1/\mu \in \varrho(A^{-1})$ (see [92, Theorem III.6.15]), so we only need to prove that $1/\mu \in \varrho(R(0))$ if and only if $1/\mu \in \varrho(A^{-1})$. To do so we first split the equation

$$\frac{1}{\mu}u - R(0)u = f \tag{4.2.2}$$

into an equivalent system of equations. Observe that P := ir is a projection. If we set $E_1 := P(E)$ and $E_2 := (I - P)(E)$, then $E = E_1 \oplus E_2$. By construction, the image of R(0) is in E_1 . As r = rP we have PR(0) = R(0)P. Setting $u_1 := Pu$ and $u_2 := (I_E - P)u$, equation (4.2.2) is equivalent to the system

$$\left(\frac{1}{\mu}I_E - R(0)\right)u_1 = Pf,$$
(4.2.3)

$$\frac{1}{\mu}u_2 = (I - P)f. \tag{4.2.4}$$

Assume now that $\mu \in \varrho(A^{-1})$, and fix $f \in E$ arbitrary. It follows that $u_1 := i(\mu^{-1}I_F - A^{-1})^{-1}Pf$ is the unique solution of (4.2.3), and $u_2 := \mu(I - P)f$ is the unique solution of (4.2.4). Hence $u := u_1 + u_2$ is the unique solution of (4.2.2) and the map $f \to (u_1, u_2)$ is continuous, showing that $1/\mu \in \varrho(R(0))$.

Next assume that $1/\mu \in \varrho(R(0))$, and that $g \in F$ is arbitrary. Set f := i(g) and note that Pf = f in that case. By assumption (4.2.3) has a unique solution u_1 . As (I-P)f = 0 the solution of (4.2.4) is zero. Hence $r(u_1)$ is the unique solution of $(\mu^{-1}I - A^{-1})u = g$, showing that $1/\mu \in \varrho(A^{-1})$. We finally prove identity (4.2.1), provided $\lambda, \mu \in \varrho(A)$. By the resolvent equation

$$(\lambda I - A)^{-1} = (\mu I - A)^{-1} (I_F - (\lambda - \mu)(\lambda I - A)^{-1}).$$

Using that $ri = I_F$ this yields

$$R(\lambda) = i(\mu I_F - A)^{-1}(I_F r - (\lambda - \mu)ri(\lambda I - A)^{-1}r)$$

= $i(\mu I_F - A)^{-1}r(I_E - (\lambda - \mu)i(\lambda I - A)^{-1}r)$
= $R(\mu)(I_E - (\lambda - \mu)R(\lambda))$
= $(\lambda - \mu)R(\mu)\left(\frac{1}{\lambda - \mu}I_E - R(\lambda)\right).$

As we know that $(\lambda - \mu)^{-1} \in \rho(R(\lambda))$, identity (4.2.1) follows by rearranging the above equation.

4.3. Convergence properties of resolvent and spectrum

We consider a situation similar to the one in Section 4.2, but with a sequence of closed operators A_n defined on Banach spaces F_n with domains $D(A_n)$. Moreover suppose that there exist a Banach space E and operators $i_n \in \mathcal{L}(F_n, E)$ and $r \in \mathcal{L}(E, F_n)$ such that $r_n i_n = I_{F_n}$. We also deal with a limit problem involving a closed densely defined operator A on a Banach space F. For $\lambda \in \varrho(A_n) \cap \varrho(A)$ we consider the *pseudo-resolvents*

$$R_n(\lambda) := i_n (\lambda I - A_n)^{-1} r_n$$
 and $R(\lambda) := i (\lambda I - A)^{-1} r_n$

similarly as in the concrete case of boundary value problems in Section 2.5. We then have the following theorem about convergence of the pseudo-resolvents.

THEOREM 4.3.1. Suppose that $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(E)$ for some $\lambda \in \mathbb{C}$. Then, for every $\mu \in \varrho(A)$ we have $\mu \in \varrho(A_n)$ for $n \in \mathbb{N}$ large enough, and $R_n(\mu) \to R(\mu)$ in $\mathcal{L}(E)$.

PROOF. Suppose that $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(E)$ for some $\lambda \in \mathbb{C}$, and that $\mu \in \varrho(A)$. By Proposition 4.2.1 we have $(\mu - \lambda)^{-1} \in \varrho(-R(\lambda))$ and so [92, Theorem IV.2.25] implies that $(\mu - \lambda)^{-1} \in \varrho(-R_n(\lambda))$ if only *n* is large enough. Applying Proposition 4.2.1 again we see that $\mu \in \varrho(A_n)$ if *n* is large enough. Using (4.2.1) we get

$$\lim_{n \to \infty} R_n(\mu) = \lim_{n \to \infty} \frac{1}{\mu - \lambda} R_n(\lambda) \left(\frac{1}{\mu - \lambda} I_E - R_n(\lambda) \right)^{-1}$$
$$= \frac{1}{\mu - \lambda} R(\lambda) \left(\frac{1}{\mu - \lambda} I_E - R(\lambda) \right)^{-1} = R(\mu)$$

in L(E). Here we use that the map $T \mapsto (\alpha I - T)^{-1}$ is continuous as a map from $\mathcal{L}(E)$ into itself if $\alpha \in \varrho(T)$ (see [121, Theorem IV.1.5]). This completes the proof of the theorem.

From the above we get the upper semi-continuity of separated parts of the spectrum and in particular the continuity of every finite system of eigenvalues. Recall that a spectral set is a subset of the spectrum which is open and closed in the spectrum. To every spectral set we can consider the corresponding spectral projection (see [92, Section III.6.4]). The following properties of the spectral projections immediately follow from [92, Theorem IV.3.16] and Proposition 4.2.1.

COROLLARY 4.3.2. Suppose that $R_n(\lambda) \to R(\lambda)$ in L(E) for some $\lambda \in \mathbb{C}$, that $\Sigma \subset \sigma(-A_{\Omega}) \subset \mathbb{C}$ is a compact spectral set, and that Γ is a rectifiable closed simple curve enclosing Σ , separating it from the rest of the spectrum. Then, for n sufficiently large, $\sigma(A_n)$ is separated by Γ into a compact spectral set Σ_n and the rest of the spectrum. Denote by P and P_n the corresponding spectral projections. Then the dimension of the images of P and P_n are the same, and P_n converges to P in norm.

REMARK 4.3.3. As a consequence of the above corollary we get the continuity of every finite system of eigenvalues (counting multiplicity) and of the corresponding spectral projection. In particular, we get the continuity of an isolated eigenvalue of simple algebraic multiplicity and its eigenvector when normalised suitably (see [92, Section IV.3.5] for these facts on perturbation theory).

In all cases of domain perturbation we look at, we have that the resolvents $R_n(\lambda)$ act on $L_p(\mathbb{R}^N)$ for all $p \in (1, \infty)$ with image in $L_{m(p)}(\mathbb{R}^N)$ with m(p) given by (2.4.5).

THEOREM 4.3.4. Suppose that for every $p \in (1, \infty)$ there exists $M, \lambda > 0$ such that

$$\|R_n(\lambda)\|_{\mathcal{L}(L_n)} + \|R_n(\lambda)\|_{\mathcal{L}(L_n, L_m(n))} \le M$$
(4.3.1)

for all $n \in \mathbb{N}$. If $R(\lambda)$ is compact on $L_p(\Omega)$ for some $p \in (1, \infty)$ and $\lambda \in \varrho(A)$, then it is compact for all $p \in (1, \infty)$ and all $\lambda \in \varrho(A)$. Moreover, the following assertions are equivalent:

- (1) There exist $p_0 \in (1, \infty)$ and $\lambda > 0$ such that $R_n(\lambda) f_n \to R(\lambda) f$ in $L_{p_0}(\mathbb{R}^N)$ whenever $f_n \rightharpoonup f$ weakly in $L_p(\mathbb{R}^N)$.
- (2) There exist $p_0 \in (1, \infty)$ and $\lambda > 0$ such that $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_{p_0}(\mathbb{R}^N))$.
- (3) For every $\lambda \in \varrho(A)$ and $p \in (1, \infty)$ we have $R_n(\lambda) f_n \to R(\lambda) f$ in $L_q(\mathbb{R}^N)$ for all $q \in [p, m(p))$, whenever $f_n \to f$ weakly in $L_p(\mathbb{R}^N)$.
- (4) For every $\lambda \in \varrho(A)$ and $p \in (1, \infty)$ $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [p, m(p))$.

Assertions (2) and (4) are equivalent without the compactness of $R(\lambda)$.

PROOF. The equivalence of (1) and (2) follow directly from Proposition 4.1.1. We show that (2) implies (4). Note that the argument does not make use of the compactness of $R(\lambda)$. Together with Theorem 4.3.1 it follows from (2) that $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N))$ for

all $\lambda \in \varrho(A)$. Fix $p \in (1, \infty)$ and then $p_1 \in (1, \infty)$ such that either p_0 $or <math>p_0 > p > p_0$. Choose $\lambda \in \varrho(A)$ such that (4.3.1) holds for $p = p_1$. Then by the Riesz–Thorin interpolation theorem

$$\begin{aligned} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_p)} &\leq \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_{p_0})}^{1-\theta} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_{p_1})}^{\theta} \\ &\leq (2M)^{\theta} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_{p_0})}^{1-\theta} \end{aligned}$$

if we choose $\theta \in (0, 1)$ such that

$$\frac{1}{p} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}$$

(see [22, Theorem 1.1.1]). Hence $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N))$. If p < q < m(p), then again by (4.3.1) and the Riesz–Thorin interpolation theorem

$$\begin{aligned} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_p, L_q)} &\leq \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_p)}^{1-\theta} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_p, L_{m(p)})}^{\theta} \\ &\leq (2M)^{\theta} \|R_n(\lambda) - R(\lambda)\|_{\mathcal{L}(L_p)}^{1-\theta} \end{aligned}$$

with $\theta \in (0, 1)$ such that

$$\frac{1}{q} = \frac{1-\theta}{p} + \frac{\theta}{m(p)}.$$

Hence $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [p, m(p))$. Recall that we had fixed λ for this argument. Let $\mu \in \varrho(A)$ be arbitrary with $\lambda \neq \mu$. Since the map $T \mapsto (\alpha I - T)^{-1}$ is continuous as a map from $\mathcal{L}(E)$ into itself if $\alpha \in \varrho(T)$ (see [121, Theorem IV.1.5])

$$S_n := \frac{1}{\mu - \lambda} \left(\frac{1}{\mu - \lambda} I_E - R_n(\lambda) \right)^{-1} \to S := \frac{1}{\mu - \lambda} \left(\frac{1}{\mu - \lambda} I_E - R(\lambda) \right)^{-1}$$

in $\mathcal{L}(L_p(\mathbb{R}^N))$. Hence, using the identity (4.2.1),

$$\|R_{n}(\mu) - R(\mu)\|_{\mathcal{L}(L_{p},L_{q})} = \|R_{n}(\lambda)S_{n} - R(\lambda)S\|_{\mathcal{L}(L_{p},L_{q})}$$

$$\leq \|R_{n}(\lambda)\|_{\mathcal{L}(L_{p},L_{q})}\|S_{n} - S\|_{\mathcal{L}(L_{p})} + \|R_{n}(\lambda) - R(\lambda)\|_{\mathcal{L}(L_{p},L_{q})}\|S\|_{\mathcal{L}(L_{p})}$$

for all $n \in \mathbb{N}$. Using what we have proved already, we get $R_n(\mu) \to R(\mu)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [p, m(p))$. Since all operators $R(\lambda)$ interpolate, a compactness property of the Riesz–Thorin interpolation theorem shows that $R(\lambda)$ is compact as an operator in $\mathcal{L}(L_p(\mathbb{R}^N))$ and $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $p \in (1, \infty)$ and $q \in [p, m(p))$ if $R(\lambda)$ is compact (see [94]). Now the equivalence of (3) and (4) follows from Proposition 4.1.1.

5. Perturbations for linear Dirichlet problems

The most complete results on domain perturbation are for problems with Dirichlet boundary conditions. After stating the main assumptions we will give a complete characterisation of convergence of solutions for the Dirichlet problem on a domain Ω_n to a solution of the corresponding problem on Ω .

Theorem 5.2.4 is the main theorem on strong convergence and Theorem 5.2.6 the main result on convergence in the operator norm. Section 5.3 is then concerned with *necessary conditions* and Section 5.4 with *sufficient conditions* for convergence.

5.1. Assumptions and preliminary results

Given open sets $\Omega_n \subset \mathbb{R}^N$ ($N \ge 2$) we ask under what conditions the solutions of

$$\mathcal{A}_n u + \lambda u = f_n \quad \text{in } \Omega_n,$$

$$u = 0 \quad \text{on } \partial \Omega_n$$
 (5.1.1)

converge to a solution of the corresponding problem

$$\mathcal{A}u + \lambda u = f \quad \text{in } \Omega,$$

$$u = 0 \quad \text{on } \partial \Omega$$
(5.1.2)

on a limit domain Ω as $n \to \infty$. Most of the results in this section are taken from [58], but here we allow perturbations of \mathcal{A} as well. We make the following basic assumptions on the operators \mathcal{A}_n below.

ASSUMPTION 5.1.1. We let A_n be operators of the form

$$-\operatorname{div}(A_{0n}(x)\nabla u + a_n(x)u) + b_n(x) \cdot \nabla u + c_{0n}u$$
(5.1.3)

with $A_{0n} \in L_{\infty}(\mathbb{R}^N, \mathbb{R}^{N \times N})$, $a_n, b_n \in L_{\infty}(\mathbb{R}^N, \mathbb{R}^N)$ and $c_{0n} \in L_{\infty}(\mathbb{R}^N)$. Moreover, assume that the ellipticity constant $\alpha_0 > 0$ can be chosen uniformly with respect to $n \in \mathbb{N}$, and that

$$\sup_{n \in \mathbb{N}} \{ \|A_{0n}\|_{\infty}, \|a_n\|_{\infty}, \|b_n\|_{\infty}, \|c_{0n}\|_{\infty} \} < \infty.$$
(5.1.4)

We also assume that A is an operator of the form (2.1.2) and that

 $\lim_{n \to \infty} A_{0n} = A_0, \qquad \lim_{n \to \infty} a_n = a, \qquad \lim_{n \to \infty} b_n = b, \qquad \lim_{n \to \infty} c_{0n} = c_0 \ (5.1.5)$

almost everywhere in \mathbb{R}^N .

The condition that A_n , A be defined on \mathbb{R}^N is no restriction since by Remark 2.1.1 we can always extend them to \mathbb{R}^N . The bilinear forms associated with the boundary value problem are given by

$$a_n(u,v) := \int_{\Omega} (A_{0n} \nabla u + a_n u) \cdot \nabla v + (b_n \cdot \nabla u + c_{0n} u) v \, dx \tag{5.1.6}$$

for all $u, v \in H_0^1(\Omega_n)$ and by

$$a(u, v) := \int_{\Omega} (A_0 \nabla u + au) \cdot \nabla v + (b \cdot \nabla u + c_0 u) v \, dx.$$

for all $u, v \in H_0^1(\Omega)$. Applying Proposition 2.1.6 we get the following properties.

PROPOSITION 5.1.2. Suppose that Assumption 5.1.1 is satisfied. Then there exists M > 0 such that

$$|a_n(u,v)| \le M \|u\|_{H_0^1} \|v\|_{H_0^1}$$
(5.1.7)

for all $u, v \in H_0^1(\mathbb{R}^N)$ and all $n \in \mathbb{N}$. Moreover,

$$\frac{\alpha_0}{2} \|\nabla u\|_2^2 \le a_n(u, u) + \lambda \|u\|_2^2$$

for all $\lambda \in \mathbb{R}$ *with*

$$\lambda \ge \lambda_{\mathcal{A}} := \sup_{n \in \mathbb{N}} \left(\|c_{0n}^{-}\|_{\infty} + \frac{1}{2\alpha_{0}} \|a_{n} + b_{n}\|_{\infty} \right),$$
(5.1.8)

and

$$\frac{\alpha_0}{2} \|u\|_{H^1}^2 \le a_n(u, u) + \lambda \|u\|_2^2$$

for all $\lambda \in \mathbb{R}$ *with*

$$\lambda \ge \lambda_0 := \lambda_{\mathcal{A}} + \frac{\alpha_0}{2} \tag{5.1.9}$$

for all $u \in H_0^1(\mathbb{R}^N)$ and all $n \in \mathbb{N}$. Similar inequalities hold for $a(\cdot, \cdot)$ with the same constants. Finally,

$$\lim_{n \to \infty} a_n(u_n, v_n) = a(u, v) \tag{5.1.10}$$

if
$$u_n \rightarrow u$$
 weakly and $v_n \rightarrow v$ strongly in $H^1(\mathbb{R}^N)$ or vice versa.

PROOF. The first properties follow from Proposition 2.1.6. For the last, note the following fact. If c_n is bounded in $L_{\infty}(\mathbb{R}^N)$ with $c_n \to c$ pointwise and $w_n \to w$ in $L_2(\mathbb{R}^N)$, then $c_n w_n \to cw$ in $L_2(\mathbb{R}^N)$ as well. Indeed,

$$\|c_n w_n - cw\|_2 \le \|c_n (w_n - w)\|_2 + \|(c_n - c)w\|_2$$

$$\le \|c_n\|_{\infty} \|w_n - w\|_2 + \|(c_n - c)w\|_2,$$

where the first term on the right-hand side converges to zero because $||c_n||_{\infty}$ is bounded and $w_n \to w$ in $L_2(\mathbb{R}^N)$. The second term converges to zero by the dominated convergence theorem. Hence under the given assumptions, every term in (5.1.6) is the L_2 inner product of a strongly and a weakly converging sequence and therefore (5.1.10) follows.

Depending on the domains we can use $\|\nabla u\|_2$ as a norm on $H_0^1(\Omega)$, for instance if the measure of Ω_n is uniformly bounded, or if all Ω_n are contained between two parallel hyperplanes. Since we do not want to restrict ourselves to such a situation we will generally work with $\lambda \ge \lambda_0$ as given in (5.1.9). From the results in Section 2.2 we construct operators

$$A_n \in \mathcal{L}(H_0^1(\Omega_n), H^{-1}(\Omega_n))$$
 and $A \in \mathcal{L}(H_0^1(\Omega), H^{-1}(\Omega))$

where by definition

$$H^{-1}(\Omega) = (H^1_0(\Omega))'.$$

Generally, the right-hand side f of (5.1.2) is in $H^{-1}(\Omega)$, so the linear functional f is defined on the closed subspace $H_0^1(\Omega)$ of $H^1(\mathbb{R}^N)$. By the Hahn–Banach theorem (see [125, Theorem 6.5.1]) there exists an extension \tilde{f} of f with $\|\tilde{f}\|_{H^{-1}(\mathbb{R}^N)} = \|f\|_{H^{-1}(\Omega)}$. Hence we can assume without loss of generality that $f \in H^{-1}(\mathbb{R}^N)$.

Suppose that $R_n(\lambda)$, $R(\lambda)$ are given as in Definition 2.5.2. From Theorem 2.2.2 we conclude that

$$[\lambda_0,\infty)\subset \varrho(-A_n)\cap \varrho(-A),$$

and also the uniform estimate

$$\|R_n(\lambda)\|_{\mathcal{L}(H^{-1},H_0^1)} \le \frac{2}{\alpha_0}$$
(5.1.11)

for all $\lambda \geq \lambda_0$ and all $n \in \mathbb{N}$.

We summarise the results of this section in the following proposition. It is a uniform a priori estimate for weak solutions of (5.1.1) and (5.1.2).

PROPOSITION 5.1.3. *If* $\lambda \ge \lambda_0$ *, then*

$$\|R_n(\lambda)\|_{\mathcal{L}(H^{-1}(\mathbb{R}^N),H^1(\mathbb{R}^N)} \leq \frac{2}{\alpha_0}$$

for all $n \in \mathbb{N}$. A similar estimate with the same constant hold for $R(\lambda)$.

PROOF. The claim follows by combining Lemma 2.5.3, Proposition 2.1.6, Theorem 2.2.2 and Proposition 5.1.2. $\hfill \Box$

5.2. The main convergence result

In this section we summarise the main convergence results for Dirichlet problems. The bulk of the proof will be given in Section 5.5.

When proving that the solutions of (5.1.1) converge to a solution of (5.1.2), the following two conditions appear very naturally.

ASSUMPTION 5.2.1. Suppose that Ω_n , $\Omega \subset \mathbb{R}^N$. The weak limit points of every sequence $u_n \in H_0^1(\Omega_n)$ lie in $H_0^1(\Omega)$.

ASSUMPTION 5.2.2. Suppose that Ω_n , $\Omega \subset \mathbb{R}^N$ are open sets and for every $u \in H_0^1(\Omega)$ there exists $u_n \in H_0^1(\Omega_n)$ such that $u_n \to u$ in $H^1(\mathbb{R}^N)$.

If the above conditions are satisfied it is often said that $\Omega_n \to \Omega$ in the sense of Mosco as this is equivalent to $H_0^1(\Omega_n) \to H_0^1(\Omega)$ as subspaces of $H^1(\mathbb{R}^N)$ in the sense of Mosco [102, Section 1]. The conditions also appear in a more disguised form in [116], and explicitly in [119]. A discussion in terms of capacity appears in [25].

DEFINITION 5.2.3 (Mosco convergence). We say $\Omega_n \to \Omega$ in the sense of Mosco, if the open sets $\Omega_n, \Omega \subset \mathbb{R}^N$ satisfy Assumption 5.2.1 and Assumption 5.2.2.

For the formulation of the main convergence result for Dirichlet problems we use the notation and framework introduced in Section 5.1. In particular, $R_n(\lambda)f$ and $R(\lambda)f$ are the weak solutions of (5.1.1) and (5.1.2) extended to \mathbb{R}^N by zero. Also recall that we can choose f_n , $f \in H_0^1(\mathbb{R}^N)$ without loss of generality by extending the functionals by means of the Hahn–Banach Theorem if necessary.

THEOREM 5.2.4. If $\lambda \ge \lambda_0$, then the following assertions are equivalent.

- (1) $\Omega_n \to \Omega$ in the sense of Mosco;
- (2) $R_n(\lambda)f_n \rightarrow R(\lambda)f$ weakly in $H^1(\mathbb{R}^N)$ whenever $f_n \rightarrow f$ weakly in $H^{-1}(\mathbb{R}^N)$;
- (3) $R_n(\lambda) f_n \to R(\lambda) f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$;
- (4) $R_n(\lambda) f \rightarrow R(\lambda) f$ weakly in $H^1(\mathbb{R}^N)$ for f in a dense subset of $H^{-1}(\mathbb{R}^N)$.

The property that $R_n(\lambda)f \rightarrow R(\lambda)f$, at least in the case of the Laplace operator, is often called the γ -convergence of the solutions (see for instance [27]). Note that in particular, the above theorem implies that convergence is independent of the operator under consideration, a result that has been proved for a restricted class of operators in [15].

The above theorem does not say anything about convergence in the operator norm, it is only a theorem on the strong convergence of the resolvent operators. Strong convergence does not imply the convergence of the eigenvalues to the corresponding eigenvalues of the limit problem. However, according to Corollary 4.3.2 we get convergence of every finite part of the spectrum if the pseudo-resolvents converge in the operator norm.

If we assume that there is a bounded open set *B* such that Ω_n , $\Omega \subset B$ for all $n \in \mathbb{N}$, then we get convergence in the operator norm.

COROLLARY 5.2.5. Suppose that $\Omega_n \to \Omega$ in the sense of Mosco. Moreover, suppose that there exists a bounded open set B such that $\Omega_n, \Omega \subset B$ for all $n \in \mathbb{N}$. Finally let $\lambda \in \varrho(-A)$. Then $\lambda \in \varrho(-A_n)$ for n large enough, and $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(H^{-1}(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, 2d/(d-2))$.

PROOF. If $\lambda \geq \lambda_0$, then from Theorem 5.2.4 we have that $u_n := R_n(\lambda) f_n \rightarrow u := R(\lambda) f$ weakly in $H^1(\mathbb{R}^N)$ whenever $f_n \rightarrow f$ weakly in $H^{-1}(\Omega)$. Since $u_n \in H^1_0(B)$ for all $n \in \mathbb{N}$ and B is bounded, Rellich's Theorem implies that $u_n \rightarrow u$ in $L_q(\mathbb{R}^N)$ for all $q \in [1, 2d/(d-2))$. Hence $R_n(\lambda) \rightarrow R(\lambda)$ in $\mathcal{L}(H^{-1}(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, 2d/(d-2))$ by Proposition 4.1.1. The remaining assertions follow from Theorem 4.3.1. We now want to look at the situation where only Ω is bounded, but not necessarily Ω_n . We then get necessary and sufficient conditions for convergence in the operator norm. We denote by $\lambda_1(U)$ the *spectral bound* of $-\Delta$ on the open set U with Dirichlet boundary conditions. It is given by the variational formula

$$\lambda_1(U) = \inf_{\substack{u \in H_0^1(U) \\ u \neq 0}} \frac{\|\nabla u\|_{L_2(U)}^2}{\|u\|_{L_2(U)}^2}.$$
(5.2.1)

For convenience we set

 $\lambda_1(\emptyset) := \infty.$

We then have the following characterisation of convergence in the operator norm.

THEOREM 5.2.6. Suppose that Ω is bounded and that $\Omega_n \to \Omega$ in the sense of Mosco. Then the following assertions are equivalent.

- (1) There exists $\lambda > 0$ such that $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(H^{-1}(\mathbb{R}^N), L_2(\mathbb{R}^N))$.
- (2) For every $\lambda \in \varrho(-A)$ we have $\lambda \in \varrho(-A_n)$ for n large enough, and $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [p, m(p))$ and all $p \in (1, \infty)$, where m(p) is defined by (2.4.5) with d = N.
- (3) There exists an open set B with $\overline{\Omega} \subset B$ such that $\lambda_1(\Omega_n \cap \overline{B}^c) \to \infty$ as $n \to \infty$.

PROOF. We know that $R(\lambda) \in \mathcal{L}(H^{-1}(\mathbb{R}^N), H^1(\mathbb{R}^N))$. Since Ω is bounded, Rellich's Theorem implies that $R(\lambda) \in \mathcal{L}(H^{-1}(\mathbb{R}^N), L_2(\mathbb{R}^N))$ is compact. Hence by Proposition 4.1.1, assertion (1) is equivalent to the following statement:

(1') For some λ large enough $R_n(\lambda) f_n \to R(\lambda) f$ in $L_2(\mathbb{R}^N)$ whenever $f_n \rightharpoonup f$ weakly in $H^{-1}(\mathbb{R}^N)$.

Note that the above implies that $R_n(\lambda) f_n \to R(\lambda) f$ in $L_2(\mathbb{R}^N)$ whenever $f_n \rightharpoonup f$ weakly in $L_2(\mathbb{R}^N)$. From Theorem 2.4.2 we have uniform a priori estimates for $\lambda > 0$ large enough, and therefore Theorem 4.3.4 shows that (1') is equivalent to (2). Hence it remains to show that (1') is equivalent to (3).

Suppose that (1') is true, but not (3). Then there exists a bounded open set B containing $\overline{\Omega}$ such that $\lambda_1(\Omega_n \setminus \overline{B}^c) \not\rightarrow \infty$. Hence for every $k \in \mathbb{N}$ there exist $n_k > k$ and $\varphi_{n_k} \in C_c^{\infty}(\Omega_{n_k} \setminus \overline{B}^c)$ and c > 0 such that $\|\varphi_{n_k}\|_2 = 1$ and

$$0 \le \lambda_{n_k} \le \|\nabla \varphi_{n_k}\|_2^2 \le \epsilon$$

for all $k \in \mathbb{N}$. We define functionals $f_{n_k} \in H^{-1}(\mathbb{R}^N)$ by

 $\langle f_{n_k}, v \rangle := a_{n_k}(\varphi_{n_k}, v) + \lambda \langle \varphi_{n_k}, v \rangle$

for all $v \in H^1(\mathbb{R}^N)$. By (5.1.7) and the choice of φ_{n_k} we have

$$\|f_{n_k}\|_{H^{-1}} \le (M+\lambda)\|\varphi_{n_k}\|_{H^1} \le (M+\lambda)\sqrt{1+c^2}$$

for all $k \in \mathbb{N}$. This means that (f_{n_k}) is a bounded sequence in $H^{-1}(\mathbb{R}^N)$, and therefore has a subsequence converging weakly in $H^{-1}(\mathbb{R}^N)$ to some $f \in H^{-1}(\mathbb{R}^N)$. We denote

that subsequence again by (f_{n_k}) . By definition of f_{n_k} we have $\varphi_{n_k} = R_{n_k}(\lambda) f_{n_k}$ and by assumption (1')

$$\varphi_{n_k} = R_{n_k}(\lambda) f_{n_k} \to R(\lambda) f$$

in $L_2(\mathbb{R}^N)$. Since $\sup(\varphi_{n_k}) \cap \overline{\Omega} = \emptyset$, the definition of f_{n_k} implies that $f|_{H^{-1}(\Omega)} = 0$ and so $\varphi_{n_k} \to 0$ in $L_2(\mathbb{R}^N)$. However, this is impossible since we chose φ_{n_k} such that $\|\varphi_{n_k}\|_2 = 1$ for all $k \in \mathbb{N}$. Hence we have a contradiction, so (1') implies (3).

We finally prove that (3) implies (1'). Suppose that $f_n \rightharpoonup f$ weakly in $H^{-1}(\mathbb{R}^N)$. Then by Theorem 5.2.4

$$u_n := R_n(\lambda) f_n \rightharpoonup u := R(\lambda) f$$

weakly in $H^1(\mathbb{R}^N)$. Let *B* be an open set as in (3) and choose an open bounded set *U* with $\overline{B} \subset U$. Then by Rellich's theorem $u_n \to u$ in $L_2(U)$. Hence it remains to show that $u_n \to 0$ in $L_2(\mathbb{R}^N \setminus U)$. We choose a cutoff function $\psi \in C^{\infty}(\mathbb{R}^N)$ such that $0 \le \psi \le 1, \psi = 0$ on \overline{B} and $\psi = 1$ on $\mathbb{R}^N \setminus U$. Then, $\psi u_n \in H^1(\Omega \cap \overline{B}^c)$, and setting $\lambda_n := \lambda_1(\Omega \setminus \overline{B})$ we get

$$\lambda_{n} \|u_{n}\|_{L_{2}(\mathbb{R}^{N}\setminus U)}^{2} \leq \lambda_{n} \|\psi u_{n}\|_{L_{2}(\mathbb{R}^{N}\setminus\bar{B})}^{2}$$

$$\leq \|\nabla(\psi u_{n})\|_{L_{2}(\mathbb{R}^{N}\setminus\bar{B})}^{2} = \|\nabla(\psi u_{n})\|_{2}^{2}$$
(5.2.2)

for all $n \in \mathbb{N}$. Since (f_n) is bounded in $H^{-1}(\mathbb{R}^N)$, the sequence (u_n) is bounded in $H^1(\mathbb{R}^N)$. Hence

$$\|\nabla(\psi u_n)\|_2^2 \le (\|\psi\|_{\infty}^2 + \|\psi\|_{\infty}^2) \|u_n\|_{H^1}^2$$

is bounded. Because $\lambda_n \to \infty$ by assumption, (5.2.2) implies that $u_n \to 0$ in $L_2(\mathbb{R}^N \setminus U)$. Hence $u_n \to u$ in $L_2(\mathbb{R}^N)$ as claimed. This completes the proof of the theorem. \Box

REMARK 5.2.7. (a) Note that condition (3) in the above theorem is always satisfied if $\lambda_1(\Omega_n \setminus \overline{\Omega}^c) \to \infty$ as $n \to \infty$. Indeed, from the monotonicity of the first eigenvalue of the Dirichlet problem with respect to the domain

$$\lambda_1(\Omega_n \setminus \Omega^c) \leq \lambda_1(\Omega_n \setminus B^c)$$

for every bounded set *B* with $\Omega \subset B$. The monotonicity is a consequence of the variational formula (5.2.1).

(b) Note that (3) is also satisfied if $|\Omega_n \cap \overline{\Omega}^c| \to 0$, whether or not Ω_n is bounded. To see this let B_n be a ball of the same volume as $\Omega_n \cap \overline{\Omega}^c$. As the measure goes to zero $\lambda_1(B_n) \to \infty$, and by the isoperimetric inequality for the first eigenvalue of the Dirichlet problem (see [20,83]) we get

$$\lambda_1(\Omega_n \cap \Omega^c) \ge c\lambda_1(B_n) \to \infty.$$

EXAMPLE 5.2.8. We give a situation, where we get convergence in the operator norm, but Ω_n has unbounded measure for all $n \in \mathbb{N}$. We can take a disk and attach an infinite strip. We then let the width of the strip tend to zero. Then by Friedrich's inequality (2.1.7)

$$\lambda_1(\Omega_n\setminusar\Omega)\geq rac{1}{D^2} o\infty$$

if the width D of the strip goes to zero. The situation is depicted in Figure 5.1.



Fig. 5.1. Disc with an infinite strip attached.



Fig. 5.2. Disc with an infinite cone attached.

In contrast, if we attach a cone of angle β_n rather than a strip, then convergence is not in the operator norm if $\beta_n \to 0$ as shown in Figure 5.2. The fact that these domains converge in the sense of Mosco follows from Theorem 5.4.5 below. Note that $\lambda_1(\Omega_n) = 0$ for all $n \in \mathbb{N}$, and therefore $\lambda_1(\Omega_n) \not\rightarrow \lambda_1(\Omega)$. This means that there is no convergence of the spectrum. More examples are given in [58, Section 8]. Examples where just part of the spectrum converges can be found in [107].

5.3. Necessary conditions for convergence

In this section we collect some necessary conditions for convergence in the sense of Mosco. For a convergence result such as the one in Theorem 5.2.4 we clearly need that the support of the limit function is in $\overline{\Omega}$. We give a simple characterisation of such a requirement in terms of the spectral bound of the Laplacian on bounded sets outside the limit set $\overline{\Omega}$.

THEOREM 5.3.1. For open sets Ω_n , $\Omega \subset \mathbb{R}^N$ the following assertions are equivalent.

- (1) The weak limit points of every sequence $u_n \in H^1_0(\Omega_n)$, $n \in \mathbb{N}$, in $H^1(\mathbb{R}^N)$ have support in $\overline{\Omega}$;
- (2) For every open bounded set B with $\overline{B} \subset \mathbb{R}^N \setminus \overline{\Omega}$ $\lim_{n \to \infty} \lambda_1(\Omega_n \cap B) = \infty;$ (5.3.1)
- (3) There exists an open covering \mathcal{O} of $\mathbb{R}^N \setminus \overline{\Omega}$ such that (5.3.1) holds for all $B \in \mathcal{O}$.

PROOF. Suppose that (1) holds and let *B* be a bounded open set with $\overline{B} \subset \mathbb{R}^N \setminus \overline{\Omega}$. Set $\lambda_n := \lambda_1(\Omega_n \cap B)$. Then, by the variational characterisation (5.2.1) of the spectral bound, for every $n \in \mathbb{N}$ there exists $v_n \in C_c^{\infty}(\Omega_n \cap B)$ with

$$(\lambda_n + 1) \|v_n\|_2^2 \ge \|\nabla v_n\|_2^2 = 1.$$
(5.3.2)

Since *B* is bounded (2.1.7) implies that (v_n) is bounded in $H_0^1(B)$. Hence there exists a subsequence (v_{n_k}) converging weakly to some v in $H_0^1(B)$. By assumption $\sup (v) \subset \overline{B} \subset \mathbb{R}^N \setminus \overline{\Omega}$, and so (1) implies that v = 0. As *B* is bounded Rellich's Theorem shows that $||v_{n_k}||_2 \to 0$. Hence, (5.3.2) can only be true if $\lambda_{n_k} - 1 \to \infty$, implying that $\lambda_{n_k} \to \infty$ as $k \to \infty$. The above arguments apply to every weakly convergent subsequence of (v_n) and therefore (1) implies (2).

Clearly (2) implies (3) and so it remains to prove that (3) implies (1). Suppose that $u_n \in H_0^1(\Omega_n)$, and that $u_{n_k} \rightarrow u$ weakly in $H^1(\mathbb{R}^N)$ as $k \rightarrow \infty$. Let \mathcal{O} be an open covering of $\mathbb{R}^N \setminus \overline{\Omega}$ with the properties stated in (3). Fix $B \subset \mathcal{O}$ and let $\varphi \in C_c^\infty(B)$. Then $\varphi u_n \in H_0^1(\Omega_n \cap \overline{\Omega}^c \cap B)$. The map $u_n \rightarrow u_n \varphi$ is a bounded linear map from $H^1(\mathbb{R}^N)$ to $H_0^1(B)$ and therefore is weakly continuous. Hence, if $u_{n_k} \rightarrow u$ weakly in $H^1(\mathbb{R}^N)$, then $\varphi u_{n_k} \rightarrow \varphi u$ weakly in $H_0^1(B)$. As *B* is bounded, Rellich's theorem implies that $\varphi u_{n_k} \rightarrow \varphi u$ in $L_2(\mathbb{R}^N)$. Now by (5.3.1) and the boundedness of the sequence $(\|\nabla(\varphi u_{n_k})\|_2)$

$$\|\varphi u\|_{2}^{2} = \lim_{k \to \infty} \|\varphi u_{n_{k}}\|_{2}^{2} \le \lim_{k \to \infty} \frac{\|\nabla(\varphi u_{n_{k}})\|_{2}^{2}}{\lambda_{1}(\Omega_{n_{k}} \cap B)} = 0.$$

Hence $\varphi u = 0$ almost everywhere for all $\varphi \in C_c^{\infty}(B)$, so u = 0 almost everywhere in *B*. As \mathcal{O} is a covering of $\mathbb{R}^N \setminus \overline{\Omega}$ it follows that supp $u \subset \overline{\Omega}$ as claimed.

REMARK 5.3.2. The above condition does not imply Assumption 5.2.1. The reason is that a function $u \in H^1(\mathbb{R}^N)$ with supp $(u) \subset \overline{\Omega}$ does not need to be in $H_0^1(\Omega)$. We discuss conditions for that in the next subsection.

We next give a characterisation of Assumption 5.2.2 in terms of capacity. A related result appears in [108, Proposition 4.1] and a proof is given in [77, page 75] or [119, page 24]. Our exposition follows [58, Section 7]. Recall that the capacity (or more precisely (1, 2)-capacity) of a compact set $E \subset \mathbb{R}^N$ is given by

$$\operatorname{cap}(E) := \inf\{\|u\|_{H^1}^2 \colon u \in H_0^1(\mathbb{R}^N) \text{ and } u \ge 1 \text{ in a neighbourhood of } E\}$$

(see [80, Section 2.35]). We could also define capacity with respect to an open set U and define for $E \subset U$ compact $E \subset \mathbb{R}^N$ given by

 $\operatorname{cap}_{U}(E) := \inf\{\|u\|_{H^{1}}^{2} : u \in H_{0}^{1}(U) \text{ and } u \ge 1 \text{ in a neighbourhood of } E\}.$

It turns out that $\operatorname{cap}_U(E) = 0$ if and only if $\operatorname{cap}(E) = 0$. Moreover, we can work with $u \in C_c^{\infty}(\mathbb{R}^N)$ and $u \in C_c^{\infty}(U)$, respectively rather than the Sobolev spaces.

PROPOSITION 5.3.3. Let $\Omega_n, \Omega \subset \mathbb{R}^N$ be open sets. Then the following conditions are equivalent.

- (1) *Assumption* 5.2.2;
- (2) For every open set $B \subset \mathbb{R}^N$ and every $\varphi \in C_c^{\infty}(\Omega \cap B)$ there exists $\varphi_n \in C_c^{\infty}(\Omega_n \cap B)$ such that $\varphi_n \to \varphi$ in $H_0^1(\Omega \cap B)$;
- (3) For every compact set $K \subset \Omega$ $\lim_{n \to \infty} \operatorname{cap}(K \cap \Omega_n^c) = 0.$
PROOF. We prove that (1) implies (3). Fix a compact set $K \subset \Omega$ and let $\varphi \in C_c^{\infty}(\Omega)$ with $0 \le \varphi \le 1$ and $\varphi = 1$ in a neighbourhood of K. By assumption there exists $u_n \in H_0^1(\Omega)$ such that $u_n \to \varphi$ in $H^1(\mathbb{R}^N)$. As $C_c^{\infty}(\Omega)$ is dense in $H_0^1(\Omega)$ there exists $\varphi_n \in C_c^{\infty}(\Omega)$ such that $||u_n - \varphi_n||_{H^1} < 1/n$. Then

$$\|\varphi_n - \varphi\|_{H^1} \le \|\varphi_n - u_n\|_{H^1} + \|u_n - \varphi\|_{H^1} \le \frac{1}{n} + \|u_n - \varphi\|_{H^1} \to 0.$$

Now set $\psi_n := \varphi - \varphi_n$. Then by construction $\psi_n \in H^1(\mathbb{R}^N)$ and $\psi_n = 1$ in a neighbourhood of $K \cap \Omega_n^c$. Hence by definition of capacity

$$\operatorname{cap}(K \cap \Omega_n^c) \le \|\psi_n\|_{H^1}^2 = \|\varphi_n - \varphi\|_{H^1}^2 \to 0$$

as claimed.

We next prove that (3) implies (2). We fix an open set $B \subset \mathbb{R}^N$. Clearly we only need to consider the case where $\Omega \cap B \neq \emptyset$. Let $\varphi \in C_c^{\infty}(\Omega \cap B)$. By definition of capacity there exists $\psi_n \in C_c(\Omega \cap B)$ such that $\psi_n = 1$ on supp $\varphi \cap \Omega_n^c$ and such that

$$\|\psi_n\|_{H^1}^2 \le \operatorname{cap}(\operatorname{supp} \varphi \cap \Omega_n^c) + \frac{1}{n}$$

for all $n \in \mathbb{N}$. Hence by assumption $\psi_n \to 0$ in $H^1(\mathbb{R}^N)$. We now set $\varphi_n := (1 - \psi_n)\varphi$. Then by construction $\varphi_n \in C_c^{\infty}(\Omega \cap B)$ and

$$\begin{aligned} \|\varphi_n - \varphi\|_{H^1} &= \|\varphi\psi_n\|_{H^1} \le \|\varphi\psi_n\|_2 + \|\psi_n\nabla\varphi + \varphi\nabla\psi_n\|_2 \\ &\le (\|\varphi\|_{\infty} + \|\nabla\varphi\|_{\infty})\|\psi_n\|_2 + \|\varphi\|_{\infty}\|\nabla\psi_n\|_2 \\ &\le 2(\|\varphi\|_{\infty} + \|\nabla\varphi\|_{\infty})\|\psi_n\|_{H^1} \to 0. \end{aligned}$$

Hence we have found $\varphi_n \in C_c^{\infty}(\Omega \cap B)$ with $\varphi_n \to \varphi$ in $H^1(\mathbb{R}^N)$, proving (2).

We finally prove that (2) implies (1). For given $u \in H_0^1(\Omega)$ there exists $\varphi_k \in C_c^{\infty}(\Omega)$ such that $\varphi_n \to u$ in $H^1(\Omega)$. Now by (2) there exists $\varphi_{k,n} \in C_c^{\infty}(\Omega_n)$ such for every fixed $k \in \mathbb{N}$ we have $\varphi_{k,n} \to \varphi_k$ in $H^1(\mathbb{R}^N)$ as $n \to \infty$. Hence for every $k \in \mathbb{N}$ there exists $n_k \in \mathbb{N}$ such that $\|\varphi_{k,n} - \varphi_k\|_{H^1} < 1/k$ for all $n > n_k$. We can also arrange that $n_k < n_{k+1}$ for all $k \in \mathbb{N}$. Now we set $u_n := \varphi_{k,n}$ whenever $n_k < n \le n_{k+1}$. Then $u_n \in H_0^1(\Omega_n)$ and our aim is to show that $u_n \to u$ in $H^1(\mathbb{R}^N)$. To do so fix $\varepsilon > 0$. As $\varphi_k \to u$ in $H^1(\mathbb{R}^N)$ there exists $k_0 \in \mathbb{N}$ such that $1/k + \|\varphi_k - u\|_{H^1} < \varepsilon$ for all $k > k_0$. Given $n > n_{k_0+1}$ there exists $k > k_0$ such that $n_k < n \le n_{k+1}$ and so by construction $u_n = \varphi_{k,n}$ and

$$\|u_n - u\|_{H^1} \le \|\varphi_{k,n} - \varphi_k\|_{H^1} + \|\varphi_k - u\|_{H^1} \le 1/k + \|\varphi_k - u\|_{H^1} < \varepsilon.$$

This shows that $||u_n - u||_{H^1} < \varepsilon$ for all $n > n_{k_0+1}$. As $\varepsilon > 0$ was arbitrary we conclude that $u_n \to u$ in $H^1(\mathbb{R}^N)$ as claimed.

5.4. Sufficient conditions for convergence

In this section we collect some simple sufficient conditions for $\Omega_n \to \Omega$ in the sense of Mosco. First we look at approximations of an open set Ω by open sets from the inside.

PROPOSITION 5.4.1. Suppose that $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets. If supposing that $\Omega_n \subset \Omega_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$, and $\Omega = \bigcup_{n \in \mathbb{N}} \Omega_n$, then $\Omega_n \to \Omega$ in the sense of Mosco.

PROOF. Since $H_0^1(\Omega_n) \subset H_0^1(\Omega)$ for all $n \in \mathbb{N}$, Assumption 5.2.1 is clearly satisfied. Suppose that $u \in H_0^1(\Omega)$. If $\varphi \in C_c^{\infty}(\Omega)$, then by assumption there exists $n_0 \in \mathbb{N}$ such that supp $(\varphi) \subset \Omega_n$ for all $n \ge n_0$. We now choose $\phi_n \in C_c^{\infty}(\Omega_n)$ arbitrary for $1 \le n \le n_0$ and $\varphi_n := \varphi$ for all $n > n_0$. Then clearly $\varphi_n \to \varphi$ in $H^1(\mathbb{R}^N)$ and Assumption 5.2.2 follows from Proposition 5.3.3.

For approximations from the outside we need a weak regularity condition on the boundary of Ω . We define

$$H_0^1(\bar{\Omega}) := \{ u \in H^1(\mathbb{R}^N) : u = 0 \text{ almost everywhere on } \bar{\Omega}^c \}$$

We make the following definition.

DEFINITION 5.4.2. We say the open set $\Omega \subset \mathbb{R}^N$ is stable if $H_0^1(\Omega) = H_0^1(\overline{\Omega})$.

The above notion of stability is the same as the stability of the Dirichlet problem for harmonic functions on Ω as introduced in Keldyš [93]. An excellent discussion of bounded stable sets is given in [79]. A discussion on the connections between stability of the Dirichlet problem for harmonic functions and the Poisson problem by more elementary means is presented in [9].

PROPOSITION 5.4.3. An open set $\Omega \subset \mathbb{R}^N$ is stable if one of the following conditions is satisfied:

(1) Ω has the segment property except possibly on a set of capacity zero;

(2) for all $x \in \partial \Omega$ except possibly a set of capacity zero $\liminf_{r \to 0} \frac{\operatorname{cap}(\bar{\Omega}^c \cap B(x, r))}{\operatorname{cap}(\Omega^c \cap B(x, r))} > 0,$ where B(x, r) is the ball of radius r centred at x.

The last condition is necessary and sufficient for the stability of Ω .

PROOF. For a proof of (1) we refer to [77, p. 77/78], [119, Section 3.2] or [124, Satz 4.8]), and for (2) to [1, Theorem 11.4.1].

More characterisations of stability are in [79, Theorem 11.9]. Note that, if Ω is Lipschitz (or even smoother), then Ω satisfies the segment condition and Ω is therefore stable. According to [67, Theorem V.4.4], the segment condition is equivalent to $\partial \Omega$ to be continuous.

PROPOSITION 5.4.4. Suppose that $\Omega \subset \Omega_{n+1} \subset \Omega_n$ for all $n \in \mathbb{N}$, and that $\bigcap_{n \in \mathbb{N}} \Omega_n \subset \overline{\Omega}$. If Ω is stable, then $\Omega_n \to \Omega$ in the sense of Mosco.

PROOF. Since $H_0^1(\Omega) \subset H_0^1(\Omega_n)$ for all $n \in \mathbb{N}$, Assumption 5.2.2 is clearly satisfied. Suppose now that (u_n) is a sequence in $H_0^1(\Omega_n)$. Since $\bigcap_{n \in \mathbb{N}} \Omega_n \subset \overline{\Omega}$ it follows that every weak limit point u of that sequence has support in $\overline{\Omega}$. Hence by the stability of Ω

Domain perturbation for linear and semi-linear boundary value problems



Fig. 5.3. Cracking domain.

we get that $u \in H_0^1(\Omega)$ as required in Assumption 5.2.1. Hence $\Omega_n \to \Omega$ in the sense of Mosco.

THEOREM 5.4.5. Suppose that Ω_n , $\Omega \subset \mathbb{R}^N$ are open (not necessarily bounded) sets, and that Ω is stable. Then $\Omega_n \to \Omega$ in the sense of Mosco if and only if the following two conditions are satisfied.

- (1) $\operatorname{cap}(K \cap \Omega_n^c) \to 0$ as $n \to \infty$ for all compact sets $K \subset \Omega$;
- (2) There exists an open covering \mathcal{O} of $\mathbb{R}^N \setminus \overline{\Omega}$ such that $\lambda_1(U \cap \Omega_n) \to \infty$ as $n \to \infty$ for all $U \in \mathcal{O}$;

PROOF. The assertion follows from Proposition 5.3.3 and Theorem 5.3.1, together with the definition of stability. $\hfill \Box$

Note that the requirement that Ω be stable is too much in certain cases. An example is a cracking domain as in Figure 5.3. It is sufficient to require a condition on

$$\Gamma := \bigcap_{n \in \mathbb{N}} \left(\overline{\bigcup_{k \ge n} (\Omega_k \cap \partial \Omega)} \right) \subset \partial \Omega.$$
(5.4.1)

For instance for the cracking domain we consider, the set Γ consists of the end point of the crack. As that set is of capacity zero, we get convergence in the sense of Mosco. Also if $\Gamma \subset \partial \Omega$ satisfies a segment condition except at a set of capacity zero, we also get convergence of Ω_n . A discussion of this condition is given in [58, Section 7] or [119]. Examples of cracking domains also appear in [124].

5.5. *Proof of the main convergence result*

To prove Theorem 5.2.4 we will proceed as follows. First we prove that (1) implies (2) and that (2) implies (3). We then observe that Assumption 5.2.1 follows from (2), whereas Assumption 5.2.2 follows from (3). Hence we could try to prove that (3) implies (2) to get from (3) back to (1). However, this does not seem to be possible. Instead we prove that (3) implies a statement similar to (2), but for the formally adjoint problem. That still implies Assumption 5.2.2, so (3) implies (1). We then prove the equivalence to (4) separately by using the uniform estimate on the norm of $R_n(\lambda)$ and the density.

We start by proving that (1) implies (2).

PROPOSITION 5.5.1. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets with $\Omega_n \to \Omega$ in the sense of Mosco. If $\lambda \geq \lambda_0$, then $R_n(\lambda) f_n \rightharpoonup R(\lambda) f$ weakly in $H^1(\mathbb{R}^N)$ whenever $f_n \rightharpoonup f$ weakly in $H^{-1}(\Omega)$.

PROOF. Let $f_n \rightharpoonup f$ weakly in $H^{-1}(\mathbb{R}^N)$. Set $u_n := R_n(\lambda) f_n$. By Proposition 5.1.3 we have

$$||u_n||_{H^1} \le \frac{2}{\alpha_0} ||f_n||_{H^{-1}}$$

for all $n \in \mathbb{N}$. Since (f_n) is weakly convergent and therefore bounded, the sequence (u_n) is bounded in $H^1(\mathbb{R}^N)$. Hence it has a weakly convergent subsequence (u_{n_k}) with limit v. By Assumption 5.2.1 we have $v \in H_0^1(\Omega)$. Given $\varphi \in C_c^{\infty}(\Omega)$ Assumption 5.2.2 implies that there exist $\varphi_n \in H_0^1(\Omega_n)$ with $\varphi_n \to \varphi$ in $H^1(\mathbb{R}^N)$. Because u_n is a weak solution of (5.1.1) we get

$$a_n(u_n,\varphi_n)=\langle f_n,\varphi_n\rangle$$

for all $n \in \mathbb{N}$. Since $u_n \rightarrow v$ weakly and $\varphi_n \rightarrow \varphi$ strongly we can use (5.1.10) to pass to the limit in the above identity. Hence

$$a_n(v,\varphi) = \langle f,\varphi \rangle$$

for all $\varphi \in C_c^{\infty}(\Omega)$, showing that *v* is a weak solution of (5.1.2). Since (5.1.2) has a unique solution we conclude that $v = R(\lambda) f$ and that the whole sequence converges.

Next we prove that (2) implies (3).

PROPOSITION 5.5.2. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets and that $\lambda \geq \lambda_0$. Moreover, suppose $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$ with $R_n(\lambda)f_n \to R(\lambda)f$ weakly in $H^1(\mathbb{R}^N)$. Then $R_n(\lambda)f_n \to R(\lambda)f$ in $H^1(\mathbb{R}^N)$.

PROOF. Assume that $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$, so that $u_n := R_n(\lambda) f_n \to u := R(\lambda) f$ weakly in $H^1(\mathbb{R}^N)$. Hence because in every instance a strongly and a weakly convergent sequence is paired,

$$\lim_{n \to \infty} (a_n(u_n, u_n) + \lambda \|u_n\|_2^2) = \lim_{n \to \infty} \langle f_n, u_n \rangle = \langle f, u \rangle = a(u, u) + \lambda \|u\|_2^2,$$

and also

$$\lim_{n \to \infty} (a_n(u_n, u) + \lambda \langle u_n, u \rangle) = \lim_{n \to \infty} (a_n(u, u_n) + \lambda \langle u, u_n \rangle) = a(u, u) + \lambda ||u||_2^2.$$

Therefore

$$a_n(u_n - u, u_n - u) + \lambda ||u_n - u||_2^2 = a_n(u_n, u_n) + \lambda ||u_n||_2^2 - (a_n(u_n, u) + \lambda \langle u_n, u \rangle) - (a_n(u, u_n) + \lambda \langle u, u_n \rangle) + a(u, u) + \lambda ||u||_2^2 \to 0.$$

From Proposition 5.1.2 we get

$$\frac{2}{\alpha_0} \|u_n - u\|_{H^1}^2 \le a_n (u_n - u, u_n - u) + \lambda \|u_n - u\|_2^2 \to 0,$$

showing that $u_n \to u$ strongly in $H^1(\mathbb{R}^N)$.

		н	
		н	

As remarked earlier we cannot prove directly the converse of the above proposition, but we can prove the corresponding weak convergence property for the formally adjoint problem we introduced in Section 2.3.

PROPOSITION 5.5.3. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets and that $\lambda \geq \lambda_0$. Suppose that $R_n(\lambda) f_n \to R(\lambda) f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$. Then $R_n^{\sharp}(\lambda) f_n \to R^{\sharp}(\lambda)$ weakly in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ weakly in $H^{-1}(\mathbb{R}^N)$.

PROOF. Assume that $f_n \rightarrow f$ weakly in $H^{-1}(\mathbb{R}^N)$ and fix $g \in H^{-1}(\mathbb{R}^N)$. Then by (3) $R_n(\lambda)g \rightarrow R(\lambda)g$ and so by Lemma 2.5.3

$$\langle g, R_n^{\sharp}(\lambda) f_n \rangle = \langle R_n(\lambda)g, f_n \rangle \to \langle R(\lambda)g, f \rangle = \langle g, R^{\sharp}(\lambda)f \rangle,$$

completing the proof of the proposition.

We now prove that the weak convergence property (2) implies Assumption 5.2.2 and and the strong convergence property (3) implies Assumption 5.2.1.

LEMMA 5.5.4. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets and that $\lambda \geq \lambda_0$. Suppose that $R_n(\lambda)f_n \rightharpoonup R(\lambda)f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \rightharpoonup f$ in $H^{-1}(\mathbb{R}^N)$. Then Assumption 5.2.1 holds.

PROOF. Let $u_n \in H_0^1(\Omega_n)$ and define functionals $f_n \in H^{-1}(\mathbb{R}^N)$ by

 $\langle f_n, v \rangle := a_n(u_n, v) + \lambda \langle u_n, v \rangle$

for all $v \in H^1(\mathbb{R}^N)$. By (5.1.7)

$$||f_n||_{H^{-1}} \le (M+\lambda) ||u_n||_{H^1}$$

for all $n \in \mathbb{N}$. Now assume that $u_{n_k} \rightharpoonup u$ weakly in $H^1(\mathbb{R}^N)$. We need to show that $u \in H_0^1(\Omega)$. Since (u_{n_k}) is bounded, the above shows that (f_{n_k}) is bounded in $H^{-1}(\mathbb{R}^N)$. Because every bounded sequence in a Hilbert space has a weakly convergent subsequence there exists a further subsequence, denoted again by (f_{n_k}) , with $f_{n_k} \rightharpoonup f$. Now by assumption and the definition of f_n

$$u_{n_k} := R_{n_k}(\lambda) f_{n_k} \rightharpoonup R(\lambda) f = u$$

Since $u \in H_0^1(\Omega)$ Assumption 5.2.1 follows.

LEMMA 5.5.5. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets and that $\lambda \geq \lambda_0$. Suppose that $R_n(\lambda) f_n \to R(\lambda) f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$. Then Assumption 5.2.2 holds.

PROOF. Fix $\varphi \in H_0^1(\Omega)$ and define $f \in H^{-1}(\mathbb{R}^N)$ by

$$\langle f, v \rangle := a(\varphi, v) + \lambda \langle \varphi, v \rangle$$

for all $v \in H^1(\mathbb{R}^N)$. Then by (3) and the definition of f

$$\varphi_n := R_n(\lambda) f \to R(\lambda) f = \varphi$$

in $H^1(\mathbb{R}^N)$. Since $\varphi_n \in H^1_0(\Omega_n)$, Assumption 5.2.2 follows.

41

We finally need to prove that (4) is equivalent to the other assertions. Clearly (2) implies (4), so we only need to prove that (4) implies (3). The proof is an abstract argument just using a uniform bound on the norms of $R_n(\lambda)$.

LEMMA 5.5.6. Suppose $\Omega_n, \Omega \subset \mathbb{R}^N$ are open sets and that $\lambda \geq \lambda_0$. Suppose that $R_n(\lambda)f \to R(\lambda)f$ for f in a dense subset of $H^{-1}(\mathbb{R}^N)$, then $R_n(\lambda)f_n \to R(\lambda)f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ in $H^{-1}(\mathbb{R}^N)$.

PROOF. Let *V* be the dense set of $H^{-1}(\mathbb{R}^N)$ for which $R_n(\lambda)g \to R(\lambda)g$ weakly in $H^1(\mathbb{R}^N)$ for all $g \in V$. Then by Proposition 5.5.2 convergence is actually in $H^1(\mathbb{R}^N)$, so $R_n(\lambda) \to R(\lambda)$ strongly on the dense subset *V*. Since the norms of $||R_n(\lambda)||$ is uniformly bounded by Proposition 5.1.3, we have strong convergence on $H^1(\mathbb{R}^N)$. Also because of the uniform bound and the strong convergence, $R_n(\lambda)f_n \to R(\lambda)f$ in $H^1(\mathbb{R}^N)$ whenever $f_n \to f$ in $H^{-1}(\Omega)$.

6. Varying domains and Robin boundary conditions

An interesting feature of Robin problems is, that for a sequence of domains, the boundary condition can change in the limit. We consider three different cases.

6.1. Summary of results

Given open sets $\Omega_n \subset \mathbb{R}^N$ $(N \ge 2)$ we consider convergence of solutions of the Robin problems

$$\mathcal{A}_n u + \lambda u = f_n \qquad \text{in } \Omega_n,$$

$$\frac{\partial}{\partial \nu_A} u + b_{0n} u = 0 \qquad \text{on } \partial \Omega_n \qquad (6.1.1)$$

to a solution of a limit problem

$$\begin{aligned} \mathcal{A}u + \lambda u &= f & \text{ in } \Omega, \\ \mathcal{B}u &= 0 & \text{ on } \partial \Omega \end{aligned}$$
(6.1.2)

on a domain Ω as $n \to \infty$. On the operators \mathcal{A}_n we make the same assumptions as in the case of the Dirichlet problem. We also need a positivity assumption on the boundary coefficient b_0 .

ASSUMPTION 6.1.1. Suppose that A_n , A satisfy Assumption 5.1.1. Moreover, let $b_{0n} \ge \beta$ for some constant $\beta > 0$.

The above conditions allow us to make use of the domain-independent a priori estimates proved in Section 2.4.2.

The situation is not as simple as in the case of Dirichlet boundary conditions, where the limit problem satisfies Dirichlet boundary conditions as well. Here, the boundary



Fig. 6.1. Dumbbell like domain converging to two circles.



Fig. 6.2. Domains with fast oscillating boundaries approaching a disc.

conditions of the limit problem depends on how the domains Ω_n approach Ω . We consider the following three cases:

- (1) The boundary $\partial\Omega$ is only modified in the neighbourhood of a very small set, namely a set of capacity zero. A prototype of such an approximation is a dumbbell with a handle C_n shrinking to a line. The limit set consists of two disconnected sets as shown in Figure 6.1. It is also possible to cut small holes and shrink them to a set of capacity zero. The limit problem is then a Robin problem with the same boundary conditions as the approximating problems. How much boundary we add is irrelevant. A precise statement is in Theorem 6.3.3.
- (2) The boundaries of the approaching domains are wildly oscillating. If the oscillations, which do not necessarily need to be periodic, are very fast, then the limit problem turns out to be a Dirichlet problem. See Figure 6.2 for an example. A magnification of the boundary of the last domain shown in the sequence is displayed in Figure 6.3. The precise result is stated in Theorem 6.4.3.
- (3) The boundaries of the approaching domains oscillate moderately, not necessarily in a periodic fashion. The limit domain can then be a Robin problem with a different coefficient b_0 in the boundary conditions. Again, an example is as shown in Figure 6.2, with boundary not oscillating quite as fast. The precise result is stated in Theorem 6.5.1.

To see that such phenomena are to be expected, look at the model of heat conduction. The boundary conditions describe a partially insulated boundary, where the loss of heat is proportional to the temperature at the boundary. If the boundary becomes longer the loss is bigger. If the additional boundary is only connected to the main body by a small set as for instance the handle in the case of the dumbbell, then its influence on the temperature inside the major parts of the body is negligible. This corresponds to case one. In the second case, the oscillating boundary will act like a radiator and cool the body better and better, the longer the boundary gets. As the length of the boundary goes to infinity, the





Fig. 6.3. Enlarged portion of a very fast oscillating boundary.

cooling becomes perfect and we get Dirichlet boundary conditions. In the third case the boundary oscillates, but its surface area does not go to infinity and therefore we just get a better cooling, meaning that we have Robin boundary conditions with a possibly larger different boundary coefficient. It is worth noting that the cooling can only get better, not worse.

The first case shows that in a way, the Robin problem behaves very similar to the Dirichlet problem, where we get convergence in the operator norm and therefore convergence of the spectrum. This is in sharp contrast to the Neumann problem. The last two phenomena are *boundary homogenisation results*, where we get the *effective boundary conditions* in the limit. Our exposition follows [51]. In parts we make stronger assumptions to avoid lengthy technical proofs. For periodic oscillations, using very different techniques, other boundary homogenisation results complementing ours are proved in [21,36,71,109] with very different methods. Problems with small holes (obstacles) were considered in [105,106,122,123]. There are applications to problems with nonlinear boundary conditions in [16].

According to Table 2.2 we can set $\lambda_0 = \lambda_A$ given by (5.1.8). Therefore, by Theorem 2.2.2, the Robin problem (6.1.1) is uniquely solvable for all $\lambda \ge \lambda_A$ and all $f_n \in L_p(\Omega_n)$ if $p \ge 2N/(N+1)$. Let $i_{\Omega_n}(f)$ be the extension of $f \in L_p(\Omega)$ by zero outside Ω_n and $r_{\Omega_n}(f)$ the restriction of $f \in L_p(\mathbb{R}^N)$ to Ω_n . Finally let A_n be the operator induced by the form $a_n(\cdot, \cdot)$ induced by (6.1.1). We let $R_n(\lambda)$ and $R(\lambda)$ be the pseudo-resolvents associated with the problems $(\mathcal{A}_n, \mathcal{B}_n)$ and the limit problem $(\mathcal{A}, \mathcal{B})$ as in Definition 2.5.2. In all three cases we show that, if $p \in (1, \infty)$, then

$$R_n(\lambda) \to R(\lambda)$$

in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, m(p))$, where m(p) is given by

$$m(p) := \begin{cases} Np(N-p)^{-1} & \text{if } p \in (1, N), \\ \infty & \text{if } p > N. \end{cases}$$
(6.1.3)

Since $R(\lambda)$ is compact, Theorem 4.3.4 shows that the above is equivalent to

$$\lim_{n\to\infty}R_n(\lambda)f_n=R(\lambda)f$$

in $L_q(\mathbb{R}^N)$ for all $q \in [1, m(p))$, whenever $f_n \rightarrow f$ weakly in $L_p(\mathbb{R}^N)$. The latter property is essential for dealing with semi-linear problems.

After proving some preliminary results we devote to each of the three cases a separate section, where we give a precise statement and a proof of the convergence theorems.

6.2. Preliminary results

To prove the three results mentioned in the previous section we make use of some preliminary results. They are about convergence to a solution of the limit problem on Ω without consideration of boundary conditions.

Given open sets Ω_n we work with functions in $u_n \in H^1(\Omega_n)$ and look at their convergence properties in an open set Ω . We do not assume that Ω_n has an extension property. Even if it has, then its norm does not need to be uniformly bounded with respect to $n \in \mathbb{N}$. Instead we extend u_n and ∇u_n by zero outside Ω_n and consider convergence in $L_2(\mathbb{R}^N)$. We denote these functions by \tilde{u}_n and $\tilde{\nabla u}_n$. The argument used is very similar, but slightly more complicated than the one given on the proof of Proposition 5.5.1. The complication arises because in the present case u_n cannot be considered as an element of $H^1(\mathbb{R}^N)$.

PROPOSITION 6.2.1. Suppose that $\Omega_n, \Omega \subset \mathbb{R}^N$ satisfy Assumption 5.2.2 and \mathcal{A}_n Assumption 5.1.1. Furthermore, suppose that $u_n \in H^1(\Omega_n)$ are weak solutions of $\mathcal{A}_n u_n = f_n$ with $f_n \in L_2(\mathbb{R}^N)$ and $f_n \rightharpoonup f$ weakly in $L_2(\mathbb{R}^N)$. If $||u_n||_{H^1}$ is uniformly bounded, then there exists a subsequence (u_{n_k}) and $u \in H^1(\Omega)$ such that $\tilde{u}_{n_k} \rightharpoonup u$ and $\widetilde{\nabla u}_{n_k} \rightharpoonup \nabla u$ weakly in $L_2(\Omega)$ and $L_2(\Omega, \mathbb{R}^N)$, respectively. Moreover, u is a weak solution of $\mathcal{A}u = f$ in Ω .

PROOF. By assumption $||u_n||_{H^1}$ is uniformly bounded, and so the functions u_n and ∇u_n , extended by zero outside Ω_n are bounded sequences in $L_2(\mathbb{R}^N)$. Hence there exists a subsequence such that $u_{n_k} \rightarrow u$ weakly in $L_2(\Omega)$ and $\nabla u_{n_k} \rightarrow v$ weakly in $L_2(\Omega, \mathbb{R}^N)$. Renumbering the subsequence we assume that (u_n) and (∇u_n) converge. Fix now $\varphi \in C_c^{\infty}(\Omega)$. By assumption and Proposition 5.3.3 there exists $\varphi_n \in C_c^{\infty}(\Omega_n \cap \Omega)$ such that $\varphi_n \rightarrow \varphi$ in $H^1(\Omega)$. As $u_n, \varphi_n \in H^1(\Omega_n)$

$$\int_{\Omega} \nabla u_n \varphi_n \, dx = \int_{\Omega_n} \nabla u_n \varphi_n \, dx = -\int_{\Omega_n} u_n \nabla \varphi_n \, dx = -\int_{\Omega} u_n \nabla \varphi_n \, dx$$

for all $n \in \mathbb{N}$. Because \tilde{u}_n , $\widetilde{\nabla u}_n$ converge weakly and φ_n , $\nabla \varphi_n$ strongly we can pass to the limit and get

$$\int_{\Omega} v\varphi \, dx = -\int_{\Omega} u\nabla\varphi \, dx.$$

As $\varphi \in C_c^{\infty}(\Omega)$ was arbitrary, this means that $v = \nabla u$ is the weak gradient of uin Ω , so $u \in H^1(\Omega)$. Next we observe that, as in the proof of (5.1.10), we have $a_n(u_n, \varphi_n) \to a(u, \varphi)$. We know that $a_n(u_n, \varphi_n) = \langle f_n, \varphi_n \rangle$, and passing to the limit $a(u, \varphi) = \langle f, \varphi \rangle$. Hence Au = f as claimed. \Box

The above is a very weak result. In particular, there are no assumptions on Ω_n outside Ω . If we add some more assumptions we get strong convergence.

PROPOSITION 6.2.2. In addition to the assumptions of Proposition 6.2.1, suppose that (u_n) is bounded in $L_r(\mathbb{R}^N)$ for some r > 2. Moreover, suppose that for every $\varepsilon > 0$ there exists a compact set $K \subset \Omega$ and $n_0 \in \mathbb{N}$ such that $K \subset \Omega_n$ and $|(\Omega_n \cup \Omega) \setminus K| < \varepsilon$ for all $n > n_0$. Then there exists a subsequence (\tilde{u}_{n_k}) such that $\tilde{u}_{n_k} \to u$ in $L_q(\mathbb{R}^N)$ for all $q \in [2, r)$, and u = 0 almost everywhere in Ω^c .

PROOF. Proposition 6.2.1 guarantees that a subsequence of (\tilde{u}_n) converges weakly in $L_2(\mathbb{R}^N)$ to a solution of Au = f. Denote that subsequence again by (\tilde{u}_n) . We show that convergence takes place in the L_2 -norm. Because (\tilde{u}_n) is also bounded in $L_r(\mathbb{R}^N)$ we can also assume that the subsequence converges weakly in $L_r(\mathbb{R}^N)$, or weakly* if $r = \infty$. Hence $u \in L_r(\mathbb{R}^N)$.

Fix $\varepsilon > 0$ and a compact set $K \subset \Omega$ and $n_1 \in \mathbb{N}$ such that $K \subset \Omega_n$ and $|(\Omega_n \cup \Omega) \setminus K| < \varepsilon/2$ for all $n > n_1$. Then choose an open set $U \subset K$ with $|(\Omega_n \cup \Omega) \setminus U| < \varepsilon$ for all $n > n_1$. By assumption (\tilde{u}_n) is bounded in $H^1(K)$ and so Rellich's theorem implies that $\tilde{u}_n \to u$ in $L_2(U)$. Hence there exists $n_2 \in \mathbb{N}$ such that $\|\tilde{u}_n - u\|_{L_2(U)} < \varepsilon$ for all $n > n_2$. Using that $u_n \in L_r(\mathbb{R}^N)$ we get by Hölder's inequality

$$\begin{split} \|\tilde{u}_n - u\|_2 &= \|\tilde{u}_n - u\|_{L_2(U)} + \|\tilde{u}_n - u\|_{L_2((\Omega_n \cup \Omega) \setminus U)} \\ &< \varepsilon + \left| (\Omega_n \cup \Omega) \setminus U \right|^{1/2 - 1/r} (\|\tilde{u}_n\|_r + \|u\|_r) < \varepsilon + \varepsilon^{1/2 - 1/r} (\|\tilde{u}_n\|_r + \|u\|_r) \end{split}$$

for all $n > n_0 := \max\{n_1, n_2\}$. By the uniform bound on $||u_n||_r$ we conclude that $\tilde{u}_n \to u$ in $L_2(\mathbb{R}^N)$. Since $\Omega^c \subset U^c$ the above argument also shows that

$$||u_n||_{L_2(\Omega^c)} \le ||u_n||_{L_2((\Omega_n \cup \Omega) \setminus U)} < \varepsilon^{1/2 - 1/r} ||u_n||_r$$

for all $n > n_0$. Hence $\tilde{u}_n \to 0$ in $L_2(\Omega^c)$ and thus u = 0 on Ω^c almost everywhere. By the uniform bound on $||u_n||_r$ and interpolation

$$\|\tilde{u}_n - u\|_q \le \|\tilde{u}_n - u\|_2^{\theta} \|\tilde{u}_n - u\|_r^{1-\theta} \to 0$$

for $q \in [2, r)$, where $\theta = \frac{2(r-q)}{q(r-2)}$. Hence $\tilde{u}_n \to u$ in $L_q(\mathbb{R}^N)$ for all $q \in [2, \infty)$.

We now use the a priori estimates for the solutions of the Robin problem to verify the boundedness assumptions made in the previous propositions.

COROLLARY 6.2.3. Suppose Ω_n , Ω are bounded Lipschitz domains p > N and $f_n \rightharpoonup f$ in $L_p(\mathbb{R}^N)$. Let u_n be the weak solution of (6.1.1). If $\lambda \ge \lambda_0$, then there exists a constant M independent of $n \in \mathbb{N}$ such that

$$\|u_n\|_{V_n} + \|u_n\|_{\infty} \le M. \tag{6.2.1}$$

Moreover, if $\Omega \subset \Omega_n$ for all $n \in \mathbb{N}$ and $|\Omega_n \setminus \Omega| \to 0$, then there exists a subsequence (u_{n_k}) of (u_n) converging to a weak solution of

$$Au + \lambda u = f$$

in $L_q(\mathbb{R}^N)$ for all $q \in [2, \infty)$. Finally, u = 0 on $\overline{\Omega}^c$ almost everywhere.

PROOF. By Theorem 2.4.1 with constants from Table 2.2 there exists a constant *C* only depending on *N*, *p* and an upper bound for $|\Omega_n|$ such that

$$||u_n||_{\infty} \le C \max\left\{\frac{1}{\alpha_0}, \frac{1}{\beta}\right\} ||f_n||_p.$$
 (6.2.2)

Similarly, using the norm

$$\|u_n\|_{V_n} := \left(\|\nabla u_n\|_2 + \int_{\partial\Omega_n} u_n^2 \, d\sigma\right)^{1/2}$$

we get from Theorem 2.2.2 that

$$|u_n||_{V_n} \le C_1 ||f_n||_{V'_n} \le C_2 ||f_n||_p$$

for all $n \in \mathbb{N}$ with constants C_1, C_2 independent of $n \in \mathbb{N}$. Using that weakly convergent sequences in $L_p(\mathbb{R}^N)$ are bounded we get the existence of a constant M independent of $n \in \mathbb{N}$ such that (6.2.1) holds for all $n \in \mathbb{N}$. The second part follows from Proposition 6.2.2 because Assumption 5.2.2 and all other assumptions are clearly satisfied.

6.3. Small modifications of the original boundary

Without further mentioning we use the notation and setup from Section 6.1 and suppose (A_n, B_n) satisfy Assumption 6.1.1. We look at a situation where the original boundary remains largely unperturbed. How much boundary we add outside the domain or inside as holes is almost irrelevant. What we mean by small modifications of $\partial \Omega$ we specify as follows.

ASSUMPTION 6.3.1. Suppose that $\Omega_n \subset \mathbb{R}^N$ are bounded open sets satisfying a Lipschitz condition. Let Ω be an open set and $K \subset \overline{\Omega}$ be a compact set of capacity zero such that for every neighbourhood U of K there exists $n_0 \in \mathbb{N}$ such that

$$\Omega \cap (\Omega_n \cap (\Omega \cup U)^c) = \emptyset \quad \text{and} \quad \Omega \subset \Omega_n \cup U \tag{6.3.1}$$

for all $n > n_0$. Moreover, assume that

$$\lim_{n\to\infty}|\Omega_n\cap\Omega^c|=0.$$

Note that the first condition in (6.3.1) means that U allows us to separate $\overline{\Omega}$ from $\overline{\Omega_n} \cap (\Omega \cup U)^c$ which is the part of $\overline{\Omega_n}$ outside $\overline{\Omega}$ as shown in Figure 6.4. Note that the above assumption also allows us to cut holes in Ω shrinking to a set of capacity zero as $n \to \infty$. If (6.3.1) holds, then also

$$\partial \Omega \cap U^c \subset \partial \Omega_n. \tag{6.3.2}$$

This means that $\partial\Omega$ is contained in $\partial\Omega_n$ except for a very small set. To see this let $x \in \partial\Omega \cap U^c$. If W is a small enough neighbourhood of x, then in particular $W \cap \Omega \subset W \cap \Omega_n$ by (6.3.1) and so $W \cap \Omega_n \neq \emptyset$ for every neighbourhood of x. Moreover,

 $\partial \Omega \cap \Omega_n \cap U^c \subset \Omega^c \cap \Omega_n \cap U^c \subset \overline{\Omega_n \cap (\Omega \cup U)^c}$



Fig. 6.4. Separation of the part of Ω_n from $\overline{\Omega}$ by U.

and because $\partial \Omega \subset \overline{\Omega}$ (6.3.1) implies that $\partial \Omega \cap \Omega_n \cap U^c = \emptyset$. Hence (6.3.2) follows. We next use the above to construct a sequence of cutoff functions.

LEMMA 6.3.2. Suppose Ω_n , Ω satisfy Assumption 6.3.1. Then there exists a sequence $\psi_n \in C^{\infty}(\overline{\Omega}) \cap C^{\infty}(\overline{\Omega}_n)$ such that $0 \leq \psi_n \leq 1$, $\psi_n \to 0$ on $\Omega_n \cap \overline{\Omega}^c$ and $\psi_n \to 1$ in $H^1(\Omega)$.

PROOF. Fix a sequence U_k of bounded open sets such that $K \subset \overline{U}_{k+1} \subset U_k$ for all $k \in \mathbb{N}$, that $\bigcap U_k = K$ and that $\operatorname{cap}(\overline{U}_k) \to 0$ converges to zero. We can make such a choice because K is a compact set with $\operatorname{cap}(K) = 0$. By (6.3.1), for every $k \in \mathbb{N}$ there exists $n_k \in \mathbb{N}$ such that

$$\overline{\Omega} \cap (\overline{\Omega_n \cap (\Omega \cup U_k)^c}) = \emptyset$$
 and $\Omega \subset \Omega_n \cup U_k$

for all $n > n_k$. Let now $\theta_k \in C_c^{\infty}(U_{k+1})$ be such that $0 \le \theta_k \le 1$ and $\theta_k = 1$ on \overline{U}_k . Since $\operatorname{cap}(\overline{U}_k)$ converges to zero we can make that choice such that $\theta_k \to 0$ in $H^1(\mathbb{R}^N)$. For $n_k < n \le n_{k+1}$ we now set

$$\psi_n(x) := \begin{cases} 1 - \theta_k(x) & \text{if } x \in \bar{\Omega} \cup \bar{U}_k, \\ 0 & \text{if } x \in (\bar{\Omega} \cup \bar{U}_k)^c. \end{cases}$$

By choice of n_k and θ_k the function ψ_n is well defined with the required properties.

We now prove the main theorem of this section.

THEOREM 6.3.3. Suppose Ω_n are bounded Lipschitz domains satisfying Assumption 6.3.1 and that $b_{0n} \rightarrow b_0$ weakly in $L_r(\partial \Omega)$ for some $r \in (1, \infty)$ or weakly^{*} in $L_{\infty}(\partial \Omega)$. If $1 and <math>\lambda \in \varrho(-A)$, then for n large enough $\lambda \in \varrho(-A_n)$ and $R_n(\lambda) \rightarrow R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, m(p))$. Here m(p) is given by (6.1.3) and $R(\lambda)$ is the resolvent operator associated with the problem

$$\mathcal{A}u + \lambda u = f \qquad \text{in } \Omega,$$

$$\frac{\partial}{\partial v_A} u + b_0 u = 0 \qquad \text{on } \partial \Omega.$$
 (6.3.3)

PROOF. We only need to consider p > N and $\lambda > \lambda_A$ with λ_A defined by (5.1.8). Given that $f_n \rightarrow f$ weakly in $L_p(\mathbb{R}^N)$, we show that $R_n(\lambda)f_n \rightarrow R(\lambda)f$ in $L_p(\mathbb{R}^N)$. Since $R(\lambda)$ is compact Theorem 4.3.4 and the uniform a priori estimates from Theorem 2.4.1 in conjunction with Table 2.2 complete the proof.

Let p > N, $\lambda > \lambda_A$ and suppose that $f_n \rightharpoonup f$ weakly in $L_p(\mathbb{R}^N)$. Set $u_n := R(\lambda)f_n$ and $u := R(\lambda)f$. Then Corollary 6.2.3 implies the uniform bound (6.2.1). Also note that Assumption 6.3.1 implies Assumption 5.2.2 and hence Proposition 6.2.2 guarantees that a subsequence of $u_n := R_n(\lambda)f_n$ converges to some $u \in H^1(\Omega)$ in $L_q(\mathbb{R}^N)$ for all $q \in (1, \infty)$. Moreover, u = 0 in $\mathbb{R}^N \setminus \Omega$ almost everywhere and u satisfies $Au + \lambda u = f$ in Ω . Since (6.3.3) has a unique solution, u is the only possible accumulation point for the sequence (u_n) , and therefore the whole sequence converges. To simplify notation we therefore assume that u_n converges.

We assumed that Ω_n are Lipschitz domains and that for every open set U containing Kthe set boundary $\partial\Omega$ is contained in $\partial\Omega_n \cup U$ for n large enough. Hence $\partial\Omega$ satisfies a Lipschitz condition except possibly at $K \cap \partial\Omega$. Because the latter set has capacity zero the restrictions of $C_c^{\infty}(\mathbb{R}^n)$ to Ω are dense in $H^1(\Omega)$. Fix $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ and set $\varphi_n := \psi_n \varphi$ with ψ_n the cutoff functions from Lemma 6.3.2. Then by construction $\varphi_n \in H^1(\Omega_n) \cap H^1(\Omega)$ with $\varphi_n \to \varphi$ in $H^1(\Omega)$. In particular

$$a_n(u_n,\varphi_n) + \lambda \langle u_n,\varphi_n \rangle = \langle f_n,\varphi_n \rangle$$

for all $n \in \mathbb{N}$. Clearly $\langle u_n, \varphi_n \rangle \to \langle u, \varphi \rangle$ and $\langle f_n, \varphi_n \rangle \to \langle f, \varphi \rangle$, so we only have to deal with $a_n(u_n, \varphi_n)$. An argument similar to that used in the proof of Proposition 5.1.2 shows that

$$\lim_{n \to \infty} a_{0n}(u_n, \varphi_n) = a_0(u, \varphi_n)$$

where $a_{0n}(\cdot, \cdot)$ is the form corresponding to $(\mathcal{A}_n, \mathcal{B}_n)$ excluding the boundary integral as in Definition 2.1.2. We only need to show that

$$\lim_{n \to \infty} \int_{\partial \Omega_n} b_{0n} u_n \varphi_n \, d\sigma = \lim_{n \to \infty} \int_{\partial \Omega} b_{0n} u_n \varphi_n \, d\sigma = \int_{\partial \Omega} b_0 u \varphi \, d\sigma \tag{6.3.4}$$

for all $\varphi \in C_c^{\infty}(\mathbb{R}^N)$. The first equality is because of the properties of the cutoff functions ψ_n , so we prove the second one. Fix an open set U such that $K \cap \partial \Omega \subset U$. Because $\partial \Omega \cap U^c$ is Lipschitz, the trace operator from $H^1(\Omega)$ into $L_2(\partial \Omega \cap U^c)$ is compact (see [104, Théorème 2.6.2]). As $u_n \rightarrow u$ weakly in $H^1(\Omega)$ the corresponding traces converge in $L_2(\partial \Omega \cap U^c)$ and by the uniform bound (6.2.1) and interpolation in $L_s(\partial \Omega \cap U^c)$ for all $s \in [1, \infty)$. By construction, for big enough n, we have $\varphi_n = \varphi$ on $\partial \Omega \cap U^c$. Hence by the assumptions on b_{0n}

$$\lim_{n\to\infty}\int_{\partial\Omega\cap U^c}b_{0n}u_n\varphi_n\,d\sigma=\lim_{n\to\infty}\int_{\partial\Omega\cap U^c}b_{0n}u_n\varphi\,d\sigma=\int_{\partial\Omega\cap U^c}b_{0}u\varphi\,d\sigma.$$

The assumptions on b_{0n} also imply that (b_{0n}) is bounded in $L_r(\partial \Omega)$, and therefore

$$\left|\int_{\partial\Omega\cap U}b_{0n}u_n\varphi\,d\sigma\right|\leq \|u_n\varphi_n\|_{\infty}\|b_{0n}\|_{L_r(\partial\Omega)}\sigma(\partial\Omega\cap U).$$

Since we can choose the measure $\sigma(\partial \Omega \cap U)$ to be arbitrarily small and $||u_n \varphi_n||_{\infty}$ is uniformly bounded (6.3.4) follows, completing the proof of the theorem.



Fig. 6.5. $\partial \Omega$, $\partial \Omega_n$ are locally graphs with respect to the same coordinate system.

6.4. Boundary homogenisation: Limit is a Dirichlet problem

In this section we look at a sequence of problems of the form (5.1.1), where $\partial \Omega_n$ is different to $\partial \Omega$ on large parts of the domain. We assume that $\partial \Omega_n$ and $\partial \Omega$ are, at least after a change of coordinates, the graph of a function with respect to the same coordinate system for all $n \in \mathbb{N}$ as illustrated in Figure 6.5. We make this more precise as follows.

ASSUMPTION 6.4.1. Let Ω , Ω_n be domains in \mathbb{R}^N such that $\Omega \subset \Omega_n$ and

$$\lim_{n\to\infty}|\Omega_n\setminus\Omega|=0$$

We further assume that for every $x \in \partial \Omega$ there exists a coordinate system with x at the centre and a cylinder $Z = B \times (-a, \infty) \subset \mathbb{R}^{N-1} \times \mathbb{R}$ for some a > 0 such that

$$\Omega_n \cap Z = \{ (x', x_N) \in B \times (-a, \infty) \colon x_N < \varphi_n(x') \}$$

and

$$\Omega \cap Z = \{ (x', x_N) \in B \times (-a, \infty) \colon x_N < \varphi_0(x') \}$$

with $\varphi_n \colon B \to \mathbb{R}$ Lipschitz continuous for all $n \in \mathbb{N}$.

REMARK 6.4.2. Note that $\varphi_n \to \varphi_0$ in $L_1(B)$. To see this note that because $\Omega \subset \Omega_n$ we have $\varphi_n \ge \varphi_0$ and hence

$$\|\varphi_n - \varphi_0\|_{L_1(B)} = \int_B \varphi_n(x') - \varphi_0(x') \, dx' = |Z \cap (\Omega_n \setminus \Omega)| \le |\Omega_n \setminus \Omega| \to 0$$

as $n \to \infty$.

The assumption that $\Omega \subset \Omega_n$ is only for simplicity to avoid overly technical proofs. It is sufficient to assume that for every compact set $K \subset \Omega$ there exists $n_0 \in \mathbb{N}$ such that $K \subset \Omega_n$ for all $n > n_0$. The argument is given in [51, Remark 5.10(a)].

We define $R_n(\lambda)$ and $R(\lambda)$ as in Definition 2.5.2, where $R(\lambda)$ is associated with the Dirichlet problem (5.1.2).

THEOREM 6.4.3. Suppose Ω_n , Ω are bounded Lipschitz domains satisfying Assumption 6.4.1. Moreover, assume that for every $x \in \partial \Omega$ the corresponding functions $\varphi_n \colon B \to \mathbb{R}$ satisfy

$$\lim_{n \to \infty} \left| \{ y \in B \colon |\nabla \varphi_n(y)| < t \} \right| = 0$$

for all t > 0. Finally suppose that $b_{0n} \ge \beta$ for some constant $\beta > 0$. If 1 $and <math>\lambda \in \varrho(-A)$, then for n large enough $\lambda \in \varrho(-A_n)$ and $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, m(p))$. Here $R(\lambda)$ is the resolvent associated with the Dirichlet problem (5.1.2) and m(p) is given by (6.1.3).

PROOF. As in the proof of Theorem 6.3.3 we only need to consider p > N and $\lambda > \lambda_A$, where λ_A is given by (5.1.8). The other cases follow from Theorem 4.3.4 and the uniform a priori estimates from Theorem 2.4.1 in conjunction with Table 2.2. Hence assume that p >N, that $\lambda > \lambda_A$ and that $f_n \rightharpoonup f$ weakly in $L_p(\mathbb{R}^N)$. We set $u_n := R_n(\lambda) f_n$. Convergence of a subsequence to a solution u of $Au + \lambda u = f$ in $L_p(\mathbb{R}^N)$ follows from Corollary 6.2.3. Hence we only need to show that u satisfies Dirichlet boundary conditions, that is, $u \in H_0^1(\Omega)$. Since the Dirichlet problem has a unique solution, u is the only possible accumulation point for the sequence (u_n) , and therefore the whole sequence converges.

The boundary conditions are local, so we only need to look at a neighbourhood of every boundary point. Fix a cylinder Z and functions φ_n as in Assumption 6.4.1. Because the domains are Lipschitz domains, it is sufficient to show that u has zero trace on $\partial \Omega \cap Z$ (see [104, Théorème 2.4.2]). We know that $u_n \rightarrow u$ weakly in $H^1(\Omega \cap Z)$. By the compactness of the trace operator $\gamma \in \mathcal{L}(H^1(\Omega \cap Z), L_2(\partial \Omega \cap Z))$ (see [104, Théorème 2.6.2]) we know that $\gamma(u_n) \rightarrow \gamma(u)$ in $L_2(\partial \Omega \cap Z)$. From now on we simply write u_n, u for the traces. We need to show that

$$\lim_{n \to \infty} \|u_n\|_{L_2(\partial \Omega \cap Z)} = 0. \tag{6.4.1}$$

We do that in two steps. To express the boundary integrals in the coordinates chosen we need the Jacobians

$$g_n := \sqrt{1 + |\nabla \varphi_n|^2}.$$

By Rademacher's theorem (see [68, Section 3.1.2]) the gradient exists almost everywhere. We can write

$$\|u_n\|_{L_2(\partial\Omega\cap Z)} = \left(\int_B \left|u_n(x',\varphi_0(x'))\right|^2 g_0(x') \, dx'\right)^{1/2} \tag{6.4.2}$$

$$\leq \|g_0\|_{\infty}^{1/2} \left(\int_B \left| u_n(x',\varphi_n(x')) - u_n(x',\varphi_0(x')) \right|^2 dx' \right)^{1/2}$$
(6.4.3)

$$+ \|g_0\|_{\infty}^{1/2} \left(\int_B \left| u_n(x', \varphi_n(x')) \right|^2 dx' \right)^{1/2}.$$
 (6.4.4)

We show separately that each term on the right-hand side of the above inequality converges to zero as $n \to \infty$. For the first term we use Fubini's theorem and the fundamental theorem of calculus to write

$$\int_{B} \left| u_{n}(x',\varphi_{n}(x')) - u_{n}(x',\varphi_{0}(x')) \right| dx' = \int_{B} \left| \int_{\varphi_{0}(x')}^{\varphi_{n}(x')} \frac{\partial}{\partial x_{N}} u_{n}(x',x_{N}) dx_{N} \right| dx'$$

$$\leq \int_{B} \int_{\varphi_{0}(x')}^{\varphi_{n}(x')} \left| \nabla u_{n}(x',x_{N}) \right| dx_{N} dx' \leq |A_{n}|^{1/2} \| \nabla u_{n} \|_{L_{2}(\Omega \cap Z)}, \tag{6.4.5}$$

where A_n is the region between graph(φ_0) and graph(φ_n) as shown in Figure 6.5. By assumption $|\Omega_n \setminus \Omega| \to 0$ and so $|A_n| \to 0$. Furthermore, by (6.2.1) and the definition of u_n the sequence $\|\nabla u_n\|_{L_2(\partial\Omega\cap Z)}$ is bounded. Since u_n is not necessarily continuously differentiable, (6.4.5) needs to be justified. First note that u_n is continuous on $Z \cap \Omega_n$ (see [76, Theorem 8.24]). Because Ω_n is Lipschitz, u_n can be extended to a function in $H^1(Z)$. Since such functions can be represented by a function such that $u_n(x', \cdot)$ is absolutely continuous in the coordinate directions, we can indeed apply the fundamental theorem of calculus as done above (see [101, Section 1.1.3] or [68, Section 4.9.2]). Because

$$\left(\int_{B} \left| u_{n}(x',\varphi_{n}(x')) - u_{n}(x',\varphi_{0}(x')) \right|^{2} dx' \right)^{1/2}$$

$$\leq 2 \|u_{n}\|_{\infty}^{1/2} \left(\int_{B} \left| u_{n}(x',\varphi_{n}(x')) - u_{n}(x',\varphi_{0}(x')) \right| dx' \right)^{1/2}$$

the bound (6.2.1) implies that (6.4.3) converges to zero.

We next prove that (6.4.4) converges to zero as well. For t > 0 we set

 $[g_n \ge t] := \{y \in B : g_n(y) \ge t\}$ and $[g_n < t] := \{y \in B : g_n(y) < t\}$

and write

$$\begin{split} \int_{B} & |u_{n}(x',\varphi_{n}(x'))|^{2} dx' \\ & = \int_{[g_{n} \leq t]} |u_{n}(x',\varphi_{n}(x'))|^{2} dx' + \int_{[g_{n} > t]} |u_{n}(x',\varphi_{n}(x'))|^{2} dx'. \end{split}$$

By (6.2.1) there exists a constant $M_1 > 0$ such that

$$\int_{[g_n \le t]} |u_n(x', \varphi_n(x'))|^2 \, dx' \le ||u_n||_{\infty}^2 |[g_n \le t]| \le M_1 |[g_n \le t]|$$

and that

$$\begin{split} &\int_{[g_n>t]} \left| u_n(x',\varphi_n(x')) \right|^2 dx' \\ &\leq \frac{1}{t} \int_B \left| u_n(x',\varphi_n(x')) \right|^2 g_n(x') dx' \leq \frac{1}{t} \| u_n \|_{L_2(\partial\Omega\cap Z)}^2 \leq \frac{M_1}{t} \end{split}$$

for all t > 0. Therefore

$$\int_{B} \left| u_n(x',\varphi_n(x')) \right|^2 dx' \leq M_1 \left(\left| [g_n \leq t] \right| + \frac{1}{t} \right).$$



Fig. 6.6. Perturbation of a disc with limit problem a Robin problem.

Fix now $\varepsilon > 0$ and choose t > 0 such that $M_1/t < \varepsilon/2$. By assumption there exists $n_0 \in \mathbb{N}$ such that $|[g_n < t]| < \varepsilon/2M_1$ for all $n > n_0$. Hence from the above

$$\int_{B} \left| u_{n}(x',\varphi_{n}(x')) \right|^{2} dx' < \varepsilon$$

for all $n > n_0$. Therefore (6.4.4) converges to zero as claimed and (6.4.1) follows.

REMARK 6.4.4. (a) The assumption that $\partial \Omega_n$ is a graph over $\partial \Omega$ is essential in the above theorem. The domain shown in Figure 6.6 has a very fast oscillating outside boundary. If we shrink the connection from the outside ring with the disc to a point and the ring itself to a circle, then the assumptions of Theorem 6.3.3 are satisfied and the limit problem is the Robin problem (6.3.3).

(b) It is possible to work with diffeomorphisms flattening the boundary locally as shown in Figure 6.7, but the proof is more complicated. We refer to [51] for details.

6.5. Boundary homogenisation: Limit is a Robin problem

As in Section 6.4 we look at oscillating boundaries. The oscillations however are slower, and it turns out that the limit problem of (5.1.1) is a Robin problem of the form

$$\mathcal{A}u + \lambda u = f \quad \text{in } \Omega,$$

$$\frac{\partial}{\partial \nu_{\mathcal{A}}} u + g b_0 u = 0 \quad \text{on } \partial \Omega$$
(6.5.1)

with first order coefficient gb_0 rather than just b_0 . Now the resolvent $R(\lambda)$ is the pseudoresolvent associated with (6.5.1). Suppose that Assumption 6.4.1 is satisfied and that φ_n are the functions associated with the parametrisation of $\partial \Omega$ and $\partial \Omega_n$. The function g is associated with the limit of the Jacobians

$$g_n := \sqrt{1 + |\nabla \varphi_n|^2}.$$

The assumption that $\Omega \subset \Omega_n$ is again only for simplicity, see [51, Remark 5.10(a)] on how to overcome it. The assumption that $\partial \Omega_n$ is a graph over $\partial \Omega$ is again essential with a similar example as in Remark 6.4.4(a). This can, in a generalised sense, be as shown in Figure 6.7 (see [51] for details).



Fig. 6.7. Simultaneously parametrised domains.

THEOREM 6.5.1. Suppose Ω_n , Ω are bounded Lipschitz domains satisfying Assumption 6.4.1 and that $b_0 \in C(\mathbb{R}^N)$. Set $b_{0n} := b_0|_{\partial\Omega_n}$ with $b_0 \ge \beta_0$ for some constant $\beta > 0$. Moreover, assume that $g \in L_{\infty}(\partial\Omega)$ such that for every $x \in \partial\Omega$ the corresponding functions $\varphi_n : B \to \mathbb{R}$ satisfy

$$\lim_{n \to \infty} \frac{\sqrt{1 + |\nabla \varphi_n|^2}}{\sqrt{1 + |\nabla \varphi_0|^2}} = \lim_{n \to \infty} \frac{g_n}{g_0} = g$$

weakly in $L_r(B)$ for some $r \in (1, \infty)$, weakly^{*} in $L_{\infty}(B)$ or strongly in $L_1(B)$. If $1 and <math>\lambda \in \varrho(-A)$, then for n large enough $\lambda \in \varrho(-A_n)$ and $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [1, m(p))$. Here $R(\lambda)$ is the resolvent associated with the Robin problem (6.5.1) and m(p) is given by (6.1.3).

PROOF. As in the previous cases considered we only need to look at p > N and $\lambda > \lambda_A$ with λ_A defined by (5.1.8). Given that $f_n \rightharpoonup f$ weakly in $L_p(\mathbb{R}^N)$, we show that $R_n(\lambda) f_n \rightarrow R(\lambda) f$ in $L_p(\mathbb{R}^N)$. Since $R(\lambda)$ is compact, Theorem 4.3.4 and the uniform a priori estimates from Theorem 2.4.1 in conjunction with Table 2.2 complete the proof. Assume that p > N, that $\lambda > \lambda_A$ and that $f_n \rightharpoonup f$ weakly in $L_p(\mathbb{R}^N)$. We set $u_n := R_n(\lambda) f_n$. By Corollary 6.2.3 the sequence u_n is bounded in $H^1(\Omega) \cap L_{\infty}(\mathbb{R}^N)$. It has a subsequence converging to some function $u \in H^1(\Omega)$ in $L_p(\mathbb{R}^N)$. Since (6.5.1) has a unique solution it follows that the whole sequence converges if we can show that usolves (6.5.1). To simplify notation we assume that u_n converges.

We assumed that Ω is a Lipschitz domain and therefore the restrictions of functions in $C_c^{\infty}(\mathbb{R}^n)$ to Ω are dense in $H^1(\Omega)$. An argument similar to that used in the proof of Proposition 5.1.2 shows that

$$\lim_{n \to \infty} a_{0n}(u_n, \psi) = a_0(u, \psi)$$

where $a_{0n}(\cdot, \cdot)$ is the form corresponding to $(\mathcal{A}_n, \mathcal{B}_n)$ excluding the boundary integral as in Definition 2.1.2. We only need to show that

$$\lim_{n\to\infty}\int_{\partial\Omega_n}b_{0n}u_n\psi\,d\sigma=\int_{\partial\Omega}b_0gu\psi\,d\sigma.$$

By a partition of unity it is sufficient to consider $\psi \in C_c^{\infty}(\mathbb{R}^N)$ in a cylinder $Z = B \times (-a, \infty)$ as in Assumption 6.4.1. In these coordinates the above becomes

$$\lim_{n \to \infty} \int_{B} (b_{0}u_{n}\psi)(x',\varphi_{n}(x'))g_{n}(x') dx' = \int_{B} (b_{0}gu\psi)(x',\varphi_{0}(x'))g_{0}(x') dx',$$
(6.5.2)

where $(b_0 u_n \psi)(x', \varphi_n(x'))$ is the product of the functions $b_0 u_n \psi$ evaluated at the point $(x', \varphi_n(x'))$. First note that

$$u_n(x',\varphi_n(x')) \rightarrow u(x',\varphi_0(x'))$$

in $L_1(B)$ by a similar argument as the one used to prove that (6.4.3) converges to zero. The product $b_0\psi$ is continuous and bounded on *B* and therefore the corresponding superposition operator on $L_1(B)$ is continuous (see [6, Theorem 3.1 and 3.7]). Since $\varphi_n \rightarrow \varphi$ in $L_1(B)$ by Remark 6.4.2 we therefore conclude that

$$(b_0\psi)(x',\varphi_n(x')) \rightarrow (b_0\psi)(x',\varphi_0(x'))$$

in $L_1(B)$. Since b_0 , ψ and u_n are bounded in $L_{\infty}(B)$ it follows that

$$(b_0 u_n \psi)(x', \varphi_n(x')) \rightarrow (b_0 u \psi)(x', \varphi_0(x'))$$

in $L_s(B)$ for all $s \in [1, \infty)$. By assumption

$$g_n(x') = \frac{g_n(x')}{g_0(x')} g_0(x') \rightharpoonup g(x, \varphi_0(x')) g_0(x')$$

weakly in $L_r(B)$ for some $r \in (1, \infty)$, weakly^{*} in $L_\infty(B)$ or strongly in $L_1(B)$. If we combine everything, then (6.5.2) follows.

REMARK 6.5.2. (a) From the above the function new weight g is larger than or equal to one, so $b_0g \ge b_0$ always. This reflects the physical description mentioned in Section 6.1, where we argued that a larger surface area of the oscillating boundary will lead to better cooling. We cannot approach a Neumann problem for that reason. The best we can do is to have g = 1. If the oscillations are too fast, then heuristically we have " $g = \infty$ " and the limit problem is a Dirichlet problem as in Theorem 6.4.3.

(b) Given a Lipschitz domain it is possible to construct a sequence of C^{∞} domains satisfying the assumptions of the above theorem in such a way that g = 1, that is, the boundary conditions of the limit domain are unchanged. For details see Section 8.3.

7. Neumann problems on varying domains

7.1. Remarks on Neumann problems

We saw in Section 2.4.3 that there are no smoothing properties for the Neumann problem uniformly with respect to the domains. This makes dealing with Neumann boundary

conditions rather more difficult. In particular, we saw that for Dirichlet and Robin problems the resolvent operators converge in the operator norm. This is not in general the case for Neumann problems. In particular, we cannot expect the spectrum to converge as in the case of the other boundary conditions, which means the resolvent operator only converges strongly (that is, pointwise) in $\mathcal{L}(L_2(\mathbb{R}^N))$. In this exposition we only prove a result similar to those for Dirichlet and Robin problems. It is beyond the scope of these notes to give a comprehensive treatment of the other phenomena. We refer to the literature, in particular to the work of Arrieta [13,17], the group of Bucur, Varchon and Zolésio [29,30] with necessary and sufficient conditions for domains in the plane in [28]. There is other work by Jimbo [87–90] and references therein. Other references include [32,33,47,49,81].

7.2. Convergence results for Neumann problems

Given open sets $\Omega_n \subset \mathbb{R}^N$ $(N \ge 2)$ we consider convergence of solutions of the Neumann problems

$$\begin{aligned} \mathcal{A}_n u + \lambda u &= f_n \quad \text{in } \Omega_n, \\ \frac{\partial}{\partial \nu_{\mathcal{A}}} u &= 0 \quad \text{on } \partial \Omega_n \end{aligned}$$
(7.2.1)

to a solution the Neumann problem

$$\mathcal{A}u + \lambda u = f \qquad \text{in } \Omega,$$

$$\frac{\partial}{\partial v_{\mathcal{A}}} u = 0 \qquad \text{on } \partial \Omega_n$$
(7.2.2)

on a domain Ω as $n \to \infty$. On the operators \mathcal{A}_n , \mathcal{A} we make the same assumptions as in the case of the Dirichlet and Robin problems, namely those stated in Assumption 5.1.1. To get a result in the spirit of the others proved so far we make the following assumptions on the domains. The conditions are far from optimal, but given the difficulties mentioned in Section 7.1, we refer to the literature cited there for more general conditions.

ASSUMPTION 7.2.1. Suppose that Ω_n , Ω are bounded open sets with the following properties.

- (1) There exists a compact set $K \subset \overline{\Omega}$ of capacity zero such that for every neighbourhood U of K there exists $n_0 \in \mathbb{N}$ with $\Omega \subset \Omega \cup U$ for all $n > n_0$.
- (2) $|\Omega_n \cap \Omega^c| \to 0 \text{ as } n \to \infty.$
- (3) $\{u|_{\Omega}: u \in C_c^{\infty}(\mathbb{R}^N)\}$ is dense in $H^1(\Omega)$.
- (4) There exists d > 2, $c_a > 0$ and $\lambda_0 \ge 0$ such that

$$u\|_{2d/(d-2)}^2 \le c_a(a_n(u, u) + \lambda_0 \|u\|_2)$$

for all $u \in H^{1}(\Omega_{n})$ and all $n \in \mathbb{N}$. Here $a_{n}(\cdot, \cdot)$ is given as in Definition 2.1.2 without the boundary integral.

REMARK 7.2.2. (a) Condition (1) allows to cut holes into Ω shrinking to a set K of capacity zero. The holes cannot be arbitrary since otherwise (4) is violated. However

Domain perturbation for linear and semi-linear boundary value problems



Fig. 7.1. Approximation of touching spheres, preserving a uniform cone condition.

if the holes have a fixed shape and are just contracted by a scalar factor, then (4) is satisfied (see [53, Section 2]).

(b) Condition (4) is satisfied if for instance all domains Ω_n , Ω satisfy a cone condition uniformly with respect to $n \in \mathbb{N}$, that is, the angle and length of the cone defining the cone condition is the same for all $n \in \mathbb{N}$ (see [2, Lemma 5.12]). But as the example with the holes in (a) shows this is only a sufficient condition for (4). Such an approach was used in [37] for instance.

(c) We might think that under the above condition we can have examples like the dumbbell shaped domains in Figure 6.1. As it turns out, condition (4) cannot be satisfied for such a case, because (4) implies convergence in the operator norm as we prove below. Convergence in the operator norm, by Corollary 4.3.2, every finite system of eigenvalues converges, but for dumbbell shaped domains or other exterior perturbations this is not the case (see [13,87]). However, if we replace the dumbbell by two touching balls opened up slightly near the touching balls, then we can ensure that a uniform cone condition is satisfied. Moreover, (3) holds because the union of two balls has a smooth boundary except at a set of capacity zero, where the balls touch (See Figure 7.1). The other conditions are obviously also satisfied.

(d) To get convergence of solutions (but not necessarily of the spectrum) we could work with conditions similar to the Mosco conditions stated in Assumption 5.2.1 and 5.2.2 in the case of the Dirichlet problem. The conditions are explicitly used and stated in [28, Section 2].

We define the resolvent operators $R_n(\lambda)$ and $R(\lambda)$ as in Definition 2.5.2 with $(\mathcal{A}_n, \mathcal{B}_n)$ and $(\mathcal{A}, \mathcal{B})$ being the operators associated with (7.2.1) and (7.2.2), respectively. By Theorem 2.2.2, the Neumann problem (6.1.1) is uniquely solvable for all $\lambda \ge \lambda_0 :=$ $\lambda_{\mathcal{A}} + \alpha_0/2$ and all $f_n \in L_2(\Omega_n)$. This means $R_n(\lambda)$ and $R(\lambda)$ is well defined for $n \ge n_0$.

THEOREM 7.2.3. Suppose that Assumption 7.2.1 holds. If $\lambda \in \varrho(-A)$, then $\lambda \in \varrho(A_n)$ for all *n* large enough. Moreover, for every $p \in (1, \infty)$ we have $R_n(\lambda) \to R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N))$ for all $q \in [p, m(p))$, where m(p) is defined by (2.4.5).

PROOF. Fix $\lambda \ge \lambda_A + \alpha_0/2$ and let $f_n \rightharpoonup f$ weakly in $L_2(\mathbb{R}^N)$. Set $u_n := R_n(\lambda) f_n$ and $u := R(\lambda) f$. Note that by assumption and (2.4.7) the operator $R(\lambda)$ is compact. By Theorem 4.3.4 and (2.4.2) it is therefore sufficient to prove that $u_n \rightarrow u$ in $L_2(\mathbb{R}^N)$. To do so first note that by Theorem 2.2.2 we have

$$\|u_n\|_{H^1(\Omega_n)} \le \frac{2}{\alpha_0} \|f_n\|_{(H^1(\Omega_n))'} \le \frac{2}{\alpha_0} \|f_n\|_{L_2(\Omega_n)}$$

for all $n \in \mathbb{N}$. Hence $||u_n||_{H^1(\Omega)}$ is uniformly bounded. Hence there is a subsequence (u_{n_k}) such that (\tilde{u}_{n_k}) and also $(\widetilde{\nabla}u_{n_k})$ converge weakly in $L_2(\mathbb{R}^N)$. Here \tilde{u}_{n_k} and $\widetilde{\nabla}u_{n_k}$ are the extensions of u_{n_k} and ∇u_{n_k} by zero outside Ω_n . If we can show that $u \in H^1(\Omega)$ and that u solves (7.2.2), then the whole sequence converges since the limit problem admits a unique solution. For simplicity we denote the subsequence chosen again by (u_n) .

First note that our assumptions make it possible to apply Proposition 6.2.2 and thus $u \in H^1(\Omega)$ and $u_n \to u$ in $L_2(\mathbb{R}^N)$. Note also that this function has support in $\overline{\Omega}$. We show that u solves (7.2.2). By assumption $\operatorname{cap}(\overline{\Omega} \cap \Omega_n^c) \to 0$ and so there exists $\varphi_n \in C_c^{\infty}(\mathbb{R}^N)$ such that $0 \leq \varphi_n \leq 1$, $\varphi_n = 1$ in a neighbourhood of $\overline{\Omega} \cap \Omega_n^c$ and $\varphi_n \to 0$ in $H^1(\mathbb{R}^N)$. Let now $\psi \in C_c^{\infty}(\Omega)$ and set $\psi_n := \psi(1 - \psi_n)$. Then $\psi_n \to \psi$ in $H^1(\mathbb{R}^N)$. Now

$$a_n(u_n, \psi_n) + \langle u_n, \psi_n \rangle = \langle f_n, \psi_n \rangle$$

for all $n \in \mathbb{N}$. Every term in the above identity involves a pair of a strongly and a weakly convergent sequence and therefore an argument similar to that in the proof of Proposition 5.1.2 shows that we can pass to the limit to get

$$a(u, \psi) + \langle u, \psi \rangle = \langle f, \psi \rangle.$$

Because the restrictions of functions in $C_c^{\infty}(\Omega)$ to Ω are dense in $H^1(\Omega)$ we conclude that u is the weak solution of (7.2.2) as claimed.

8. Approximation by smooth data and domains

The above can be used to approximate problems on nonsmooth domains by a sequence of problems on smooth domains. This is a useful tool to get results for nonsmooth domains, using results on smooth domains. Such techniques were for instance central in [12,54,86, 93,98]. The technique can be used to prove isoperimetric inequalities for eigenvalues, given they are known for smooth domains and involve constants independent of the geometry of the domain. A recent collection of such inequality for which the technique could be applied appears in [18]. Such an approach was also used in [60] for Robin boundary conditions.

8.1. Approximation by operators having smooth coefficients

Consider an operator \mathcal{A} as in Section 2.1 with diffusion matrix $A_0 = [a_{ij}]$, drift terms $a = (a_1, \ldots, a_N)$ and $b = (b_1, \ldots, b_N)$, and potential c_0 in $L^{\infty}(\mathbb{R}^N)$. Also assume that \mathcal{A} satisfies the ellipticity condition (2.1.3). If \mathcal{A} is only given on an open set Ω , then extend it to an operator on \mathbb{R}^N as in Remark 2.1.1.

Define the nonnegative function $\varphi \in C_c^{\infty}(\mathbb{R}^N)$ by

$$\varphi(x) := \begin{cases} c \exp\left(\frac{1}{|x|^2 - 1}\right) & \text{if } |x| < 1\\ 0 & \text{if } |x| \ge 1 \end{cases}$$

with c > 0 chosen such that

$$\int_{\mathbb{R}^N} \varphi(x) \, dx = 1$$

For all $n \in \mathbb{N}$ define φ_n by $\varphi_n(x) := n^N \varphi(nx)$. Then $(\varphi_n)_{n \in \mathbb{N}}$ is an approximate identity in \mathbb{R}^N . For all $n \in \mathbb{N}$ and i, j = 1, ..., N we set

$$a_{ij}^{(n)} := \varphi_n * a_{ij}, \qquad a_i^{(n)} := \varphi_n * a_i, \qquad b_i^{(n)} := \varphi_n * b_i \text{ and } c_{0n} := \varphi_n * c_0.$$

We then define A_n by

$$\mathcal{A}_n u := -\operatorname{div}(A_n \nabla u + a_n u) + b_n \cdot \nabla u + c_{0n} u \tag{8.1.1}$$

with $A_n = [a_{ij}^{(n)}]$, $a_n = (a_1^{(n)}, \ldots, a_N^{(n)})$, $b_n = (b_1^{(n)}, \ldots, b_N^{(n)})$. The following proposition shows that the family of operators \mathcal{A}_n in particular satisfies Assumption 5.1.1.

PROPOSITION 8.1.1. The family of operators A_n as defined above has coefficients of class C^{∞} and satisfies Assumption 5.1.1. Moreover, A_n has the same ellipticity constant α_0 as A and

$$\lambda_{\mathcal{A}_n} := \|c_{0n}^-\|_{\infty} + \frac{1}{2\alpha_0} \|a_n + b_n\|_{\infty}^2 \le \lambda_{\mathcal{A}}$$
(8.1.2)

for all $n \in \mathbb{N}$, where λ_A is defined by (2.1.10).

PROOF. Let $g \in L_{\infty}(\mathbb{R}^N)$. By the properties of convolution

$$-\|g^-\|_{\infty} = -\int_{\mathbb{R}^N} \varphi_n(y) \|g^-\|_{\infty} \, dy \le \int_{\mathbb{R}^N} \varphi_n(y) g(y-x) \, dy$$
$$= \varphi_n * g \le \int_{\mathbb{R}^N} \varphi_n(y) \|g^+\|_{\infty} \, dy = \|g^+\|_{\infty}$$

for all $n \in \mathbb{N}$ if we use that $\|\varphi_n\|_1 = 1$. Here $g^+ := \max\{u, 0\}$ and $g^- := \max\{-u, 0\}$ are the positive and negative parts of g. In particular, from the above

 $\|\varphi_n * g\|_{\infty} \le \|g\|_{\infty}$ and $\|(\varphi_n * g)^-\|_{\infty} \le \|g^-\|_{\infty}$.

Assuming that the ellipticity condition (2.1.3) holds we get

$$\xi^{T} A_{n}(x)\xi = \sum_{i,j=1}^{N} a_{ij}^{(n)}(x)\xi_{j}\xi_{i} = \sum_{i,j=1}^{N} \int_{\mathbb{R}^{N}} \varphi_{n}(y)a_{ij}(x-y) \, dy\xi_{j}\xi_{i}$$
$$= \int_{\mathbb{R}^{N}} \varphi_{n}(y) \sum_{i,j=1}^{N} a_{ij}(x-y)\xi_{j}\xi_{i} \, dy \ge \int_{\mathbb{R}^{N}} \varphi_{n}(y)\alpha_{0}|\xi|^{2} \, dy = \alpha_{0}|\xi|^{2}$$

for all $n \in \mathbb{N}$. Hence \mathcal{A}_n is elliptic with the same ellipticity constant as \mathcal{A} . Using the smoothing properties of convolution, the coefficients of \mathcal{A}_n are of class C^{∞} . Moreover, they converge to the corresponding coefficients of \mathcal{A} almost everywhere (see [68, Section 4.2.1, Theorem 1]).

8.2. Approximation by smooth domains from the interior

The purpose of this section is to prove that every open set in \mathbb{R}^N can be exhausted by a sequence of smoothly bounded open sets. This fact is frequently used, but proofs are often not given, only roughly sketched or very technical (see [63, Section II.4.2, Lemma 1] and [67, Theorem V.4.20]). We give a simple proof based on the existence of suitable cutoff functions and Sard's lemma. The idea is similar to the proof of [67, Theorem V.4.20], where the existence of a sequence of approximating domains with analytic boundary is shown. For open sets U, V we write

 $U \subset V$

if U is bounded and $\overline{U} \subset V$.

PROPOSITION 8.2.1. Let $\Omega \subset \mathbb{R}^N$ be an open set. Then there exists a sequence of bounded open sets Ω_n with boundary of class C^{∞} such that $\Omega_n \subset \Omega_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$ and $\Omega = \bigcup_{n \in \mathbb{N}} \Omega_n$. If Ω is connected we can choose Ω_n to be connected as well.

PROOF. Given an open set $\Omega \subset \mathbb{R}^N$, define

$$V_n := \{x \in \Omega : \operatorname{dist}(x, \partial \Omega) > 1/n, |x| < n\}$$

for all $n \in \mathbb{N}$. Then $U_n := V_n$ is open and $U_n \subset U_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$ and $\bigcup_{n \in \mathbb{N}} U_n = \Omega$. Next choose cutoff functions $\psi_n \in C_c^{\infty}(U_{n+1})$ such that $0 \le \psi \le 1$ with $\psi = 1$ on \overline{U}_n . By Sard's lemma (see [85, Theorem 3.1.3]) we can choose regular values $t_n \in (0, 1)$ of ψ_n for every $n \in \mathbb{N}$. Then for every $n \in \mathbb{N}$ we set

 $E_n := \{ x \in \Omega \colon \psi_n(x) > t_n \}.$

Let Ω_n consist of the connected components of E_n containing U_n . With this choice $U_n \subset \Omega_n \subset U_{n+1}$, and since t_n is a regular value of ψ_n , by the implicit function theorem, $\partial \Omega_n$ is of class C^{∞} . By the properties of U_n also $\Omega_n \subset \Omega_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$ and $\Omega = \bigcup_{n \in \mathbb{N}} \Omega_n$.

If Ω is connected we can choose a sequence U_n of connected subsets of V_n with $U_n \subset U_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$. Then the Ω_n constructed above are connected as required. We finish this proof by showing that we can indeed choose U_n connected if Ω is connected. There exists $n_0 \in \mathbb{N}$ such that $V_n \neq \emptyset$ for all $n \ge n_0$. Fix $x_0 \in V_{n_0}$ and denote by U_{n_0} the connected component of V_{n_0} containing x_0 . For $n > n_0$ we inductively define U_n to be the connected component of V_n containing U_{n-1} . Then $U := \bigcup_{n \in \mathbb{N}} U_n$ is a nonempty open set. If we show that $\Omega \setminus U$ is open, then the connectedness of Ω implies that $U = \Omega$. Let $x \in \Omega \setminus U$. Since Ω is open there exists $m \in \mathbb{N}$ such that $B(x, 2/m) \subset \Omega$. Hence $B(x, 1/m) \subset V_m$ and so $B(x, 1/m) \cap U_n = \emptyset$ for all $n \in \mathbb{N}$, since otherwise $B(x, 1/m) \subset U_n$ for some n. Therefore $B(x, 1/m) \subset \Omega \setminus U$ and thus $\Omega \setminus U$ is open. \Box

8.3. Approximation from the exterior for Lipschitz domains

Approximation from the inside by smooth domains is a useful technique for the Dirichlet problem. The situation is more difficult for Robin problems, where we saw in Section 6

that the limit problem very much depends on the boundary of the domains Ω_n . We want to state an existence theorem on a sequence of smooth domains where the boundary measure converges to the correct measure of the limit domain. The result originally goes back to Nečas [103]. We state it as proved in [66, Theorem 5.1].

THEOREM 8.3.1. Let Ω be a bounded Lipschitz domain in \mathbb{R}^N . Then there exists a sequence of domains Ω_n of class C^{∞} satisfying Assumption 6.4.1. Moreover, the functions $\varphi_n : B \to \mathbb{R}$ from Assumption 6.4.1 can be chosen such that

- (1) $\varphi_n \in C^{\infty}(B)$ for $n \ge 1$;
- (2) $\varphi_n \rightarrow \varphi_0$ uniformly on *B*;
- (3) $\|\varphi_n\|_{\infty}$ is uniformly bounded with respect to $n \in \mathbb{N}$;
- (4) $\nabla \varphi_n \to \nabla \varphi_0$ in $L_p(B)$ for all $p \in [1, \infty)$.

The situation is shown in Figure 6.5. Note that Condition (4) is the most difficult to achieve.

REMARK 8.3.2. We can apply the above theorem to get a sequence of smooth domains approaching a given Lipschitz domain by smooth domains in such a way that Theorem 6.5.1 applies with g = 1, that is, the boundary conditions on the limit domain are the same as on the domains Ω_n .

Let φ_n be as in the above theorem and set

$$g_n := \sqrt{1 + |\nabla \varphi_n|^2}.$$

Since for $0 \le b < a$ the function $t \mapsto \sqrt{t + a^2} - \sqrt{t + b^2}$ is decreasing as a function of $t \ge 0$ we get

$$\sqrt{1+a^2} - \sqrt{1+b^2} \le a-b.$$

Therefore

$$\left|\sqrt{1+|\nabla\varphi_n|^2}-\sqrt{1+|\nabla\varphi_0|^2}\right| \le \left||\nabla\varphi_n|-|\nabla\varphi_0|\right| \le |\nabla\varphi_n-\nabla\varphi_0|,$$

and so from (4) in the above theorem

$$\left\|\frac{g_n}{g_0} - 1\right\|_p \le \|g_n - g_0\|_p \le \|\nabla\varphi_n - \nabla\varphi_0\|_p \to 0$$

as $n \to \infty$ for all $p \in [1, \infty)$. Hence

$$\lim_{n \to \infty} \frac{g_n}{g_0} = 1$$

in $L_p(\Omega)$ for all $p \in [1, \infty)$ as required in Theorem 6.5.1.

9. Perturbation of semi-linear problems

9.1. Basic convergence results for semi-linear problems

Consider the boundary value problem

$$\mathcal{A}u = f(x, u(x)) \quad \text{in } \Omega, \mathcal{B}u = 0 \qquad \text{on } \partial\Omega,$$
(9.1.1)

and perturbations

$$\mathcal{A}_n u = f(x, u(x)) \quad \text{in } \Omega_n, \mathcal{B}_n u = 0 \qquad \text{on } \partial \Omega_n.$$
(9.1.2)

We assume throughout that A_n , A satisfy Assumption 5.1.1. We considered linear problems with various boundary operators \mathcal{B}_n . Let $R_n(\lambda)$ and $R(\lambda)$ be the resolvent operators corresponding to the boundary value problems (A_n, B_n) and the relevant limit problem $(\mathcal{A}, \mathcal{B})$ as given in Definition 2.5.2. Finally, let A_n , A be the operators induced by $(\mathcal{A}_n, \mathcal{B}_n)$ and $(\mathcal{A}, \mathcal{B})$ as defined in Section 2.2. In all cases we proved that the following assumptions hold.

ASSUMPTION 9.1.1. Let Ω be bounded. For every $\lambda \in \varrho(-A)$ there exists $n_0 \in \mathbb{N}$ such that $\lambda \in \varrho(-A_n)$ for all $n > n_0$. Moreover, for every $p \in (1, \infty)$

$$\lim_{n\to\infty} R_n(\lambda) = R(\lambda)$$

in $\mathcal{L}(L_p(\mathbb{R}^N), L_q(\mathbb{R}^N)) \cap \mathcal{L}(L_p(\mathbb{R}^N))$ for all $q \in [p, m(p))$. Here m(p) is given by (2.4.5) with *d* depending on to the type of problem under consideration (see Tables 2.1–2.3).

We summarise the various cases below.

- (1) Dirichlet boundary conditions on Ω_n and the limit domain Ω (see Section 5.2, in particular Theorem 5.2.6).
- (2) Robin boundary conditions on Ω_n and Ω, perturbing the original boundary ∂Ω only very slightly (see Section 6.3). The dumbbell problem shown in Figure 6.1 is an example.
- (3) Robin boundary conditions on Ω_n with very fast oscillating boundary. On the limit domain Ω we have Dirichlet boundary conditions (see Section 6.4).
- (4) Robin boundary conditions on Ω_n with moderately fast oscillating boundary. On the limit domain Ω we have Robin boundary conditions with a different boundary coefficient (see Section 6.5).
- (5) Neumann boundary conditions and an additional assumption on the domains Ω_n . In the limit we also have Neumann boundary conditions (see Section 7).

As seen in Section 3.1 we can rewrite (9.1.1) and (9.1.2) as fixed point equations in $L_p(\Omega)$ and $L_p(\Omega_n)$, respectively. We want write the problems as equations on $L_p(\mathbb{R}^N)$.

Suppose that $f : \mathbb{R}^N \times \mathbb{R}$ satisfies Assumption 3.1.1 and let *F* be the corresponding superposition operator. In the spirit of Proposition 3.1.3 we let

$$G(u) := R(\lambda)(F(u) + \lambda u)$$
 and $G_n(u) := R_n(\lambda)(F(u) + \lambda u)$

for $\lambda \geq 0$ big enough so that $R_n(\lambda)$, $R(\lambda)$ are well defined. The only difference to the definition given in Proposition 3.1.3 is, that G acts on $L_p(\mathbb{R}^N)$ rather than $L_p(\Omega)$, and similarly G_n . This allows us to work in a big space independent of Ω . By Proposition 3.1.3, finding a solution of (9.1.1) in $L_p(\Omega)$ is equivalent to finding a solution to the fixed point equation

$$u = G(u) \tag{9.1.3}$$

in $L_p(\mathbb{R}^N)$ for $p \ge 2d/(d-2)$. Similarly, (9.1.2) is equivalent to

$$u = G_n(u) \tag{9.1.4}$$

in $L_p(\mathbb{R}^N)$. We now prove some basic properties of G_n, G .

PROPOSITION 9.1.2. Suppose that $f : \mathbb{R}^N \times \mathbb{R}$ satisfies Assumption 3.1.1, and that γ , p are such that (3.1.5) holds. Let Ω be bounded and suppose that Assumption 9.1.1 is satisfied. Then

- (1) $G, G_n \in C(L_p(\mathbb{R}^N, L_p(\mathbb{R}^N)))$ for all $n \in \mathbb{N}$;
- (2) If $u_n \to u$ in $L_p(\mathbb{R}^N)$, then $G_n(u_n) \to G(u)$ in $L_p(\mathbb{R}^N)$;
- (3) For every bounded sequence (u_n) in $L_p(\mathbb{R}^N)$, the sequence $(G_n(u_n))_{n \in \mathbb{N}}$ has a convergent subsequence in $L_p(\mathbb{R}^N)$.

Moreover, the map G_n is compact if Ω_n is bounded.

PROOF. By the growth condition (3.1.5) and Assumption 9.1.1 it follows that $R_n(\lambda) \rightarrow R(\lambda)$ in $\mathcal{L}(L_p(\mathbb{R}^N), L_{p/\gamma}(\mathbb{R}^N)) \cap \mathcal{L}(L_p(\mathbb{R}^N))$. The first assertion (1) then directly follows from Proposition 3.1.3. Also the compactness of *G* and G_n if Ω and Ω_n are bounded follows from Proposition 3.1.3. Statement (2) follows from the continuity of *F* proved in Lemma 3.1.2 and the assumptions on $R_n(\lambda)$. To prove (3) note that by Proposition 3.1.3 the sequence $(F(u_n))_{n\in\mathbb{N}}$ is bounded for every bounded sequence (u_n) in $L_p(\Omega)$. Hence there exists a subsequence such that $F(u_{n_k}) + \lambda u_{n_k} \rightharpoonup g$ weakly in $L_{p/\gamma}(\mathbb{R}^N) + L_p(\mathbb{R}^N)$. By Theorem 4.3.4

$$G_{n_k}(u_{n_k}) = R_{n_k}(\lambda)(F(u_{n_k}) + \lambda u_{n_k}) \to R(\lambda)g$$

in $L_p(\mathbb{R}^N)$, completing the proof of the proposition.

REMARK 9.1.3. Sometimes, when working with positive solutions and comparison principles it is useful to make sure $f(x, \xi) + \lambda \xi$ is increasing as a function of ξ , and that $R(\lambda)$ is a positive operator. Both can be achieved by choosing $\lambda > 0$ large enough if $f \in C^1(\mathbb{R}^N \times \mathbb{R})$.

REMARK 9.1.4. For the assertions of the above propositions to be true Assumption 3.1.1 is not necessary. We discuss some examples. For instance consider a linear f of the form

$$f(x, u) = w_n u$$

with $w_n \rightarrow w$ weakly in $L_q(\mathbb{R}^N)$ for some q > d/2. Such cases arise in dealing with semi-linear problems (see for instance [50, Lemma 2]). More generally, suppose that γ , p satisfy (3.1.5) and that $q > p/(\gamma - \delta)$. If

$$f(x, u) = w_n(x)|u|^{\delta - 1}u$$

with $w_n \rightharpoonup w$ weakly in $L_q(\mathbb{R}^N)$, then the assertions of the above propositions hold in $L_p(\mathbb{R}^N)$.

THEOREM 9.1.5. Suppose that $f : \mathbb{R}^N \times \mathbb{R}$ satisfies Assumption 3.1.1, and that γ , p are such that (3.1.5) holds. Further suppose that u_n are solutions of (9.1.2) such that the sequence (u_n) is bounded in $L_p(\mathbb{R}^N)$. Finally suppose that Assumption 9.1.1 is satisfied. Then there exists a subsequence (u_{n_k}) converging to a solution u of (9.1.1) in $L_q(\mathbb{R}^N)$ for all $q \in [p, \infty)$.

PROOF. We know from the above discussion that $u_n = G_n(u_n)$ for all $n \in \mathbb{R}^N$. Since (u_n) is bounded, Proposition 9.1.2 property (3) implies that there exists a subsequence (u_{n_k}) such that

$$u_{n_k} = G_{n_k}(u_{n_k}) \to u$$

in $L_p(\mathbb{R}^N)$. Now by Proposition 9.1.2 property (2) we get that $G_{n_k}(u_{n_k}) \to G(u)$ in $L_p(\mathbb{R}^N)$. Theorem 3.2.1 implies that (u_n) is bounded in $L_\infty(\mathbb{R}^N)$. Hence by interpolation, convergence is in $L_q(\mathbb{R}^N)$ for all $q \in [p, \infty)$.

REMARK 9.1.6. (a) If we assume that $1 \le \gamma < (d+2)/(d-2)$, then a bound in $L_2(\mathbb{R}^N)$ is sufficient in the above theorem.

(b) Solutions to semi-linear problems are in general not unique, so we do not expect the whole sequence to converge. If we happen to know by some means that (u_n) has at most one limit, then the whole sequence converges.

We next look at problems *without growth conditions* on the nonlinearity, but instead assume that the solutions on Ω_n are bounded uniformly with respect to $n \in \mathbb{N}$. We still assume that $f : \mathbb{R}^N \times \mathbb{R}$ is Carathéodory (see Assumption 3.1.1 for what this means).

THEOREM 9.1.7. Suppose that $f: \mathbb{R}^N \times \mathbb{R}$ is a Carathéodory function, and that $\sup_{n \in \mathbb{N}} |\Omega_n| < \infty$. Further suppose that u_n are solutions of (9.1.2) such that the sequence (u_n) is bounded in $L_{\infty}(\mathbb{R}^N)$. Finally suppose that Assumption 9.1.1 is satisfied. Then there exists a subsequence (u_{n_k}) converging to a solution u of (9.1.1) in $L_p(\mathbb{R}^N)$ for all $p \in [1, \infty)$.

PROOF. Let $M := \sup_{n \in \mathbb{N}} ||u_n||_{\infty}$. Let $\psi \in C^{\infty}(\mathbb{R})$ be a monotone function with $\psi(\xi) = \psi$ if $\xi \leq M$ and $\psi(\xi) = M + 1$ if $|\xi| \geq M + 1$. Then define $\tilde{f}(x, \xi) := f(x, \psi(\xi))$ and by \tilde{F} the corresponding superposition operator. Then \tilde{f} is bounded and therefore satisfies

Assumption 3.1.1 with $\gamma = 1$. Since $||u_n||_{\infty} \leq M$ we clearly have $\tilde{F}(u_n) = F(u_n)$ and so we can replace F by \tilde{F} in the definition of G_n , G without changing the solutions. Hence we can apply Theorem 9.1.5 to conclude that $u_n \to u$ in $L_2(\mathbb{R}^N)$. By the uniform L_{∞} -bound convergence is in $L_p(\mathbb{R}^N)$ for all $p \in [2, \infty)$, and by the uniform boundedness of $|\Omega_n|$ also for $p \in [1, 2)$.

REMARK 9.1.8. An L_{∞} -bound as required above follows from an L_p -bound under suitable growth conditions on the nonlinearity f as shown in Section 3.2.

9.2. Existence of nearby solutions for semi-linear problems

Suppose that the limit problem (9.1.1) has a solution. In this section we want to prove that under certain circumstances the perturbed problem (9.1.2) has a solution nearby. In the abstract framework this translates into the question whether the fixed point equation (9.1.4) has, at least for *n* large enough, a solution near a given solution of (9.1.3). Of course, we do not expect this for arbitrary solutions. Note that the results here do not just apply to domain perturbation problems, but to other types of perturbations having similar properties as well.

One common technique to prove existence of such solutions is by means of the *Leray–Schauder degree* (see [65, Chapter 2.8] or [99, Chapter 4]). We assume that G is a compact operator, that is, if it maps bounded sets onto relatively compact sets. Then, if $U \subset E$ is an open bounded set such that $u \neq G_{\Omega}(u)$ for all $u \in \partial U$, then the Leray–Schauder degree, deg $(I - G_{\Omega}, U, 0) \in \mathbb{Z}$, is well defined. If we deal with positive solutions we can use the degree in cones as in [3,44]. In order to do that we need some more assumptions. These are satisfied for the concrete case of semi-linear boundary value problems as shown in Proposition 9.1.2 with $E = L_p(\mathbb{R}^N)$ for some $p \in (1, \infty)$.

ASSUMPTION 9.2.1. Suppose E is a Banach space and suppose

- (1) $G, G_n \in C(E, E)$ are compact for all $n \in \mathbb{N}$;
- (2) If $u_n \to u$ in *E*, then $G_n(u_n) \to G(u)$ in *E*.
- (3) For every bounded sequence (u_n) in E the sequence $(G_n(u_n))_{n \in \mathbb{N}}$ has a convergent subsequence in E.

The following is the main result of this section. The basic idea of the proof for specific domain perturbation problems goes back to [45]. The proof given here is a more abstract version of the ones found in [52,8] for some specific domain perturbation problems.

THEOREM 9.2.2. Suppose that G_n , G satisfy Assumption 9.2.1. Moreover, let $U \subset E$ be an open bounded set such that $G(u) \neq u$ for all $u \in \partial U$. Then there exists $n_0 \in \mathbb{N}$ such that $G_n(u) \neq u$ for all $u \in \partial U$ and

$$\deg(I - G, U, 0) = \deg(I - G_n, U, 0)$$
(9.2.1)

for all $n \ge n_0$.

PROOF. We use the homotopy invariance of the degree (see [65, Section 2.8.3] or [99, Theorem 4.3.4]) to prove (9.2.1). We define the homotopies $H_n(t, u) := tG_n(u) + (1-t)G(u)$

for $t \in [0, 1]$, $u \in E$ and $n \in \mathbb{N}$. To prove (9.2.1) it is sufficient to show that there exists $n_0 \in \mathbb{N}$ such that

$$u \neq H_n(t, u) \tag{9.2.2}$$

for all $n \ge n_0$, $t \in [0, 1]$ and $u \in \partial U$. Assume to the contrary that there exist $n_k \to \infty$, $t_{n_k} \in [0, 1]$ and $u_{n_k} \in \partial U$ such that $u_{n_k} = H_{n_k}(t_{n_k}, u_{n_k})$ for all $k \in \mathbb{N}$. As U is bounded in E, Assumption 9.2.1 (1) and (3) guarantee that $t_{n_k} \to t_0$ in [0, 1], $G_{n_k}(u_{n_k}) \to v$ and $G(u_{n_k}) \to w$ in E for some $v, w \in E$ if we pass to a further subsequence. Hence

$$u_{n_k} = H_{n_k}(t_{n_k}, u_{n_k}) = t_{n_k} G_{n_k}(u_{n_k}) + (1 - t_{n_k}) G(u_{n_k})$$
$$\xrightarrow{k \to \infty} u := t_0 v + (1 - t_0) w$$

in *E* and so $u_{n_k} \to u$ in *E* and $u \in \partial U$ since ∂U is closed. Now the continuity of *G* and Assumption 9.2.1 part (2) imply that v = w = G(u), so that $u_{n_k} \to u = t_0 G(u) + (1 - t_0)G(u) = G(u)$. Hence u = G(u) for some $u \in \partial U$, contradicting our assumptions. Thus there exists $n_0 \in \mathbb{N}$ such that (9.2.2) is true for all $n \ge n_0$, completing the proof of the theorem.

Of course, we are most interested in the case deg $(I - G_{\Omega}, U, 0) \neq 0$. Then, by the solution property of the degree (see [99, Theorem 4.3.2]), (9.1.3) has a solution in U. As a corollary to Theorem 9.2.2 we get the existence of a solution of (9.1.4) in U.

COROLLARY 9.2.3. Suppose that G_n , G satisfy Assumption 9.2.1 and that $U \subset E$ is open and bounded with $u \neq G(u)$ for all $u \in \partial U$. If deg $(I - G, U, 0) \neq 0$, then there exists $n_0 \in \mathbb{N}$ such that (9.1.4) has a solution in U for all $n \ge n_0$.

Now we consider an isolated solution u_0 of (9.1.3) and recall the definition of its *index*. Denote by $B_{\varepsilon}(u_0)$ the open ball of radius $\varepsilon > 0$ and centre u_0 in E. Then $\deg(I - G, B_{\varepsilon}(u_0), 0)$ is defined for small enough $\varepsilon > 0$. Moreover, by the excision property of the degree $\deg(I - G, B_{\varepsilon}(u_0), 0)$ stays constant for small enough $\varepsilon > 0$. Hence the fixed point index of u_0 ,

$$\mathbf{i}(G, u_0) := \lim_{\varepsilon \to 0} \deg(I - G, B_{\varepsilon}(u_0), 0)$$

is well defined.

THEOREM 9.2.4. Suppose that G_n , G satisfy Assumption 9.2.1. If u_0 is an isolated solution of (9.1.3) with $i(G, u_0) \neq 0$, then for n large enough there exist solutions u_n of (9.1.4) such that $u_n \rightarrow u_0$ in E as $n \rightarrow \infty$.

PROOF. By assumption there exists $\varepsilon_0 > 0$ such that

$$i(G, u_0) = \deg(I - G, B_{\varepsilon}(u_0), 0) \neq 0$$

for all $\varepsilon \in (0, \varepsilon_0)$. Hence by Corollary 9.2.3 problem (9.1.4) has a solution in $B_{\varepsilon}(u_0)$ for all $\varepsilon \in (0, \varepsilon_0)$ if only *n* large enough. Hence a sequence as required exists.

Without additional assumptions it is possible that there are several different sequences of solutions of (9.1.4) converging to u_0 . However, if $G \in C^1(E, E)$ and u_0 is *nondegenerate*, that is, the linearised problem

$$v = DG(u_0)v \tag{9.2.3}$$

has only the trivial solution, then u_n is unique for large $n \in \mathbb{N}$.

THEOREM 9.2.5. Suppose that $G_n, G \in C^1(E, E)$ satisfy Assumption 9.2.1. Further assume that $DG_n(u_n) \to DG(u)$ in $\mathcal{L}(E)$. If u_0 is a nondegenerate solution of (9.1.3), then there exists $\varepsilon > 0$ such that (9.1.4) has a unique solution in $B_{\varepsilon}(u_0)$ for all n large enough and this solution is nondegenerate.

PROOF. As u_0 is nondegenerate $I - DG(u_0)$ is invertible with bounded inverse. Moreover, since *G* is compact [65, Proposition 8.2] implies that $DG(u_0)$ is compact as well. By [99, Theorem 5.2.3 and Theorem 4.3.14] $i(G, u_0) = \pm 1$, so by Theorem 9.2.4 there exists a sequence of solutions u_n of (9.1.3) with $u_n \rightarrow u$ as $n \rightarrow \infty$. As the set of invertible linear operators is open in $\mathcal{L}(E)$ we conclude that $I - DG_n(u_n)$ is invertible for *n* sufficiently large. Hence u_n is nondegenerate for *n* sufficiently large.

We now show uniqueness. Suppose to the contrary that there exist solutions u_n and v_n of (9.1.4) converging to u_0 with $u_n \neq v_n$ for all $n \in \mathbb{N}$ large enough. As $G_n \in C^1(E, E)$ we get

$$G(u_n) - G(v_n) = \int_0^1 DG_n(tu_n + (1-t)v_n) dt(u_n - v_n).$$

Hence if we set $w_n := \frac{u_n - v_n}{\|u_n - v_n\|}$, then $\|w_n\| = 1$ and

$$w_n = \frac{G_n(u_n) - G_n(v_n)}{\|u_n - v_n\|} = \int_0^1 DG_n(tu_n + (1 - t)v_n) dt w_n$$
(9.2.4)

for all $n \in \mathbb{N}$. As $u_n, v_n \to u_0$ we get $tu_n + (1-t)v_n \to u_0$ in *E* for all $t \in [0, 1]$ and hence by assumption $DG_n(tu_n + (1-t)v_n) \to DG(u_0)$ in $\mathcal{L}(E)$ for all $t \in [0, 1]$. By the continuity of DG_n and the compactness of [0, 1] there exists $t_n \in [0, 1]$ such that

$$\sup_{t\in[0,t]} \|DG_n(tu_n + (1-t)v_n)\| = \|DG_n(t_nu_n + (1-t_n)v_n)\|.$$

Since $t_n u_n + (1 - t_n)v_n \rightarrow u_0$ by assumption $DG_n(tu_n + (1 - t)v_n) \rightarrow DG(u_0)$ in $\mathcal{L}(E)$ and hence

$$\sup_{\substack{t\in[0,t]\\n\in\mathbb{N}}} \left\| DG_n(tu_n + (1-t)v_n) \right\| < \infty.$$

By the dominated convergence theorem

$$\int_0^1 DG_n(tu_n + (1-t)v_n) \, dt \to DG(u_0)$$

in $\mathcal{L}(E)$. As the set of invertible operators is open we see that

$$I - \int_0^1 DG_n(tu_n + (1 - t)v_n) \, dt$$

has a bounded inverse for *n* sufficiently large. But this contradicts (9.2.4) and therefore uniqueness follows. \Box

9.3. Applications to boundary value problems

We now discuss how to apply the abstract results in the previous section to boundary value problems. In addition to the assumptions made on f in Section 9.1 we assume that the corresponding superposition operator is differentiable. Conditions for that can be found in [6, Theorem 3.12]. We also assume that there exists a bounded set B such that

$$\Omega_n, \Omega \subset B$$

for all $n \in \mathbb{N}$.

THEOREM 9.3.1. Suppose that $f \in C(\mathbb{R}^N \times \mathbb{R})$ satisfies Assumption 3.1.1, and that γ , p are such that (3.1.5) holds. Further suppose that Assumption 9.1.1 is satisfied, and that $u \in L_p(\mathbb{R}^N)$ is a nondegenerate isolated solution of (9.1.1). Then, there exists $\varepsilon > 0$ such that, for n large enough, equation (9.1.2) has a unique solution $u_n \in L_p(\Omega)$ with $||u_n - u||_p < \varepsilon$. Moreover, $u_n \to u \in L_q(\mathbb{R}^N)$ for all $q \in (1, \infty)$.

PROOF. By Proposition 9.1.2 all assumptions of Theorem 9.2.5 are satisfied if we choose $E = L_p(\mathbb{R}^N)$. Hence there exist $\varepsilon > 0$ and $u_n \in L_p(\mathbb{R}^N)$ as claimed. By Theorem 3.2.1 the sequence (u_n) is bounded in $L_{\infty}(\mathbb{R}^N)$ and therefore the convergence is in $L_q(\mathbb{R}^N)$ for all $q \in [p, \infty)$. Since the measure of Ω_n is uniformly bounded, convergence takes place in $L_q(\mathbb{R}^N)$ for $q \in [1, p)$ as well.

REMARK 9.3.2. (a) We could for instance choose $f(u) := |u|^{\gamma-1}u$, or a nonlinearity with that growth behaviour. Then we can choose p big enough such that (3.1.5) is satisfied. If $\gamma < (d+2)/(d-2)$, then the theorem applies to all nondegenerate weak solutions.

(b) The above theorem does not necessarily imply that (9.1.2) has only one solution near the solution u of (9.1.1) in $L_2(\mathbb{R}^N)$, because a solution in $L_2(\mathbb{R}^N)$ does not need to be close to u in $L_p(\mathbb{R}^N)$. However, if $\gamma < (d+2)/(d-2)$, then u_n is the unique weak solution of (9.1.2) in the ε -neighbourhood of u in $L_2(\mathbb{R}^N)$.

We next want to look at a problem without any growth conditions on f. In such a case we have to deal with solutions in $L_{\infty}(\mathbb{R}^N)$ only. The idea is to truncate the nonlinearity and apply Theorem 9.3.1.

THEOREM 9.3.3. Suppose that $f \in C(\mathbb{R}^N \times \mathbb{R})$ satisfies Assumption 3.1.1, and that γ , p are such that (3.1.5) holds. Further suppose that Assumption 9.1.1 is satisfied, and that $u \in L_{\infty}(\mathbb{R}^N)$ is a nondegenerate isolated solution of (9.1.1). Then, there exists $\varepsilon > 0$ such that, for n large enough, equation (9.1.2) has a unique solution $u_n \in L_{\infty}(\Omega)$ with $\sup_{n \in \mathbb{N}} ||u_n||_{\infty} < \infty$ and $u_n \to u \in L_q(\mathbb{R}^N)$ for all $q \in [1, \infty)$.

PROOF. Fix p > d/2 with *d* from Assumption 9.1.1. Because $u \in L_{\infty}(\mathbb{R}^N)$ and $f \in C(\overline{B} \times \mathbb{R})$ we have $F(u) \in L_p(\mathbb{R}^N)$, where *F* is the superposition operator induced by *f*. By Theorem 2.4.1 and the assumptions on $(\mathcal{A}_n, \mathcal{B}_n)$ there exists a constant *C* independent of $n \in \mathbb{N}$ such that

$$||u||_{\infty} \leq M := C(||F(u)||_{p} + \lambda_{0}||u||_{p}) + ||u||_{p}.$$

Let $\psi \in C^{\infty}(\mathbb{R})$ be a monotone function with

$$\psi(\xi) = \begin{cases} \xi & \text{if } \xi \le M+1, \\ \operatorname{sgn}\xi(M+2) & \text{if } |\xi| \ge M+2. \end{cases}$$

Then define $\tilde{f}(x,\xi) := f(x,\psi(\xi))$. Then \tilde{f} is a bounded function on $B \times \mathbb{R}$ and $\tilde{f}(x,\xi) = f(x,\xi)$ whenever $|\xi| \le M + 1$. As \tilde{f} is bounded we can apply Theorem 9.3.1. Hence there exists $\varepsilon > 0$ such that, for *n* large enough, equation (9.1.2) has a unique solution $u_n \in L_p(\Omega)$ with $||u_n - u||_p < \varepsilon$. Moreover, $u_n \to u \in L_p(\mathbb{R}^N)$. By what we have seen above

$$||u_n||_{\infty} \le C(||F(u_n)||_p + \lambda_0 ||u_n||_p) + ||u_n||_p$$

for all $n \in \mathbb{N}$. Because \tilde{f} is bounded, Lemma 3.1.2 shows that $\tilde{F}(u_n) \to \tilde{F}(u)$ in $L_p(\mathbb{R}^N)$. Hence, for *n* large enough

$$||u_n||_{\infty} < M + 1.$$

By definition of \tilde{F} it follows that $\tilde{F}(u_n) = F(u_n)$ for *n* large enough, so u_n is a solution of (9.1.2). Convergence in $L_q(\mathbb{R}^N)$ for $q \in [p, \infty)$ follows by interpolation and for $q \in [1, p)$ since *B* has bounded measure.

The main difficulty in applying the above results is to show that the solution of (9.1.1) is nondegenerate. A typical example is to look at a problem on two or more disjoint balls and get multiple solutions by taking different combinations of solutions. For instance if we have a trivial and a nontrivial solution, then on two disjoint balls we get four solutions: the trivial solution on both, the trivial on one and the nontrivial on the other and vice versa, and the nontrivial solution on both. We can then connect the domains either as a dumbbell domain with a small strip (see Figure 6.1) or touching balls (see Figure 7.1). The dumbbell example works for Dirichlet and Robin boundary conditions, but not for Neumann boundary conditions. For Neumann boundary conditions, the example of the touching spheres applies.

To illustrate the above we give an example to the Gelfand equation from combustion theory (see [72, §15]) due to [45]. The example shows that a simple equation can have multiple solutions on simply connected domains.

EXAMPLE 9.3.4. Consider the Gelfand equation

$$-\Delta u = \mu e^{u} \quad \text{in } \Omega,$$

$$u = 0 \qquad \text{on } \partial \Omega$$
(9.3.1)

on a bounded domain of class C^2 . If $\mu > 0$, then $\mu e^u > 0$ for all u and thus by the maximum principle every solution of (9.3.1) is positive. By [73, Theorem 1] positivity implies that all solutions are radially symmetric if Ω is a ball.

It is well known that there exists $\mu_0 > 0$ such that (9.3.1) has a minimal positive solution for $\mu \in [0, \mu_0]$ and no solution for $\mu > \mu_0$ (see [4,41]). Moreover, for $\mu \in (0, \mu_0)$ this minimal solution is nondegenerate (see [41, Lemma 3]). Let now $\Omega = B_0 \cup B_1$ be the union of two balls B_0 and B_1 of the same radius and Ω_n the dumbbell-like domains as shown in Figure 6.1. If N = 2 and $\mu \in (0, \mu_0)$, then there exists a second solution for the problems on B_0 and B_1 . In fact, the two solutions are the only solutions on a ball if N = 2(see [91, p. 242] or [72, §15, p. 359]). Note that this is not true for $N \ge 3$ as shown in [91]. Equation (8) on page 415 together with the results in Section 2 in [43] imply that there is bifurcation from every degenerate solution. Since we know that there is no bifurcation in the interval $(0, \mu_0)$, it follows that the second solution is also nondegenerate.

We now show that there are possibly more than two solutions on (simply connected) domains other than balls. Looking at $\Omega = B_0 \cup B_1$ we have four nontrivial nondegenerate solutions of (9.3.1). Hence by Theorem 9.3.3 there exist at least four nondegenerate solutions of (9.3.1) on Ω_n for *n* large. Note that similar arguments apply to the nonlinearity $|u|^{p-1}u$ for *p* subcritical as discussed in [45].

9.4. Remarks on large solutions

If all solutions of (9.1.1) are nondegenerate it is tempting to believe that the number of solutions of (9.1.2) is the same for *n* sufficiently large. However, this is not always true, and Theorem 9.3.1 only gives a lower bound for the number of solutions of the perturbed problem. If there are more solutions on Ω_n , then Theorem 9.1.5 implies that their L_{∞} -norm goes to infinity as otherwise they converge to one of the solutions on Ω and hence are unique. To prove precise multiplicity of solutions of the perturbed problem, the task is to find a universal bound on the L_{∞} norm valid for all solutions to the nonlinear problem. This is quite different from the result in Theorem 3.2.1 which just shows that under suitable growth conditions on the nonlinearity, a bound in $L_p(\mathbb{R}^N)$ implies a bound in $L_{\infty}(\mathbb{R}^N)$. However, it is still unclear whether in general there is a uniform bound on the L_p -norm for some *p*. Such uniform bounds are very difficult to get in general. We prove their existence for solutions to

$$-\Delta u = f(u(x)) \quad \text{in } \Omega,$$

$$u = 0 \qquad \text{on } \partial\Omega,$$
(9.4.1)

provided f is sublinear in the sense that

$$\lim_{|\xi| \to \infty} \frac{|f(\xi)|}{|\xi|} = 0.$$
(9.4.2)

The above class clearly includes all bounded nonlinearities.

PROPOSITION 9.4.1. Suppose that $f \in C(\mathbb{R})$ satisfies (9.4.2). Then there exists a constant M depending only on the function f and the diameter of Ω such that $||u||_{\infty} < M$ for every weak solution of (9.4.1).

PROOF. By (9.4.2), for every $\varepsilon > 0$ there exists $\alpha > 0$ such that

$$\frac{|f(\xi)|}{|\xi|} < \varepsilon$$

whenever $|\xi| > \alpha$. Setting $m_{\varepsilon} := \sup_{|\xi| < \alpha} |f(\xi)|$ we get

$$|f(\xi)| \le m_{\varepsilon} + \varepsilon |\xi|$$

for all $\xi \in \mathbb{R}$. In particular f satisfies (3.1.4) with $\gamma = 1$, and so by Theorem 3.2.1 it is sufficient to prove a uniform L_2 -bound for all weak solutions of (9.4.1). If u is a weak solution of (9.4.1), then from the above

$$\begin{aligned} \|\nabla u\|_{2}^{2} &= \int_{\Omega} f(u)u \, dx \leq \int_{\Omega} |f(u)||u| \, dx \leq \int_{\Omega} (m_{\varepsilon} + \varepsilon |u|)|u| \, dx \\ &\leq \int_{\Omega} m_{\varepsilon} |u| \, dx + \varepsilon \|u\|_{2}^{2} \leq (m_{\varepsilon} |\Omega|)^{1/2} \|u\|_{2} + \varepsilon \|u\|_{2}^{2}. \end{aligned}$$

Using an elementary inequality we conclude that

 $\|\nabla u\|_2^2 \le \varepsilon^{-1} m_\varepsilon |\Omega| + 2\varepsilon \|u\|_2^2.$

If D denotes the diameter of Ω , then from Friedrich's inequality (2.1.7)

$$\|u\|_2^2 \le D^2 \|\nabla u\|_2^2 \le \varepsilon^{-1} D^2 m_\varepsilon |\Omega| + 2D^2 \varepsilon \|u\|_2^2.$$

We next choose $\varepsilon := 1/4D^2$ and so we get

$$\|u\|_2^2 \le 8D^4 m_\varepsilon |\Omega|$$

for every weak solution of (9.4.1). The right-hand side of the above inequality only depends on the quantities listed in the proposition, and hence the proof is complete.

A priori estimates similar to the above can be obtained also for superlinear problems, but they involve the shape of the domain. An extensive discussion of the phenomena can be found in [45, Section 5] as well as [46,48,35]. The techniques to prove uniform a priori bounds are derived from Gidas and Spruck [75,74].

We complete this section by showing a simple example where large solutions occur. More sophisticated examples are in [46,48,35], in particular the example of the dumbbell as in Figure 6.1, where large solutions occur for the equation

$$-\Delta u = |u|^{p-1} u \quad \text{in } \Omega,$$

$$u = 0 \qquad \text{on } \partial\Omega,$$
(9.4.3)

for 1 . We give a simple example for the above equation, where large solutions occur, similar to examples given in [45, Section 5].



Fig. 9.1. Attaching a small shrinking ball.

EXAMPLE 9.4.2. By [114, Section I.2] the above equation has a positive solution u_r of $\Omega = B_r$ which is a ball of radius r. By a simple rescaling, it turns out that $||u_r||_{\infty} \to \infty$ as $r \to 0$. That solution is unique and nondegenerate by [45, Theorem 5], if N = 2. Then consider a domain Ω_n constructed from two touching balls $B_1 \cup B_{1/n}$ with a small connection as shown in Figure 9.1. When we let $n \to \infty$, then from Theorem 5.4.5 we get that $\Omega_n \to \Omega := B_1$ in the sense of Mosco. Since on every ball there are two solutions, the trivial solution and a nontrivial solution, there are four solutions on the union $B_1 \cup B_{1/n}$. If we make a small enough connection between the balls, then by Theorem 9.3.1 there are still at least four solutions. Hence we can construct a sequence of domains Ω_n with the required property. However, we know that there are precisely two solutions on Ω , the trivial and the nontrivial solution. The solutions on Ω_n not converging to one of the two solutions on Ω are such that $||u_n||_{\infty} \to \infty$.

9.5. Solutions by domain approximation

Consider now the nonlinear Dirichlet problem

$$\mathcal{A}u = f(u) \quad \text{in } \Omega$$

$$u = 0 \qquad \text{on } \partial \Omega$$
 (9.5.1)

on some domain Ω , bounded or unbounded. We assume that $f \in C^1(\mathbb{R})$. We then let Ω_n be open sets such that $\Omega_n \to \Omega$ and consider the sequence of problems

$$\mathcal{A}_n u = f(u) \quad \text{in } \Omega_n,$$

$$u = 0 \qquad \text{on } \partial \Omega_n.$$
(9.5.2)

The difference to the results in Section 9.3 is that we allow unbounded domains Ω . One possibility is to use the technique to prove results on nonsmooth or unbounded domains by approximation by smooth bounded domains from inside applying the results from Section 8.

THEOREM 9.5.1. Suppose that $f \in C^1(\mathbb{R})$ and that $\Omega_n \to \Omega$ in the sense of Mosco. Suppose that u_n are weak solutions of (9.5.2) and that the sequence (u_n) is bounded in $H^1(\mathbb{R}^N) \cap L_{\infty}(\mathbb{R}^N)$. Then there exists a subsequence (u_{n_k}) converging to a solution u of (9.5.1) weakly in $H^1(\mathbb{R}^N)$ and strongly in $L_{p,\text{loc}}(\mathbb{R}^N)$ for all $p \in [2, \infty)$.

PROOF. By the boundedness of (u_n) in $H^1(\mathbb{R}^N)$ there exists a subsequence (u_{n_k}) converging to $u \in H^1(\mathbb{R}^N)$ weakly. Since $H^1(\mathbb{R}^N)$ is compactly embedded into
$L_{2,\text{loc}}(\mathbb{R}^N)$ we have $u_{n_k} \to u$ in $L_{2,\text{loc}}(\mathbb{R}^N)$ as $k \to \infty$. Because (u_n) is bounded in $L_{\infty}(\mathbb{R}^N)$, convergence takes place in $L_{p,\text{loc}}(\mathbb{R}^N)$ for all $p \in [2, \infty)$. Since we have a uniform bound on (u_n) which is bounded in $L_{\infty}(\mathbb{R}^N)$ we can also truncate the nonlinearity f as in the proof of Theorem 9.3.3 and assume it is bounded. Then from Lemma 3.1.2 we get that $f(u_{n_k}) \to f(u)$ in $L_{2,\text{loc}}(\mathbb{R}^N)$.

Now fix $\varphi \in C_c^{\infty}(\Omega)$. Then there exists a ball *B* with supp $\varphi \subset B$. As $\Omega_n \to \Omega$, using Proposition 5.3.3, there exists $\varphi_n \in C_c^{\infty}(\Omega_n \cap B)$ such that $\varphi_n \to \varphi$ in $H^1(\mathbb{R}^N)$. Hence by using Proposition 5.1.2 and the fact that the support of φ_n is in *B* for all $n \in \mathbb{N}$ we get

$$\lim_{k \to \infty} \langle f(u_{n_k}), \varphi_n \rangle = \langle f(u), \varphi \rangle.$$

Since $u_{n_k} \rightharpoonup u$ weakly in $H^1(\mathbb{R}^N)$ and $\varphi_n \rightarrow \varphi$ in $H^1(B)$ we have

$$\lim_{k\to\infty}a_{n_k}(u_{n_k},\varphi_n)=a(u,\varphi)$$

Finally, by assumption $a(u_{n_k}, \varphi_{n_k}) = \langle f(u_{n_k}), \varphi_{n_k} \rangle$ for all $k \in \mathbb{N}$ we get $a(u, \varphi) = \langle f(u), \varphi \rangle$ for all $\varphi \in C_c^{\infty}(\Omega)$, that is, u is a weak solution of (9.5.1).

Note that in general we cannot expect the solution u whose existence the above theorem proves to be nonzero if Ω_n or Ω are unbounded, or more precisely if the resolvents of the linear problems do not converge in the operator norm.

REMARK 9.5.2. Note that under suitable growth conditions, a uniform L_p -bound or even an L_2 -bound on the solutions of (9.5.2) implies a uniform L_{∞} -bound by Theorem 3.2.1.

9.6. Problems on unbounded domains

Suppose that $\Omega \subset \mathbb{R}^N$ is an unbounded domain. By approximation by a sequence of bounded domains Ω_n we want to construct a nonnegative weak solution of

$$-\Delta u = |u|^{p-2} u \quad \text{in } \Omega$$

$$u = 0 \qquad \text{on } \partial \Omega$$
 (9.6.1)

with subcritical growth $p \in (2, 2N/(N-2))$. We also assume that the spectral bound of the Dirichlet Laplacian is positive. It is given as the infimum of the Rayleigh coefficient

$$s = \inf_{u \in H_0^1(\Omega)} \frac{\|\nabla u\|_2^2}{\|u\|_2^2} > 0.$$
(9.6.2)

Examples of such a domain are infinite strips of the form $\mathbb{R} \times U$ with U a domain in \mathbb{R}^{N-1} .

To construct a solution we let Ω_n be bounded open sets such that $\Omega_n \subset \Omega_{n+1} \subset \Omega$ for all $n \in \mathbb{N}$ and $\bigcup_{n \in \mathbb{N}} \Omega_n = \Omega$. We then show that a subsequence of the positive solutions of

$$-\Delta u = |u|^{p-2}u \quad \text{in } \Omega_n$$

$$u = 0 \qquad \text{on } \partial \Omega_n$$
(9.6.3)

converges to a solution of (9.6.1) as $n \to \infty$. The solution of (9.6.3) can be obtained by means of a constrained variational problem. We let

$$M_n := \{ u \in H_0^1(\Omega_n) : u > 0 \text{ and } ||u||_p^p = p \}$$

and the functional

$$J(u) := \frac{1}{2} \int_{\Omega_n} |\nabla u|^2 \, dx.$$

Then J has a minimiser $v_n \in M_n$ for every $n \in \mathbb{N}$, and that minimiser is a positive weak solution of

$$-\Delta v = \mu_n |v|^{p-2} v \quad \text{in } \Omega_n$$
$$v = 0 \qquad \text{on } \partial \Omega_n$$

with

$$\mu_n = 2J(v_n) = \|\nabla v_n\|_2^2.$$

The function

$$u_n := \mu_n^{1/(p-1)} v_n$$

then solves (9.6.3). For a proof of these facts we refer to [114, Section I.2]. We also get a bound on the solutions u_n , namely

$$\|\nabla u_n\|_2^2 = \mu_n^{2/(p-1)} \|\nabla v_n\|_2^2 = \|\nabla v_n\|_2^{4/(p-1)} \|\nabla v_n\|_2^2$$

= $\|\nabla v_n\|^{2(p+1)/(p-1)} = (2J(v_n))^{(p+1)/(p-1)}$ (9.6.4)

since $M_n \subset M_{n+1}$ for all $n \in \mathbb{N}$ we also have $J(v_{n+1}) \leq J(v_n)$. The sequences (∇v_n) and therefore (∇u_n) are bounded in $L_2(\Omega, \mathbb{R}^N)$. By (9.6.2) it follows that (u_n) and (v_n) are bounded sequences in $H_0^1(\Omega)$.

Note that the nonlinearity $|u|^{p-2}u$ satisfies Assumption 3.1.1 with g = 0 and $\gamma = p - 1 < (N + 2)/(N - 2)$. Since we can choose λ_0 (see Table 2.1), Proposition 3.2.1 shows that (u_n) is a bounded sequence in $L_{\infty}(\Omega)$. Hence there exists S > 0 such that $||u||_{\infty}^{p-2} \le S$ for all $n \in \mathbb{N}$. By (9.6.2) and since u_n is a weak solution of (9.6.3)

$$s \|\nabla u_n\|_2^2 = s \|u_n\|_p^p \le s \|u_n\|_{\infty}^{p-2} \|u_n\|_2^2 \le \|u_n\|_{\infty}^{p-2} \|\nabla u_n\|_2^2.$$

Hence $0 < s \le ||u_n||_{\infty}^{p-2}$ for all $n \in \mathbb{N}$. We have thus proved the following proposition.

PROPOSITION 9.6.1. Let s, S be as above. Then for every $n \in \mathbb{N}$ the problem (9.6.3) has a positive solution u_n such that (u_n) is bounded in $H_0^1(\Omega) \cap L_\infty(\Omega)$. Moreover, $\|\nabla u_n\|_2$ is decreasing and $0 < s \le \|u_n\|_{\infty}^{p-2} \le S$ for all $n \in \mathbb{N}$.

The next task is to show that u_n converges to a solution of (9.6.1). We have seen that (u_n) is bounded in $H_0^1(\Omega) \cap L_\infty(\Omega)$. It therefore follows from Theorem 9.5.1 that a subsequence of (u_n) converges to a solution of (9.6.1).

PROPOSITION 9.6.2. Let u_n be the solution of (9.6.3) as constructed above. Then there exists a subsequence of (u_{n_k}) converging weakly in $H^1(\Omega)$ to a solution u of (9.6.1).

The main question is whether the solution u is nonzero. If the embedding $H_0^1(\Omega) \hookrightarrow L_2(\Omega)$ is compact this is easy to see. If the embedding $H_0^1(\Omega) \hookrightarrow L_2(\Omega)$ is not compact, then we cannot expect u to be nonzero in general. However, if we assume that Ω is symmetric like for instance an infinite strip with cross-section of finite measure, we will prove the existence of a nonzero solution of (9.6.1) by exploiting the symmetry to generate compactness of a minimising sequence. We look at domains of the form

$$\Omega = \mathbb{R} \times U,$$

where $U \subset \mathbb{R}^{N-1}$ is a bounded open set. If we let $\Omega_n = (-n, n) \times U$, then by a result of Gidas–Ni–Nirenberg (see [70, Theorem 3.3]), the solutions u_n of (9.6.3) is symmetric with respect to the plane $\{0\} \times \mathbb{R}^N$ and decreasing as $|x_1|$ increases. By [97, Théorème III.2] the sequence (u_n) is compact in $L_q(\Omega)$ for 2 < q < 2N/(N-2). In the case of N = 2 a similar result appears in [115, Section 4]. It therefore follows that there exists a subsequence (u_{n_k}) converging strongly in $L_p(\Omega)$. Using u_{n_k} as a test function we get

$$\lim_{k \to \infty} \|\nabla u_{n_k}\|_2^2 = \lim_{k \to \infty} \|u_{n_k}\|_p^p = \|u\|_p^p = \|\nabla u\|_2^2.$$

As $u_{n_k} \rightarrow u$ weakly in $H_0^1(\Omega)$ it follows that $u_{n_k} \rightarrow u$ strongly in $H_0^1(\Omega)$. By Proposition 9.6.1 the sequence (u_n) is bounded in $L_{\infty}(\Omega)$ and so (u_n) is bounded in $C^{\mu}(\Omega)$ (see [76, Theorem 8.22]) as well. Hence, by Arzela–Ascoli's theorem the subsequence also converges locally uniformly on Ω . The symmetry guarantees that the maximum of (u_n) is in $U \times \{0\}$ and therefore the lower bound on $||u_n||_{\infty}^{p-2}$ from Proposition 9.6.1 implies that $u \neq 0$.

However, without the symmetry, the solution may converge to zero. As an example look at the semi-strip $\Omega = (0, \infty) \times U$ which we exhaust by the domains $\Omega_n = (0, 2n) \times U$. Then the above proposition applies. The solutions on Ω_n are just translated solutions on $(-n, n) \times U$. However, the maximum of the function u_n is in $\{n\} \times U$, and moves to infinity as $n \to \infty$. Because the solutions decrease away from the maximum, they converge to zero in $L_{2,\text{loc}}((0, \infty) \times U)$. This shows that the symmetry was essential for concluding that the limit solution is nonzero. The lower bound on $||u_n||_{\infty}^{p-2}$ from Proposition 9.6.1 and local uniform convergence do not help to get a nonzero solution.

Similarly we could look at (9.6.1) on the whole space $\Omega = \mathbb{R}^N$. We can then write Ω as a union of concentric balls Ω_n , and try to use the spherical symmetry to get some compactness from [97] similarly as above. However, since by [75] the equation (9.6.1) has no positive solution on \mathbb{R}^N , such an attempt must fail.

References

- D.R. Adams and L. Hedberg, *Function Spaces and Potential Theory*, Grundlehren der Mathematischen Wissenschaften, vol. 314, Springer-Verlag, Berlin (1996).
- [2] R.A. Adams, *Sobolev Spaces*, Pure and Applied Mathematics, vol. 65, Academic Press, New York (1975).
- [3] H. Amann, Fixed point equations and nonlinear eigenvalue problems in ordered Banach spaces, SIAM Rev. 18 (4) (1976), 620–709.
- [4] H. Amann and T. Laetsch, Positive solutions of convex nonlinear eigenvalue problems, Indiana Univ. Math. J. 25 (3) (1976), 259–270.
- [5] P.M. Anselone, Collectively Compact Operator Approximation Theory and Applications to Integral Equations, Prentice-Hall Inc., Englewood Cliffs, NJ (1971).
- [6] J. Appell and P.P. Zabrejko, Nonlinear Superposition Operators, Cambridge Tracts in Mathematics, vol. 95, Cambridge University Press, Cambridge (1990).
- [7] W. Arendt, Approximation of degenerate semigroups, Taiwanese J. Math. 5 (2) (2001), 279–295.
- [8] W. Arendt and D. Daners, Uniform convergence for elliptic problems on varying domains, Math. Nachr. 280 (1–2) (2007), 28–49.
- [9] W. Arendt and D. Daners, Varying domains: stability of the Dirichlet and Poisson problem, Discrete Contin. Dyn. Syst. (Series A) 21 (1) (2008), 21–39.
- [10] W. Arendt and S. Monniaux, Domain perturbation for the first eigenvalue of the Dirichlet Schrödinger operator, M. Demuth and B. Schulze, eds, Partial Differential Equations and Mathematical Physics, Operator Theory: Advances and Applications, vol. 78, Birkhäuser, Basel, Holzau (1994–1995).
- [11] W. Arendt and M. Warma, *The Laplacian with Robin boundary conditions on arbitrary domains*, Potential Anal. **19** (4) (2003), 341–363.
- [12] D.G. Aronson, Non-negative solutions of linear parabolic equations, Ann. Scuola Norm. Sup. Pisa 22 (1968), 607–694.
- [13] J.M. Arrieta, Neumann eigenvalue problems on exterior perturbations of the domain, J. Differential Equations 118 (1) (1995), 54–103.
- [14] J.M. Arrieta, *Elliptic equations, principal eigenvalue and dependence on the domain*, Comm. Partial Differential Equations 21 (5–6) (1996), 971–991.
- [15] J.M. Arrieta, Domain dependence of elliptic operators in divergence form, Resenhas 3 (1) (1997), 107–122. Workshop on Differential Equations and Nonlinear Analysis (Águas de Lindóia, 1996).
- [16] J.M. Arrieta and S.M. Bruschi, *Boundary oscillations and nonlinear boundary conditions*, C. R. Math. Acad. Sci. Paris 343 (2) (2006), 99–104.
- [17] J.M. Arrieta and A.N. Carvalho, Spectral convergence and nonlinear dynamics of reactiondiffusion equations under perturbations of the domain, J. Differential Equations 199 (1) (2004), 143–178.
- [18] M.S. Ashbaugh and R.D. Benguria, Isoperimetric inequalities for eigenvalues of the Laplacian, Spectral Theory and Mathematical Physics: A Festschrift in Honor of Barry Simon's 60th Birthday, Proc. Sympos. Pure Math., vol. 76, Amer. Math. Soc., Providence, RI (2007), pp. 105–139.
- [19] I. Babuška and R. Výborný, Continuous dependence of eigenvalues on the domain, Czechoslovak Math. J. 15 (90) (1965), 169–178.
- [20] C. Bandle, *Isoperimetric Inequalities and Applications*, Monographs and Studies in Mathematics, vol. 7, Pitman, Boston, Mass (1980).

- [21] A.G. Belyaev, Averaging the third boundary value problem for the Poisson equation in a domain with a rapidly oscillating boundary, Vestnik Moskov. Univ. Ser. I Mat. Mekh. (6) (1988), 63–66.
- [22] J. Bergh and J. Löfström, *Interpolation Spaces. An Introduction*, Grundlehren der Mathematischen Wissenschaften, vol. 223, Springer-Verlag, Berlin (1976).
- [23] M. Biegert and D. Daners, Local and global uniform convergence for elliptic problems on varying domains, J. Differential Equations 223 (1) (2006), 1–32.
- [24] D. Bucur, Un théorème de caractérisation pour la γ-convergence, C. R. Acad. Sci. Paris Sér. I Math. 323 (8) (1996), 883–888.
- [25] D. Bucur, Characterization for the Kuratowski limits of a sequence of Sobolev spaces, J. Differential Equations 151 (1) (1999), 1–19.
- [26] D. Bucur, Characterization of the shape stability for nonlinear elliptic problems, J. Differential Equations 226 (1) (2006), 99–117.
- [27] D. Bucur and G. Buttazzo, Variational Methods in Shape Optimization Problems, Progress in Nonlinear Differential Equations and their Applications, vol. 65, Birkhäuser Boston Inc., Boston, MA (2005).
- [28] D. Bucur and N. Varchon, *Boundary variation for a Neumann problem*, Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4) 29 (4) (2000), 807–821.
- [29] D. Bucur and N. Varchon, A duality approach for the boundary variation of Neumann problems, SIAM J. Math. Anal. 34 (2) (2002), 460–477 (electronic).
- [30] D. Bucur and J.-P. Zolésio, Continuité par rapport au domaine dans le problème de Neumann, C. R. Acad. Sci. Paris Sér. I Math. 319 (1) (1994), 57–60.
- [31] D. Bucur and J.-P. Zolésio, Optimisation de forme sous contrainte capacitaire, C. R. Acad. Sci. Paris Sér. I Math. 318 (9) (1994), 795–800.
- [32] V.I. Burenkov and E.B. Davies, *Spectral stability of the Neumann Laplacian*, J. Differential Equations **186** (2) (2002), 485–508.
- [33] V.I. Burenkov and P.D. Lamberti, Spectral stability of general non-negative self-adjoint operators with applications to Neumann-type operators, J. Differential Equations 233 (2) (2007), 345–379.
- [34] G. Buttazzo and G. Dal Maso, Shape optimization for Dirichlet problems: relaxed formulation and optimality conditions, Appl. Math. Optim. 23 (1) (1991), 17–49.
- [35] J. Byeon, Existence of large positive solutions of some nonlinear elliptic equations on singularly perturbed domains, Comm. Partial Differential Equations 22 (9–10) (1997), 1731–1769.
- [36] G.A. Chechkin, A. Friedman and A.L. Piatnitski, *The boundary-value problem in domains with very rapidly oscillating boundary*, J. Math. Anal. Appl. 231 (1) (1999), 213–234.
- [37] D. Chenais, On the existence of a solution in a domain identification problem, J. Math. Anal. Appl. 52 (2) (1975), 189–219.
- [38] M. Chipot, Variational Inequalities and Flow in Porous Media, Applied Mathematical Sciences, vol. 52, Springer-Verlag, New York (1984).
- [39] J.B. Conway, A Course in Functional Analysis, 2nd edition, Graduate Texts in Mathematics, vol. 96, Springer-Verlag, New York (1990).
- [40] R. Courant and D. Hilbert, *Methods of Mathematical Physics, Vol. 1*, Interscience, New York (1953).
- [41] M.G. Crandall and R.H. Rabinowitz, Some continuation and variational methods for positive solutions of nonlinear elliptic eigenvalue problems, Arch. Rational Mech. Anal. 58 (3) (1975), 207–218.
- [42] G. Dal Maso and U. Mosco, *Wiener's criterion and* Γ *-convergence*, Appl. Math. Optim. **15** (1) (1987), 15–63.

- [43] E.N. Dancer, On the structure of solutions of an equation in catalysis theory when a parameter is large, J. Differential Equations 37 (3) (1980), 404–437.
- [44] E.N. Dancer, On the indices of fixed points of mappings in cones and applications, J. Math. Anal. Appl. 91 (1) (1983), 131–151.
- [45] E.N. Dancer, The effect of domain shape on the number of positive solutions of certain nonlinear equations, J. Differential Equations 74 (1) (1988), 120–156.
- [46] E.N. Dancer, On the influence of domain shape on the existence of large solutions of some superlinear problems, Math. Ann. **285** (4) (1989), 647–669.
- [47] E.N. Dancer, The effect of domain shape on the number of positive solutions of certain nonlinear equations. II, J. Differential Equations 87 (2) (1990), 316–339.
- [48] E.N. Dancer, On the number of positive solutions of some weakly nonlinear equations on annular regions, Math. Z. **206** (4) (1991), 551–562.
- [49] E.N. Dancer, Addendum to [47], J. Differential Equations 97 (2) (1992), 379-381.
- [50] E.N. Dancer, Weak convergence, local bifurcations and uniqueness theorems, Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl. 17 (3) (2006), 243–255.
- [51] E.N. Dancer and D. Daners, *Domain perturbation for elliptic equations subject to Robin boundary conditions*, J. Differential Equations 138 (1) (1997), 86–132.
- [52] D. Daners, Domain perturbation for linear and nonlinear parabolic equations, J. Differential Equations 129 (2) (1996), 358–402.
- [53] D. Daners, Local singular variation of domain for semilinear elliptic equations, J. Escher and G. Simonett, eds, Topics in Nonlinear Analysis, Progr. Nonlinear Differential Equations Appl., vol. 35, Birkhäuser, Basel (1999), pp. 117–141. The Herbert Amann Anniversary Volume.
- [54] D. Daners, Existence and perturbation of principal eigenvalues for a periodic-parabolic problem, S. Cantrell and C. Cosner eds, Nonlinear Differential Equations, Electronic Journal of Differential Equations, Conference 05, University of Miami 1999 (2000) http://ejde.math.unt.edu/. URL http://www.doaj.org/abstract?id=99663.
- [55] D. Daners, Heat kernel estimates for operators with boundary conditions, Math. Nachr. 217 (2000), 13–41.
- [56] D. Daners, Robin boundary value problems on arbitrary domains, Trans. Amer. Math. Soc. 352 (9) (2000), 4207–4236.
- [57] D. Daners, A priori estimates for solutions to elliptic equations on non-smooth domains, Proc. Roy. Soc. Edinburgh Sect. A 132 (4) (2002), 793–813.
- [58] D. Daners, Dirichlet problems on varying domains, J. Differential Equations 188 (2) (2003), 591–624.
- [59] D. Daners, Perturbation of semi-linear evolution equations under weak assumptions at initial time, J. Differential Equations 210 (2) (2005), 352–382.
- [60] D. Daners, A Faber–Krahn inequality for Robin problems in any space dimension, Math. Ann. 335 (5) (2006), 767–785.
- [61] D. Daners and P. Drábek, A priori estimates for a class of quasi-linear elliptic equations, Preprint (2007).
- [62] R. Dautray and J.-L. Lions, *Mathematical Analysis and Numerical Methods for Science and Technology, Vol. 2*, Springer-Verlag, Berlin (1988).
- [63] R. Dautray and J.-L. Lions, Mathematical Analysis and Numerical Methods for Science and Technology, Vol. 1, Springer-Verlag, Berlin (1990).
- [64] R. Dautray and J.-L. Lions, *Mathematical Analysis and Numerical Methods for Science and Technology, Vol. 5*, Springer-Verlag, Berlin (1992).
- [65] K. Deimling, Nonlinear Functional Analysis, Springer-Verlag, Berlin (1985).
- [66] P. Doktor, Approximation of domains with Lipschitzian boundary, Casopis Pest. Mat. 101 (3) (1976), 237–255.

- [67] D.E. Edmunds and W.D. Evans, Spectral Theory and Differential Operators, Clarendon Press, Oxford (1987).
- [68] L.C. Evans and R.F. Gariepy, *Measure Theory and Fine Properties of Functions*, Studies in Advanced Mathematics, CRC Press, Boca Raton, FL (1992).
- [69] E. Feireisl and H. Petzeltová, On the domain dependence of solutions to the two-phase Stefan problem, Appl. Math. 45 (2) (2000), 131–144.
- [70] L.E. Fraenkel, An Introduction to Maximum Principles and Symmetry in Elliptic Problems, Cambridge Tracts in Mathematics, vol. 128, Cambridge University Press, Cambridge (2000).
- [71] A. Friedman, B. Hu and Y. Liu, A boundary value problem for the Poisson equation with multi-scale oscillating boundary, J. Differential Equations 137 (1) (1997), 54–93.
- [72] I.M. Gel'fand, Some problems in the theory of quasilinear equations, Amer. Math. Soc. Transl.
 (2) 29 (1963), 295–381.
- [73] B. Gidas, W.-M. Ni and L. Nirenberg, *Symmetry and related properties via the maximum principle*, Comm. Math. Phys. **68** (3) (1979), 209–243.
- [74] B. Gidas and J. Spruck, *Global and local behavior of positive solutions of nonlinear elliptic equations*, Comm. Pure Appl. Math. **34** (4) (1981), 525–598.
- [75] B. Gidas and J. Spruck, A priori bounds for positive solutions of nonlinear elliptic equations, Comm. Partial Differential Equations 6 (8) (1981), 883–901.
- [76] D. Gilbarg and N.S. Trudinger, *Elliptic partial differential equations of second order, Classics in Mathematics*, Springer-Verlag, Berlin (2001), Reprint of the 1998 edition.
- [77] R.D. Grigorieff, Diskret kompakte Einbettungen in Sobolewschen Räumen, Math. Ann. 197 (1972), 71–85.
- [78] J.K. Hale and J. Vegas, A nonlinear parabolic equation with varying domain, Arch. Rational Mech. Anal. 86 (2) (1984), 99–123.
- [79] L. Hedberg, *Approximation by harmonic functions, and stability of the Dirichlet problem*, Exposition. Math. **11** (3) (1993), 193–259.
- [80] J. Heinonen, T. Kilpeläinen and O. Martio, Nonlinear potential theory of degenerate elliptic equations, Oxford Mathematical Monographs, Clarendon Press, New York (1993).
- [81] R. Hempel, On the lowest eigenvalue of the Laplacian with Neumann boundary condition at a small obstacle, J. Comput. Appl. Math. 194 (1) (2006), 54–74.
- [82] A. Henrot, Continuity with respect to the domain for the Laplacian: a survey, Control Cybernet. 23 (3) (1994), 427–443.
- [83] A. Henrot, Extremum problems for eigenvalues of elliptic operators, Frontiers in Mathematics, Birkhäuser Verlag, Basel (2006).
- [84] D. Henry, Perturbation of the Boundary in Boundary-value Problems of Partial Differential Equations, London Mathematical Society Lecture Note Series, vol. 318, Cambridge University Press, Cambridge (2005).
- [85] M.W. Hirsch, *Differential Topology*, Graduate Texts in Mathematics, vol. 33, Springer-Verlag, New York (1976).
- [86] J. Húska and P. Poláčik, *The principal Floquet bundle and exponential separation for linear parabolic equations*, J. Dynam. Differential Equations 16 (2) (2004), 347–375.
- [87] S. Jimbo, Singular perturbation of domains and semilinear elliptic equation, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 35 (1) (1988), 27–76.
- [88] S. Jimbo, Singular perturbation of domains and the semilinear elliptic equation. II, J. Differential Equations **75** (2) (1988), 264–289.
- [89] S. Jimbo, *The singularly perturbed domain and the characterization for the eigenfunctions with Neumann boundary condition*, J. Differential Equations **77** (2) (1989), 322–350.
- [90] S. Jimbo, Perturbation formula of eigenvalues in a singularly perturbed domain, J. Math. Soc. Japan 45 (2) (1993), 339–356.

- [91] D.D. Joseph and T.S. Lundgren, *Quasilinear Dirichlet problems driven by positive sources*, Arch. Rational Mech. Anal. **49** (1972/73), 241–269.
- [92] T. Kato, *Perturbation Theory for Linear Operators*, 2nd edition, Grundlehren der Mathematischen Wissenschaften, vol. 132, Springer-Verlag, Berlin (1976).
- [93] M.V. Keldyš, On the solvability and stability of the Dirichlet problem, Uspekhi Mat. Nauk 8 (1941), 171–231. English Translation: Amer. Math. Soc. Translations (2) 51 (1966), 1–73.
- [94] M.A. Krasnosel'skii, On a theorem of M. Riesz, Soviet Math. Dokl. 1 (1960), 229-231.
- [95] P.C. Kunstmann, *Heat kernel estimates and* L^p *spectral independence of elliptic operators*, Bull. London Math. Soc. **31** (3) (1999), 345–353.
- [96] E.H. Lieb and M. Loss, *Analysis*, Graduate Studies in Mathematics, vol. 14, American Mathematical Society, Providence, RI (1997).
- [97] P.-L. Lions, Symétrie et compacité dans les espaces de Sobolev, J. Funct. Anal. 49 (3) (1982), 315–334.
- [98] Y. Liu, Smoothing the domain out for positive solutions, Bull. Austral. Math. Soc. 61 (3) (2000), 405–413.
- [99] N. Lloyd, Degree Theory, Cambridge University Press, Cambridge (1978).
- [100] V.G. Maz'ya, Classes of domains and imbedding theorems for function spaces, Soviet Math. Dokl. 1 (1960), 882–885.
- [101] V.G. Maz'ya, *Sobolev Spaces*, Springer Series in Soviet Mathematics, Springer-Verlag, Berlin (1985), Translated from the Russian by T. O. Shaposhnikova.
- [102] U. Mosco, *Convergence of convex sets and of solutions of variational inequalities*, Adv. Math. 3 (4) (1969), 510–585.
- [103] J. Nečas, On domains of type N, Czechoslovak Math. J. 12 (87) (1962), 274–287.
- [104] J. Nečas, Les Méthodes Directes en Théorie des Équations elliptiques, Academia, Prague, 1967.
- [105] S. Ozawa, Singular variation of domain and spectra of the Laplacian with small Robin conditional boundary I, Osaka J. Math. 29 (1992), 837–850.
- [106] S. Ozawa and S. Roppongi, Singular variation of domain and spectra of the Laplacian with small Robin conditional boundary II, Kodai Math. J. 15 (1992), 403–429.
- [107] R. Rannacher, Zur Stabilität diskreter Teilspektren singulärer Differentialoperatoren zweiter Ordnung gegenüber Störungen der Koeffizienten und der Definitionsgebiete, Abh. Math. Sem. Univ. Hamburg 43 (1975), 198–215.
- [108] J. Rauch and M. Taylor, *Potential and scattering theory on wildly perturbed domains*, J. Funct. Anal. 18 (1) (1975), 27–59.
- [109] E. Sánchez-Palencia and P. Suquet, Friction and homogenization of a boundary, Free Boundary Problems: Theory and Applications, Vol. I, II (Montecatini, 1981), Pitman, Boston, MA (1983), pp. 561–571.
- [110] G. Savaré and G. Schimperna, *Domain perturbations and estimates for the solutions of second order elliptic equations*, J. Math. Pures Appl. (9), Journal de Mathématiques Pures et Appliquées. Neuvième Série (ISSN: 0021-7824) **81** (11) (2002), 1071–1112. doi:10.1016/S0021-7824(02)01256-4.
- [111] R.E. Showalter, *Hilbert Space Methods for Partial Differential Equations*, Pitman, London (1977).
- [112] G. Stampacchia, Équations Elliptiques du Second Ordre à Coefficients Discontinus, Les Presses de l'Université de Montréal, Montréal, 1966.
- [113] P. Stollmann, A convergence theorem for Dirichlet forms with applications to boundary value problems with varying domains, Math. Z. 219 (2) (1995), 275–287.
- [114] M. Struwe, Variational Methods, 2nd edition, Ergebnisse der Mathematik und ihrer Grenzgebiete, vol. 34, Springer-Verlag, Berlin (1996).

- [115] C.A. Stuart, A variational approach to bifurcation in L^p on an unbounded symmetrical domain, Math. Ann. 263 (1) (1983), 51–59.
- [116] F. Stummel, Diskrete Konvergenz linearer Operatoren. I, Math. Ann. 190 (1970), 45–92.
- [117] F. Stummel, Diskrete Konvergenz linearer Operatoren. II, Math. Z. 120 (1971), 231-264.
- [118] F. Stummel, Diskrete Konvergenz linearer Operatoren. III, Linear Operators and Approximation, Internat. Ser. Numer. Math., vol. 20, Birkhäuser, Basel, Oberwolfach (1971– 1972).
- [119] F. Stummel, Perturbation theory for Sobolev spaces, Proc. Roy. Soc. Edinburgh Sect. A 73 (1) (1975), 5–49.
- [120] V. Šverák, On optimal shape design, J. Math. Pures Appl. (9) 72 (6) (1993), 537-551.
- [121] A.E. Taylor and D.C. Lay, *Introduction to Functional Analysis*, 2nd edition, John Wiley & Sons, New York, Chichester, Brisbane (1980).
- [122] M.J. Ward, W.D. Henshaw and J.B. Keller, Summing logarithmic expansions for singularly perturbed eigenvalue problems, SIAM J. Appl. Math. 53 (1993), 799–828.
- [123] M.J. Ward and J.B. Keller, Strong localized perturbations of eigenvalue problems, SIAM J. Appl. Math. 53 (1993), 770–798.
- [124] J. Weidmann, Stetige Abhängigkeit der Eigenwerte und Eigenfunktionen elliptischer Differentialoperatoren vom Gebiet, Math. Scand. 54 (1) (1984), 51–69.
- [125] K. Yosida, *Functional Analysis*, 6th edition, Grundlehren der Mathematischen Wissenschaften, vol. 123, Springer-Verlag, Berlin (1980).